The Future in Mind: Aspirations and Long-term Outcomes in Rural Ethiopia^{*}

Tanguy Bernard, Stefan Dercon, Kate Orkin Giulio Schinaia, Alemayehu Seyoum Taffesse

25th June 2025

Abstract

Aspirations may condition the future-oriented choices of individuals and thus may play a role in the persistence of poverty or the effort to break out of it. We run a randomised control trial in remote, rural Ethiopia to explore this and evaluate an intervention which aims to change how poor people perceive their future opportunities, alter their aspirations and, through that, modify their investment decisions. A treatment group was shown video documentaries featuring individuals from similar communities who escaped poverty through their own efforts and serve as relatable role models. Five years after the screening took place, the treated households had increased future-oriented investments in agriculture, children's education and assets. The results can be explained by an increase in aspirations in terms of lifetime goals. Overall, this research uniquely provides evidence that a light-touch behavioural intervention can have persistent economic impacts on a poor population.

Keywords: aspirations, long-term, investment, role models JEL codes: D03; I31

*Author affiliations: Bernard - University of Bordeaux & IPORA; Dercon - University of Oxford; Orkin - University of Oxford; Schinaia - University of Chicago; Taffesse - International Food Policy Research Institute. This paper is dedicated to the memory of Ayelech Kifle, the main character of the first short documentary produced for this study, which inspired the later development of other videos. We thank the subjects of the documentaries for sharing their stories and all participants for their time over the five years of the study. The authors thank research assistants Geetika Nagpal, Dylan Reich, Fanaye Tadesse, and Marc Witte, fieldwork managers Tewodros Abate, Kibrom Hirfrfot and Bezabih Tesfaye; Mekamu Kedir, Emily Schmidt, IFPRI Addis Ababa staff; Doba *Woreda* Administration, Next Studio and Tadesse Fayissa for producing documentaries, and staff at the International Food Policy Research Institute and Centre for Study of African Economies. This research was funded by the UK Department for International Development (DFID) as part of the Institutions for Pro-Poor Growth Consortium (iiG), SEVEN (Social Equity Venture Fund) and USAID through a Feed the Future Ethiopia grant. Bernard and Taffesse acknowledge the support of the IFPRI Development Strategy and Governance, and Markets, Trade and Institutions divisions. This paper replaces and substantially expands on the 2014 unpublished working paper 'The Future in Mind: Aspirations and Forward-Looking Behaviour in Rural Ethiopia', which reported on the short-run results.

1 Introduction

The persistence of extreme poverty has long concerned economists and policymakers. Recent theoretical research suggests that social and psychological factors may play a role in perpetuating it.¹ Interventions targeting specific psychological attributes can change beliefs and improve economic outcomes in the short term among low-income groups.² A few recent studies examine long-term effects of intensive clinical psychological treatments on economic outcomes of depressed adults (Baranov et al., 2020; Bhat et al., 2022) or men at risk of engaging in criminal behaviour (Blattman et al., 2023). But it remains unclear whether scalable, population-wide behavioural interventions can have lasting impacts. This paper addresses this gap: we show that a light-touch behavioural intervention has persistent economic impacts after five years on a random sample of a population mostly living in extreme poverty.

Our intervention is based on the idea that those living in extreme poverty may struggle to envision a better future, which lowers their aspirations and reduces motivation to invest. They may have had fewer successful role models from their community for inspiration (Appadurai, 2004; Durlauf, 1996; Genicot and Ray, 2020; Ray, 2006) and may face persistent social norms or values driving demotivating beliefs rooted in "zero-sum" thinking (Carvalho et al., 2023). Our intervention aims to increase individuals' economic aspirations and change their mental models of what they can become and achieve (Hoff and Stiglitz, 2016).³ The intervention uses video documentaries showcasing successful role models from similar communities, drawing on psychological theories that people's aspirations are influenced by the outcomes of others like them (Bandura, 1977; La Ferrara, 2019).

We test this intervention in a field experiment in a remote, poor district in Ethiopia with 2,112 individuals (1,152 households). Some households (head and spouse) were randomly invited to watch video documentaries we made about individuals from similar areas who had succeeded in agriculture or business through their own efforts. In the videos, the role models describe how they improved their socio-economic position from being poor to being relatively successful, through setting goals, perseverance and hard work. We compare the treated group to a placebo group, who were randomly invited to watch an Ethiopian entertainment programme, and to a control group, who were simply surveyed. We collected data before the intervention, straight after the screening occurred, after six months, and again five years later.⁴

We find that this simple intervention significantly improves economic outcomes after five

^{1.} See, for example, Appadurai (2013); Benabou (1996); Bisin and Verdier (2001); Dalton, Ghosal, and Mani (2016); Durlauf (1996); Genicot and Ray (2017); Lybbert and Wydick (2018).

^{2.} Recent studies include Ashraf et al. (2022); Barker et al. (2022); Bossuroy et al. (2022); Campos et al. (2017); Ghosal et al. (2022); Heller et al. (2016); John and Orkin (2022); McKelway (2021). See Kremer, Rao, and Schilbach (2019) for a review.

^{3.} We conceptualise aspirations as desired goals for the future, which motivate investment and effort in order to attain them (Bandura, 1977; Locke and Latham, 1990).

^{4.} A working paper, Bernard et al. (2014), reported the six month results. This paper expands on and replaces that paper. Subsequently, other experiments using videos of role models have appeared, presenting short-run results on economic and other outcomes (Batista and Seither, 2019; Cecchi et al., 2022; Leight et al., 2024; Lubega et al., 2021; Orkin et al., 2023; Rojas Valdes, Wydick, and Lybbert, 2021).

years by increasing effort and investment.⁵ First, five years after the experiment, treated households report higher labour supply and more use of agricultural inputs. They spend around one extra hour per day working on their own farms, are more likely to spend on modern crop inputs, and on animal feed and vet supplies, and have increased holdings of tools by 20 per cent. Second, treated households increase investments in human capital. At the five-year follow-up, treated households spend approximately 36 per cent more than other study participants on their children's education. Their children have attained more years of schooling: twice as many children who were of school-going age at the time of screening have completed full primary school five years later. Third, treated households experience meaningful changes in living standards: they report one and a half fewer weeks of food insecurity in the previous year, have accumulated 29 per cent larger holdings of durable goods, like cellphones and household equipment, and have improved housing quality, although consumption expenditure is not affected. We find suggestive evidence that economic behaviour began to change soon after the treatment. Using a shorter survey collected six months after the screenings, we find increases in savings and investment in education.

We find evidence consistent with these economic changes resulting from increases in the individuals' aspirations for the future, where aspirations are desired goals for the future, which motivate investment and effort. We use locally validated survey measures of aspirations (Bernard and Taffesse, 2014). These capture the level of income, assets, or children's education individuals hoped to achieve in their lifetime or expected they would achieve in ten years. The treatment group have higher aspirations and expectations after five years, relative to both placebo and control. There are positive, although noisier, effects straight after treatment and after six months, consistent with aspirations increasing slowly as people experiment with possible alternative futures (Appadurai, 2004) or learn-by-doing when demotivating beliefs start to shift (Carvalho et al., 2023). Individuals may start investing, seeing the returns of their investments, and aiming to do better in the future. We can also rule out the possibility highlighted by some existing models of aspirations (Dalton, Ghosal, and Mani, 2016; Genicot and Ray, 2017). We see positive effects on labour supply, one measure of effort, and small positive effects on subjective wellbeing, rather than negative ones.

We can exclude some plausible alternative mechanisms through which the intervention could have led to this outcome. We measure time and risk preferences, grit, information transmission, and beliefs about the returns to technology. We find no change in these. We do find some effects six months after exposure to the videos on measures of locus of control – beliefs about whether individuals or fate control people's lives. However, these do not persist after five years, unlike effects on aspirations. We view this as a less likely mechanism: if locus of control drove changes in investment, we would expect effects on investment to dissipate

^{5.} Results reported in the introduction are significant relative to the control group at 5 per cent or less, and robust to multiple hypothesis testing.

alongside effects on locus of control. The design of the intervention also rules out further mechanisms. Unlike in other studies that rely on variation in exposure to real-life role models such as teachers or peers, participants receive no mentorship or support other than exposure to our videos (Kearney and Levine, 2020). Exposure to the screening itself or to outsiders or being selected for the intervention do not account for effects: a placebo group shown a local entertainment programme are unaffected relative to the control group. Lastly, we run a suggestive test for whether there are spillovers from treated to untreated individuals, for example if untreated individuals learn by observing treated households' changes in behaviour. We find little evidence of large spillovers, even though our tests face limitations.

This study contributes to the literature in three ways. We provide the first experimental evidence on how an intervention targeting aspirations affects economic investment in the longer term. Several theoretical models posit that aspirations can influence investment (Besley, 2016; Dalton, Ghosal, and Mani, 2016; Genicot and Ray, 2017) and several papers use observational data to document aspiration-investment associations (Janzen et al., 2017; Ross, 2019; Serneels and Dercon, 2021; Eble and Escueta, 2022). A growing number of papers test interventions to shift aspirations and economic outcomes in the short-run. Some experiments use light-touch interventions similar to ours targeting households in disadvantaged circumstances (Leight et al., 2024; Lubega et al., 2021; Orkin et al., 2023) or entrepreneurs (Batista and Seither, 2019), with mixed effects. Others involve intensive multi-week training to promote future-oriented behaviour, including elements building aspirations, encouraging delayed gratification and promoting ambitious mindsets, also with mixed effects (Cecchi et al., 2022; McKenzie, Mohpal, and Yang, 2022; Rojas Valdes, Wydick, and Lybbert, 2021). We provide the first evidence on the longerterm causal effects (beyond two years) of an intervention targeting aspirations to boost economic outcomes. We track 96 per cent of households after five years, a high fraction compared to recent long-term follow-ups of experiments in low- and middle-income countries (Bouguen et al., 2019).

Second, within the literature on the psychology of poverty, we add to the limited evidence on the long-term impacts of psychological interventions, by showing how a population-wide light-touch intervention can have long-term impacts on economic outcomes. A growing body of intervention-based studies, including in poor settings, examine the effects on decision-making of other psychological characteristics beyond aspirations, including self-regulation, self-efficacy, grit, and time preferences (Alan and Ertac, 2018; Alan, Boneva, and Ertac, 2019; Ashraf et al., 2022; Blattman, Jamison, and Sheridan, 2017; Blattman et al., 2023; Bossuroy et al., 2022; Campos et al., 2017; Heller et al., 2016; Ghosal et al., 2022; John and Orkin, 2022; McKelway, 2021) and of clinically diagnosed mental health conditions (Angelucci and Bennett, 2024; Baranov et al., 2020; Barker et al., 2022; Bhat et al., 2022; Haushofer, Mudida, and Shapiro, 2020). Among these, only Baranov et al. (2020), Bhat et al. (2022), and Blattman et al. (2023) study long-term economic impacts (respectively after seven, five, and ten years). These all study intensive, multi-week psychotherapy targeted at subpopulations of adults with depression or a record of antisocial behaviour. Our study provides the first longer-term evidence on how a light-touch non-targeted population-wide behavioural intervention affects economic outcomes, showing how overcoming households' internal psychological constraints can unlock investment.

Finally, we contribute to work on the effect of role models, including in visual media, on investments. Real-world female role models improve girls' and young women's selection into and performance in male-dominated fields in high-income countries (Greene, Sullivan, and Beyard-Tyler, 1982; Stout et al., 2011; Porter and Serra, 2020) and girls' education investments and women's fertility in low- and middle-income countries (Beaman et al., 2012; Jensen and Oster, 2009). Exposure to role models also affects investments in both boys' and girls' education (Fairlie, Hoffmann, and Oreopoulos, 2014; Kofoed and McGovney, 2019; Macours and Vakis, 2014). Effects are similar when role models are featured in visual media (Chong, Duryea, and La Ferrara, 2012; Riley, 2022; for a review see La Ferrara, 2016). We add to this literature by showing that exposure to role models has persistent effects on adults' labour supply, investment and asset wealth; by using an experimental design to cleanly identify causal links between exposure to role models, changes in aspirations, and behaviour; and by examining a range of psychological mechanisms through which role model effects might occur.

These findings have important implications for designing effective poverty-reduction interventions: a carefully designed but low-cost intervention induces both psychological and economic changes that persist after five years, suggesting a promising avenue for research and intervention design. A back-of-the-envelope calculation suggests the intervention was cost-effective, with durable asset increases alone outweighing the costs by a factor of two at a reasonable scale of implementation. We are nevertheless cautious about the external validity of the findings. The study area was chosen because it was remote, with limited exposure to media, so even a single screening was a memorable experience, potentially contributing to the persistent and relatively substantial impacts. Our findings can be seen as a proof of concept, showing that this type of intervention targeting the psychological mechanism of aspirations may have persistent impacts, even if the magnitude of effects differs across contexts.

Section 2 describes the context, our sample and its characteristics at baseline. Section 3 motivates the experiment, with a conceptual framework to understand aspirations and descriptive evidence of aspirations which may limit investment in our setting. Section 4 discusses our intervention design, estimation strategy and tests for experimental integrity. Section 5 provides the results of the intervention on investment decisions; indicators of household well-being; aspirations; and alternative mechanisms, after five years. It also reports results on a subset of measures after six months, provides cost-effectiveness analysis, and tests for spillovers. Section 6 concludes.

2 Context, the sample and its characteristics

2.1 Context

Our study took place in Doba, a mountainous remote rural administrative district of Ethiopia, 380 kilometres east of the capital city of Addis Ababa. Most of Doba's population are Muslim and ethnically Oromo (88 per cent), the rest Orthodox Christian and Amhara. At the time of the experiment, Doba was one of the poorer districts in the country: it was one of the first districts selected in 2005 for the national social protection programme targeted at the most chronically food-insecure districts in Ethiopia. Doba is also extremely remote: most surveyed villages were accessible only by 4x4 vehicle and some required camel transportation. Households had limited exposure to life beyond their district. At baseline there was limited exposure to television: only 11 per cent of respondents watched TV once a week or more, 28 per cent watched at least once a month and 62 per cent watched about once a year, if ever. Only 4 per cent of the households owned a cellphone, and no household owned a television. Doba residents would be likely to find a television show featuring characters like them memorable and to pay attention to the screening content.

During our study, Ethiopia was one of the world's fastest-growing economies, with GDP growth of almost 10 per cent annually. The country has halved the poverty headcount since 2000. But 31 per cent of households still lived in extreme poverty by the time of our endline, using the global benchmark of 1.90 USD 2011 PPP per person per day.⁶ Even by Ethiopian standards, the households in our study remain extremely poor: at endline, 69 per cent had consumption per person per day below the \$1.90 benchmark, and the rest not far above this level.

2.2 Sampling and data collection

We implemented the experiment in 64 villages. To generate this list, we took the Central Statistical Agency's list of 189 rural villages in Doba with a population of 50 to 100 households from the 2007 census and we randomly selected 84 villages. We found 16 screening venues close to those villages: classrooms or agricultural extension facilities with capacity for at least fifty individuals. We selected the 64 villages in our experiment based on logistical considerations, choosing the 64 villages closest to those screening venues such that we had four villages within walking distance of the 16 screening sites. The district is mountainous and remote, so we minimised walking distance to screenings to maximise compliance. Randomisation is within the villages, so this does not affect internal validity, even if the study sample is not representative for the district as a whole. Appendix Figure A.1 shows the 84 villages, marking the 64 selected villages.

Within each village, we compiled a list of all households with the assistance of the community (*kebele*) leader (who runs three to six neighbouring villages). We randomly sampled 18 households from each of the 64 villages to survey, with replacement for households that were away, ill or had just given birth.

The main sample for analysis consists of 1,152 households and 2,112 individuals surveyed at baseline (and any subsequent follow-up) in these 64 villages.⁷ We visited villages for the baseline survey and intervention (round 1, between September and November 2010), the midline follow-up survey six months after the baseline (round 2, between March and May 2011), and a long-run follow-up (endline) survey five years after the baseline (round 3, between

^{6.} Data from the World Bank's Poverty and Inequality Platform, https://pip.worldbank.org/.

^{7.} In seven percent of households, the household head is single.

December 2015 and January 2016). Appendix Figure A.2 shows the timeline of the surveys.

Follow-up surveys were conducted at households' homes by enumerators blind to household treatment status. The household head answered questions on issues like household composition, assets and children's schooling, so most economic variables were collected at the household level. We also collect individual-level information separately from both the household head and spouse on psychological variables such as aspirations and preferences, as well as savings. Spouses were interviewed separately, usually by two different enumerators interviewing them simultaneously, either in or around their house. Appendix Section E details the construction of the variables used in our analysis.

Both midline and endline measure aspirations, expectations, other preferences and beliefs, and subjective wellbeing, as well as labour supply, effort and investment in education, savings and use of credit, using the same questionnaires, so these outcomes are consistent across rounds. The endline also captures investment in productive activities and assets and various dimensions of standard of living: food insecurity, consumption expenditure, asset wealth, housing quality, and subjective wellbeing. We collected a shorter set of outcomes in the short run because we did not think other outcomes, such as asset accumulation and consumption, could have changed, as households received no new resources.

2.3 Characteristics of the sample

Table 1 describes the economic lives of our sample at baseline. The sample consists of small farm households, on average 5.6 members. Crop agriculture and livestock-related activities, including product and animal sales, are the main sources of income. Adults spend on average five hours per day on these activities. Livestock holdings are modest, averaging 411 USD PPP per adult-equivalent, equivalent to just over one cow (370 USD PPP). Holdings of tools are low, at 24 USD PPP per adult. Savings are limited: only 23% have any, with an average of 10.25 USD PPP among savers. Education levels are low, with adult men averaging 3 years and women averaging 1 year of schooling. Most of the generation before the respondents had no education at all: only 16 per cent of the respondents' fathers completed any years of education. Although enrolment levels have increased with free primary education policies, 42 per cent of children aged 7 to 15, school-going age, were not enrolled at baseline.

We split the sample by terciles of the value of durable assets at baseline, a proxy for wealth. This shows the type of investments commonly associated with accumulating wealth in this setting. Our measure of assets is associated with other measures of living standards, such as higher housing quality. In this setting, better-off households invest more in livestock (especially cattle), spend more time in work on their farms, save more, and invest more in education.

3 Motivation for the experiment

This section motivates the core hypothesis of this paper: that increasing aspirations for the future, via exposure to similar role models, increases effort and investment, and, in turn, living standards.

| | (1) | (2) | (3) | 3) (4) | (5) | (6) | |
|---|--------------|---------|---------|---------|----------------------|--------------|--|
| | Whole sample | Lower | Middle | Upper | r <i>p</i> -value | Observations | |
| | | tercile | tercile | tercile | F | | |
| Assets | | | | | | | |
| Value of durable assets excluding tools (USD) per ad. equiv PPP | 18.72 | 0.00 | 5.49 | 51.25 | 0.00 | 1119 | |
| Value of tools (USD) per ad. equiv PPP | 24.32 | 16.28 | 17.63 | 38.18 | 0.00 | 1111 | |
| Value of house (USD) per ad. equiv PPP | 371.53 | 228.67 | 323.08 | 531.02 | 0.00 | 1082 | |
| Total savings (USD) PPP | 10.25 | 6.95 | 11.08 | 12.14 | 0.01 | 2020 | |
| Has any savings | 0.23 | 0.22 | 0.23 | 0.25 | 0.15 | 2048 | |
| Livestock | | | | | | | |
| Value of livestock (USD) per ad. equiv PPP | 411.49 | 281.09 | 348.13 | 590.91 | 0.00 | 1110 | |
| Value of cattle (USD) per ad. equiv PPP | 366.87 | 235.44 | 296.43 | 524.95 | 0.00 | 1118 | |
| Value of sheep or goat (USD) per ad. equiv PPP | 39.07 | 28.96 | 30.98 | 54.82 | 0.00 | 1118 | |
| Labour supply | | | | | | | |
| Household size | 5.61 | 5.36 | 6.28 | 5.30 | 0.72 | 1119 | |
| Daily minutes in paid work per adult aged above 15 | 11.68 | 14.28 | 9.75 | 10.11 | 0.20 | 1109 | |
| Daily minutes on family farm per adult aged above 15 | 308.90 | 300.24 | 306.35 | 318.95 | 0.06 | 1110 | |
| Human capital investment | | | | | | | |
| Share of children at school in the 7-15 age-group | 0.58 | 0.57 | 0.57 | 0.61 | 0.18 | 802 | |
| Schooling expenditure per child aged 7-15 (USD) PPP | 17.47 | 16.92 | 17.65 | 18.41 | 0.53 | 1110 | |
| Highest education level among male adults (years) | 3.43 | 3.02 | 3.22 | 4.03 | 0.00 | 1045 | |
| Highest education level among female adults (years) | 1.08 | 0.88 | 0.97 | 1.45 | 0.00 | 1025 | |
| Respondents' demographic characteristics | | | | | | | |
| Years of education | 1.33 | 0.94 | 1.15 | 1.95 | 0.00 | 2048 | |
| Listens to radio at least once a week | 0.61 | 0.50 | 0.49 | 0.83 | 0.00 | 2044 | |
| Ever lived outside of current district six months | 0.10 | 0.08 | 0.10 | 0.11 | 0.08 | 2045 | |
| Father has any education | 0.16 | 0.13 | 0.15 | 0.20 | 0.00 | 2048 | |

Table 1: ECONOMIC ACTIVITIES BY TERCILES OF DURABLE ASSETS

Notes: Column 1 is the sample mean. Columns 2-4 are sample means, conditional on the household being in the lower, middle and upper terciles of the value of durable assets (excluding tools) at baseline, an approximation for living standards. Columns 5 reports the *p*-value from a *t*-test of equality between the mean of the lower and upper tercile. Columns 6 reports the number of observations. Variables are measured at the household level at baseline, except savings and demographic characteristics, which are measured for both the household head and spouse. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 90^{th} percentile. Variables are defined in detail in Appendix E. The OECD adult equivalence scale gives weight 0.5 to each individual younger than 16 and weight 0.7 to all other adults that are not the household head. Durable assets include radios, mobile phones, jewellery, and furniture. Tools include ploughs, hoes, axes. The value of house is how much their house would cost to build today (in current prices), including materials and labour costs. Total savings refers to the value of savings held inside and outside the home. Livestock and durable assets are valued using self-reported hypothetical sale prices. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using the national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1.

First, we outline theories in social science that argue that aspirations affect long-term decisions, and that aspirations are influenced by the experiences of similar others. In contexts of poverty, it has been argued people are less exposed to others like them who have changed their economic circumstances. As a result, they tend to form lower aspirations, which in turn limits their future-oriented investments. Local belief systems may further entrench low aspirations. We offer anthropological evidence from Ethiopia and elsewhere in line with these conjectures. Second, we sketch a simple economic model that defines aspirations in economic terms. We also show how an intervention that increases aspirations might affect economic decision-making. Appendix B offers a plausible formal treatment to derive predictions. Third, we describe our measures of aspirations. We find that in our specific poor and remote setting, poorer people in our sample have lower aspirations, as reported on our quantitative measures of aspirations, and less exposure to experiences through which they might form higher aspirations.

3.1 Aspirations: concept, origins, and implications

The role of aspirations in decision-making: Theories in a number of social sciences argue that aspirations play an important role in long-term decision-making. Social psychologists theorise that aspirations provide motivation and regulation for current choices, including inspiring persistence with a long-term course of action through setbacks, and that achieving aspirations gives satisfaction (Bandura, 2001; Fishbach and Ferguson, 2007; Locke and Latham, 1990).⁸ In social anthropology, seminal work by Arjun Appadurai (2004) argues that aspirations profoundly shape people's trajectories. He defines "the capacity to aspire" as a "navigational capacity": the ability to read and then navigate "a map of a journey into the future" (p. 76). An individual's aspirations capture their understanding of the pathways available to people like them. Aspirations hence influence whether individuals consider or dismiss entire classes of options.

Social influences impact aspirations: Anthropological and psychological accounts of aspirations highlight that aspirations are influenced by both social processes and individual experiences (Hoff and Stiglitz, 2016). In psychology, aspirations are "socially learned": people observe the outcomes of other similar individuals and draw conclusions about what is possible for them (Bandura, 1977). Anthropological accounts highlight the role of exposure to how others in one's network have navigated their life choices, as well as exposure to others' experiences via travel, the media or literature (Appadurai, 2004; Willis, 1977; Wilson, 1987). In economics, Genicot and Ray (2020) survey how social influences may determine aspirations. The implication is that if people have not encountered role models similar to them who have been able to change their economic circumstances via their own actions, they may struggle to believe that people like them are able to achieve such change. This motivates our intervention.

Aspirations in states of poverty: Anthropologists in diverse contexts emphasise that high levels of poverty can result in distorted or limited aspirations and future orientation, because the social processes through which individuals form aspirations differ radically with wealth. For example, with reference to slum dwellers in Mumbai, Appadurai (2004) argues:

"The more privileged in any society simply have used the map of its norms to explore the future more frequently and realistically, and to share knowledge with one another more routinely [...]. The poorer members, because of their lack of opportunities to practice this navigational capacity (in turn because their situations permit fewer experiments and less easy archiving of alternative futures), have a more brittle horizon of aspirations. I am not saying that the poor cannot wish, want, need, plan, or aspire. But part of poverty is a diminishing of the circumstances in which these practices occur" (p. 69).

Concretely, richer people form higher aspirations because they have broader networks with

^{8.} For example, Bandura (2001, p.7) highlights "if actions were performed only on behalf of anticipated external rewards, people would behave like weather vanes, constantly shifting direction. In actuality, people display considerable self-direction in the face of competing influences."

more diverse experiences and more opportunities for experimentation with alternative futures enabled by more material resources.

Aspirations and belief systems: Aspirations are influenced by people's perceptions of social structures, norms and economic constraints, and how these limit (or open up) more general and generic possibilities and paths (Appadurai, 2004). In contexts of existing deprivation, with resource and technology constraints, existing normative belief systems may entrench lower aspirations, limiting future orientation and economic progress. These belief systems shape how society allocates economic resources and opportunities, reinforcing existing structures through limiting any space for change (Durkheim, 1893).

For example, beliefs may have emerged that all interactions are zero-sum in nature – gains for one individual come at the expense of others – leading to norms and values that regulate behaviour in line with the status quo, as in Foster (1965)'s study in Mexico. Carvalho et al. (2023) show theoretically how such beliefs lead to norms that depress effort and to lower welfare outcomes, and find evidence that zero-sum thinking regulates behaviour in the Democratic Republic of Congo and across the world.⁹ Whelan (1996)'s study of marginalised communities in Ireland documents how deprivation led to limited ambitions for altering circumstances, in turn reinforced by high levels of fatalism, "a system of beliefs which holds that everything has an appointed outcome which cannot be altered" (p.46). Similarly fatalistic beliefs have been documented in poor communities across diverse societies, including India, Thailand, Spain and Columbia (Elder, 1966; Ingersoll, 1966; Díaz et al., 2015). These anthropological accounts illustrate how belief systems can shape people's mental models of what is possible for them. In turn, both will then condition people's aspirations (Hoff and Stiglitz, 2016; World Bank, 2015).

3.2 Ethnographic view on aspirations in Ethiopia

Anthropological accounts of low aspirations in rural Ethiopia: Scholars have explicitly commented on distorted or limited aspirations in deprived Ethiopian communities, including in Levine (1965), Kebede (1999), and Wollie (2009). For example, Rahmato and Kidani (1999) documents Ethiopian proverbs describing people's life outlook such as "we live only for today", "this is a life of no thought for tomorrow", "we are waiting to die while seated", or "we have neither a dream nor an imagination". These statements highlight helplessness, desperation, fatalism and a lack of hope.

The role of local belief systems: While not unique or especially pronounced in Ethiopia, certain belief systems conceptualise destiny as determined by divine forces rather than individual agency. They imbue existing social structures and hierarchies with moral legitimacy, constraining the scope for goal-setting and imagining alternative futures (Levine, 1965; Kebede, 1999; Wollie, 2009). These ideas have multiple sources, including from religion as practised

^{9.} Such ideas are also common in the West: before the enlightenment, change and progress were seen as a threat in Europe (Mokyr, 2016), and zero-sum thinking and its historical roots can help to explain the present-day US political divide (Chinoy et al., 2025).

in the specific areas studied.¹⁰ The combined worldview tends to emphasise deference to authority, tradition and social hierarchy; social conformity and cohesion; resistance to external influence; and avoidance of the potential disruption that can be caused by deviating from the norm (Alamayo, 2021).¹¹

Evidence on belief systems that may condition aspirations: In our sample, we find quantitative evidence of fatalistic, deterministic belief systems which may distort aspirations. Consistent with the anthropological evidence above, our sample, who are largely living in poverty, agree strongly with psychological scales capturing a fatalistic, deterministic outlook. Our sample scores twice as high as Western samples on the classic Levenson (1981) measure of fatalism, whether individuals think chance, luck or fate affects their outcomes. On a scale capturing perceptions of the causes of poverty (Feagin, 1975), our sample are more likely to attribute poverty to structural or fatalistic explanations than samples in western countries (Hunt, 2004). In our sample, both measures of fatalistic beliefs are higher among poorer people.

3.3 Economic conceptual framework

A simple economic model of aspirations: In Appendix Section B, we provide one plausible way to describe the role of aspirations in economic choices formally, via an extension to standard intertemporal decision-making models. The core idea is that goals and aspirations regulate behaviour, so dampened or limited aspirations can reduce motivation for effort and investment in the future. In line with other models of aspiration failures and their consequences (Dalton, Ghosal, and Mani, 2016; Genicot and Ray, 2017, 2020), we introduce a behavioural constraint through the presence of a reference point as a goal that incentivises investment and effort in an intertemporal model with leisure and consumption (see also Kőszegi, 2010). Behaviour is affected by aspirations as captured by the reference point: effort and investment increases in the level of aspirations.¹²

The experiment within the conceptual framework: One influence on aspirations is exposure to experiences of similar others. Our experiment actively tries to shift aspirations by providing one such exposure, to role models, drawn from similar communities, who through their own agency managed to create a different future. We aim to promote positive deviance toward change (as in Durkheim, 1893) but, by using local role models, we encourage avenues that are achievable and acceptable within local society.¹³

13. We aim also to avoid negative outcomes from boosting aspirations to unrealistic levels, as discussed in

^{10.} In the traditional Waaqeffannaa religion historically followed by many Oromo (the largest ethnic group in the study area), individuals are encouraged to uphold Safuu, the moral and ethical order, which governs relationships between humans, nature, and God (Waaqa). This religion has blended since early antiquity with ideas of predestination and divine will in local variants of Orthodox Christianity (*Fekade Egziabher*) and Islam (*Qadar*) (Gemeda and Rajani, 2018).

^{11.} Similarly, Levine (1965) describes strict engrained norms and customs which limited space for modernisation and new ideas in another part of Ethiopia, in rural Amhara.

^{12.} Our conceptual framework remains agnostic about where aspirations and demotivating beliefs stem from, although the psychological and anthropological literature above highlights the role of limited social exposure to alternative futures, societal and religious influences, and poverty.

In Appendix Section B, we model an increase in aspirations as a shift in the reference point, leading to immediate increases in effort and investment. It also could be seen as a shift in the mental model, as changing the way the world is perceived, offering a new point of reference of what one wishes to become or achieve, and possibly changing behaviour (Hoff and Stiglitz, 2016). Any changes in effort, investment and other outcomes from the experiment are then expected to occur through changes in aspirations, and not alternatives such as through time or risk preferences, or information transmission. This can be tested empirically.

Change dynamics: The extent and speed of changes in aspirations and outcomes will also be an empirical issue. We would not expect these changes to be large: the underlying resource constraint of these households, their social context, and its prevailing norms and values have not changed. Changes in aspirations may not be immediate. People may need to experiment with possible alternative futures (as alluded to by Appadurai, 2004) or may gradually learn-by-doing when demotivating beliefs start to shift, as in Carvalho et al. (2023). If small initial changes in aspirations lead to small changes in wealth, these wealth increases may in turn increase aspirations (Dalton, Ghosal, and Mani, 2016; Kahneman, Knetsch, and Thaler, 1986). Learning from others through spillovers may also occur (Genicot and Ray, 2020).

Plausible outcomes of the experiment: We would expect treated households to put in more effort and investment to build a better future, even though they have no new financial resources to do so. We therefore measure whether the intervention changed plausible investments in this context. In agriculture, this could include working longer hours on the farm, adopting modern farm inputs such as seed, fertiliser, or animal feed or other veterinary supplies, or diversifying more into owning livestock. Investment in education can also be seen as an investment, in old age security (Nugent, 1985, Rossi and Godard, 2022, and specifically for Ethiopia, Woldehanna et al., 2008 and Crivello and van der Gaag, 2016).¹⁴ Table 1 showed that better-off farm households in this remote context have higher levels of all these investments.

We would expect people to be better off from these actions. However, how that would manifest itself is less self-evident. We measure consumption expenditure, purchase of assets such as durables and tools, housing quality and subjective well-being. We would expect that well-being would go up. Whether consumption would go up is more ambiguous *a priori*. It could be that perceived well-being goes up without higher consumption, as households continue to move resources into the future, either by spending on assets such as durable goods or housing, that offer a flow of consumption into the future, or via productive investments such as in tools, livestock or agricultural productivity. Formally, in the model in Appendix Section B, observed consumption in a particular period depends on the intertemporal substitution elasticity, as is standard when incentives to move resources to the future are increased.

Appadurai (2004), Genicot and Ray (2020) and McKenzie, Mohpal, and Yang (2022).

^{14.} To quote one father "If you study well, you can improve. I want [my children] to reach a better place through their education. If they [do], they can support me" (Crivello and van der Gaag, 2016, 18).

| | (1) | (2) | (3) | (4) | (5) | (6) | | |
|---|---------------|------------------|-------------------|------------------|-----------------|--------------|--|--|
| | Whole sample | Lower tercile | Middle tercile | Upper tercile | <i>p</i> -value | Observations | | |
| Aspirations: what would you like | e to achieve? | | | | | | | |
| Aspired income (USD) PPP | 22382.36 | 18012.02 | 21106.74 | 27358.05 | 0.00 | 2017 | | |
| Aspired wealth (USD) PPP | 12816.36 | 9143.45 | 11319.29 | 17365.90 | 0.00 | 2025 | | |
| Aspired education (years) | 14.06 | 13.90 | 13.88 | 14.39 | 0.00 | 2000 | | |
| Expectations: what do you think you will achieve? | | | | | | | | |
| Expected income (USD) PPP | 5079.67 | 4507.98 | 4695.10 | 5962.70 | 0.00 | 2014 | | |
| Expected wealth (USD) PPP | 5034.33 | 3981.22 | 4687.73 | 6084.35 | 0.00 | 2011 | | |
| Expected education (years) | 13.49 | 13.30 | 13.28 | 13.86 | 0.00 | 1926 | | |

Table 2: ASPIRATIONS BY TERCILES OF DURABLE ASSETS

Notes: Column 1 is the sample mean. Columns 2-4 are sample means, conditional on the household being in the lower, middle and upper terciles of the value of durable assets (excluding tools) at baseline, an approximation for living standards. Columns 5 reports the *p*-value from a *t*-test of equality between the mean of the lower and upper tercile. Columns 6 reports the number of observations. Variables are measured at baseline for both the household head and spouse. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 99^{th} percentile. Variables are defined in detail in Appendix E. To measure aspirations, respondents are asked the levels of outcomes they would like their household to achieve, on three dimensions. Annual income is the amount of cash income the household earns from all agricultural and non-agricultural activities in a year. Wealth is durable wealth (including housing, vehicles, furniture and other valuable durables). Aspired education is measured as the 'years of education that you would like your oldest child to achieve'. Expectations are measured as the levels the respondent expects to reach in ten years, on the same dimensions. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using the national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1.

3.4 Measuring aspirations in rural Ethiopia

From concept to measurement: To assess whether the experiment boosted aspirations, we must quantify levels of aspirations using measures that can be compared across respondents. To do so, we developed and validated survey measures of aspirations (Bernard and Taffesse, 2014). These capture the level of income, assets, or children's education individuals hoped to achieve in their lifetime. To measure each dimension of aspirations, respondents were asked "*What is the level of [X] that you would like to achieve?*" where [X] was: (i) annual income (from all agricultural and non-agricultural activities, or social protection programmes); (ii) value of assets (including house, furniture, consumer goods like a TV and fridge, and any vehicles); or (iii) oldest child's education. We select these domains because they are relevant to most respondents and many described an improved life in terms of advancements in these areas.¹⁵

Reported aspirations: On these measures, respondents aspire to improve their current position over their lifetime (Table 2). Although the sample mean aspirations appear high, they may be reasonable given Ethiopia's rapid economic growth during the study period, discussed above. In 2020, national GDP per capita implies that a family of 5.6, the average in our

^{15.} To help respondents conceptualise the level they aspired to, they were first asked "What is the level of [X] you have at present?". Respondents are able to estimate current assets and income. Their responses to our single questions are similar to our main survey measures of their current assets and income, which we construct from their item-by-item survey responses.

sample, would have an annual household income of \$14,522 in 2016 USD PPP. Households aspire to wealth levels over their lifetime just below that, and to have 1.5 times that national income level.¹⁶ For the level of children's education, 60 per cent of respondents aspire for their oldest child to go beyond secondary school, with the remainder expecting them to complete secondary school or less. Are these realistic levels? That is harder to assess. Fruttero, Muller, and Calvo-González (2025) found in a recent review of aspirations studies that asking these questions in an "unconstrained" way as "desired" levels tends to get relatively high answers (see also La Ferrara, 2019). In line with this, we find that positive outliers strongly influence these large mean values. Indeed, once we look at median levels, a plausibly more realistic picture emerges: aspirations for household wealth and income are both \$4,488 in 2016 USD PPP. This is roughly a third of average household income nationally, calculated from GDP per capita. Median aspirations gaps (aspiration level minus current level) are 55 per cent and 58 per cent of these median wealth and income aspirations respectively: in short, these aspirations seem reasonable, and not too distant from current levels.

Aspirations versus expectations: As a further sense check, we compare aspirations to measures of expectations and of the maximum people believe someone in their village could earn. We directly elicit respondents' expectations in the same domains, asking what they think they will reach in ten years instead of what they would like to achieve over their lifetime. As expected, expectations are lower than aspirations in all domains: average income (wealth) expectations are roughly one quarter (one third) of aspirations, while respondents aspire to their child having at least a year more of education than they expect. Measures are correlated ($\rho = 0.42$ in the indices at baseline). As intended, our measures of aspirations conceptually capture something close to the upper limits of outcomes respondents believe are possible. Respondents' aspirations exceed but remain close to their beliefs about the highest attainable outcomes someone in their village could have at the time of the survey, further suggesting aspirations are somewhat reasonable: the median ratio of own aspirations to this village maximum is 1.3 for assets and 2 for income. Thus our measures are related in sensible ways to more standard economic measures of expectations, for respondents themselves and for others.

Reliability tests: In Bernard and Taffesse (2014), we conducted a measurement study to establish reliability of measures. We repeated measurement of income, wealth and education aspirations in our pilot setting with the same respondents two weeks apart. Measures had correlation coefficient ranging from 0.77-0.98 within-respondent, provided experienced enumerators were used, which we subsequently did in this study. These reliability correlations are in line with Angrist and Krueger (1999)'s benchmarks for reliable income and education measures.

Aspirations and their correlates: Our aspirations measure also correlates in expected ways with other measures in our sample. In line with anthropological accounts of aspirations, aspirations levels are strongly correlated with wealth. Aspirations for income, wealth and edu-

^{16.} As a benchmark, a large lump sum cash transfer from the NGO GiveDirectly increases annual household consumption by 75 per cent.

cation are all significantly higher for relatively better off households (Table 2, Column 5). The mean income (asset) aspiration in the upper tercile of households by wealth is 1.5 (1.9) times that in the lower tercile.¹⁷ Also in line with theory, opportunities which might increase aspirations are correlated with wealth: education, father's education, living outside the district and listening to the radio are correlated with wealth. Unsurprisingly, a simple regression finds these characteristics all predict aspirations. These associations suggest, but do not prove, that aspirations are related to exposure to opportunities and role models. For this we turn to our causal design.

4 Experimental design and estimation strategy

4.1 Content of the video intervention

In psychology, social learning theory suggests that people often adjust their goals or aspirations through "vicarious experience": impactful moments that allow them to imagine what it might be like to live someone else's life (Bandura, 1977). These experiences frequently arise when engaging with stories, which foster a sense of identification between the audience and the characters. The audience is "transported" into the narrative, envisioning themselves as the character they observe (Cohen, 2001, p. 251) (see also Green and Brock, 2000).¹⁸ Role models are most inspiring when their success feels attainable to the audience, because role models are similar to them, and when role models demonstrate the steps to achieve success (Lockwood and Kunda, 1997; Marx and Ko, 2012).

Our intervention consisted of inviting randomly selected households to a screening session to watch four short documentaries narrating motivational life stories of real people, from a similar socio-economic background to the study participants. We worked with an Ethiopian production company to produce the films. The role models in the films improved their economic circumstances through hard work and by setting, working towards, and achieving goals. Each documentary is 15 minutes long and in Oromiffa, the language in the study site. Two stories are about male and two about female role models. Appendix Section C summarises two documentaries, the stories of Teyiba Abdella and Bashir Malim Yisak, and one placebo segment.¹⁹

The documentaries had four common themes intended to make audience members reevaluate their own aspirations. First, they emphasise the importance of working hard. For example, Teyiba describes the start of her flour-trading business by saying "In those days I used to work like a donkey...carrying heavy loads on my back." Second, the role models highlight the importance of setting goals and planning, emphasising that progress is slow and they had to persevere through setbacks. Beshir describes how he made detailed plans over three years to save for and purchase his water pump. Both he and Teyiba describe future

^{17.} We find similar differences across wealth terciles in the "gap" between aspirations and the current level reported in each of the dimensions.

^{18.} In economics, La Ferrara (2016) and Mani and Riley (2021) summarise recent examples of video-based narratives that aim to shift behaviour.

^{19.} The documentaries, with English subtitles, and one of four placebo segments are available at ht-tps://www.youtube.com/channel/UCqfoNjCzt8YPjTRWQaMQfAg.

goals they are working towards, such as building a house or moving to the nearby town. The films vividly document the emotions accompanying both setbacks and success. Third, depicted individuals take actions which are possible for the audience and relatively normal for the villages where documentaries were screened, such as installing irrigation, or starting a shop. The characters succeeded largely through their own efforts, with some assistance from community members. For example, Beshir receives help from the local extension agent in purchasing a pump from Addis Ababa, although he saves and pays for it. But no role model relied on external support that would not be available to the audience. Any concrete information in the videos was unlikely to be new for viewers, although the documentaries may have made existing information more salient, which we test for in Section 5.2.3. Fourth, all the subjects take slightly different, "deviant", courses of action to those around them: starting a small business, diversifying their source of income, or improving their farming practices.

We selected the subjects of the documentaries by inviting agricultural extension agents and NGO staff in other districts in the region to submit descriptions of life stories of potential role models who lived in their area.²⁰ All the subjects were ordinary rural residents, very similar to their audience, who were either initially poorer than those around them or of similar socio-economic status, so their achievements would seem attainable to our sample. In the six-month follow-up, 52 per cent of audience members thought the documentary subjects had initially been worse off than they currently were. However, 73 per cent said that the documentary subjects eventually became better off than they were currently.

At the time of the screening, being shown a video in a group may have been a rare event in these villagers' lives. To account for potential changes as a result of the screening alone, we also invited another group of households, in the same villages, to a "placebo" screening of an Ethiopian comedy TV show about rural life. The placebo consisted of four 15-minute segments of the comedy TV show that we selected for its entertainment value only.

4.2 Randomisation and compliance

We randomly allocate the 18 randomly selected households in each village to treatment, placebo or within-village control groups. Our main comparison, of the treatment households with the control group, identifies the effect of the aspirations intervention. Comparing the placebo to the control group examines if there is any effect of exposure to video media and outside facilitators if the media content doesn't target aspirations. We find little evidence of placebo effects.

Treatment was assigned at the household level. Both the household head and spouse received the same treatment and were invited to the same videos.²¹ At the end of their baseline interview, each household head and spouse in treatment and placebo households received

^{20.} It was almost impossible that respondents would know anyone in the videos and there is no evidence that this happened.

^{21.95} individuals were single or widowed so the household was only given one invitation.

tickets for a screening session at a particular date and time in a few days time. Respondents were told that the screening was an entertainment show, tickets were non-transferable, and they could only attend the screening at the time and place on the ticket.

Screenings occurred in farmers' training centres or schools. As described in Section 2.2, we had 16 screening venues, with each venue located to be close to four of the villages: farmers' training centres or schools are usually in between villages to ensure maximum accessibility. We conducted screenings for the treatment group from four villages at once, and then separately for the placebo group from those four villages. Screenings for treatment and placebo groups were conducted at the same screening venue on the same or adjacent days, with no fixed order – neither group consistently viewed the screening before the other.²² We used a projector and speakers connected to a power generator. On average, people walked 29 minutes to the screening venue, so people not invited to the screenings were unlikely to walk to the centre and were not allowed entry without a ticket. Rooms were shuttered to ensure higher quality screening, which further ensured that no-one could watch from outside.

At the screening venue, facilitators checked farmers' identity and the date and time of the ticket before allowing entry. Compliance with our household-level randomisation was high. Only 2 per cent of the surveyed individuals or households did not comply with treatment allocation, by either missing their screening or going to the wrong one (Appendix Table A.8). There are no differences in compliance rates across treatment and placebo groups.

The design assumes no spillovers from the treated or placebo groups to the control group. In Section 5.5, we describe an additional "pure control" sample of non-treated villages we use to test for spillover effects. We find no systematic evidence of large spillover effects, so focus on the within-village randomisation for our main results.

4.3 Balance and attrition

Appendix Tables A.9 and A.10 present balance tests for individual and household variables, for both demographic characteristics and outcome variables measured at baseline. We present means for the entire sample present at baseline, including attriters, and separately for participants still present after five years, excluding attriters.

In general, we find balance across groups. We run various omnibus tests following Kerwin, Rostom, and Sterck (2024) over all individual level variables, all household level variables, and all baseline variables in Tables A.9 and A.10. We cannot reject the null hypothesis that our sample is balanced across groups. Results are consistent at baseline and endline, possibly because attrition is low.

In variables captured at the individual level (Table A.9), the only imbalance is in the expectations index, at the 10 per cent level. In variables at the household level (Table A.10),

22. Scheduling minimised interaction between treatment and placebo groups by ensuring that one group was either completing a post-screening survey or had left before the other arrived at the venue. A separate team administered the same survey to the control group simultaneously at their homes.

there are imbalances in the number of children. There is balance in overall household size, but the composition of households across age cohorts differs slightly by treatment. Treated households have slightly more children aged 7–10 and 11–15 and slightly fewer aged 0–6 than the other groups; differences are statistically significant at the 5 per cent level, except for the youngest age group. This imbalance appears in other variables linked to the number of children: the number of children in school and total minutes in school and studying across all children. In addition, treated households are more likely to have a non-organic roof (differences across groups are significant at the 5 per cent level) and have slightly more livestock (significant at the 10 per cent level). Our main specifications and robustness checks, described in the next section, add the baseline value of the outcome to account for imbalances.

Attrition is low. Most variables are measured at household level; 96 per cent of households present at baseline are found again after five years, which is notably high compared to other long-run follow-ups of randomised control trials in development economics (Bouguen et al., 2019). Similarly, 94 per cent of baseline respondents are re-interviewed after five years.

Few observed demographic characteristics, if any, predict attrition across any follow-up rounds (Appendix Table A.11). No observed demographic characteristics predict a household attriting across any follow-up rounds. A joint *F*-test fails to reject the null that any covariate, on its own or interacted with treatment and placebo indicators, explains attrition. Those invited to the documentary screening are slightly more likely to respond in the five-year follow-up, but this effect is only significant at the 10 per cent in the specification without covariates.²³

4.4 Empirical strategy

Our main specification is:

(1)
$$y_{vi} = \delta T_{vi} + \rho P_{vi} + \theta y_{vi0} + X'_{vi0} \pi + \alpha_v + \varepsilon_{vi}$$

where y_{vi} is a household-level outcome for household *i* in village *v*, $T_{vi}=1$ if a household was invited to watch the documentary, $P_{vi}=1$ if they were invited to watch the placebo movie and the omitted category is within-village control households. α_v is a village-level fixed effect and X_{vi0} is a pre-specified vector of controls measured at baseline: the age, gender, marital status and highest school grade completed of the head of the household. We also control for the baseline value of the outcome y_{vi0} , when we collected data on it, making this an ANCOVA specification. We collect baseline values for all outcomes except input use and consumption expenditure. ε_{vi} denotes the heteroskedasticity-robust error term.²⁴ Continuous outcomes are trimmed above the 99th percentile.

For aspirations, respondents' time-use, beliefs and preferences, which we observe separately

^{23.} When estimating Lee (2009) bounds to assess robustness to differential attrition, our main results remain similar in magnitude and significance, assuming attrition is monotonic in potential outcomes (results available on request).

^{24.} Results are broadly robust to clustering standard errors at the village level.

for household heads and spouses, y_{vi} is an individual-level outcome, and we control for age, gender, marital status and education of each individual. Standard errors for these outcomes are clustered at the household level, the unit of randomisation.

We use Equation (1) to test our predictions. We examine if, after five years, our intervention increased effort and investment in productive activities and assets; or effort and investment in education. We then analyse how this impacts various dimensions of standard of living. We capture impacts on food insecurity; consumption expenditure; ownership of "non-productive" durables; housing; and subjective wellbeing. We test whether the intervention works via increasing aspirations or if alternative mechanisms, such as risk and time preferences, information transmission, beliefs about returns to one's own effort, or beliefs about the role of fate and chance in outcomes play a role. Finally, using the shorter six-month follow-up, we examine if some economic behaviours and mechanisms had already changed soon after intervention.

Multiple inference adjustments: We adjust naive p-values as follows. We group related variables within table panels. A table panel corresponds to a concept in our conceptual framework, such as labour effort, agricultural investment, productive assets, non-productive durables, housing assets, or consumption. We view the group of variables within a panel as testing the same underlying hypothesis. To correct for multiple testing, we use the Benjamini, Krieger, and Yekutieli (2006) resampling procedure to calculate sharpened q-values which correct p-values for multiple tests across outcomes within each panel.

To summarise impacts, we report impact estimates on standardised inverse-covarianceweighted indices (Anderson, 2008) constructed from all outcomes reported in the primary results tables. This results in five main indices, capturing agricultural investment, educational investment, welfare, aspirations, and expectations. Following Kling, Liebman, and Katz (2007), we also aggregate the standardised indices into a single omnibus index.

Prespecification and robustness tests: We pre-registered analysis for the five-year follow-up in February 2017, after data collection was completed but before analysis.²⁵ There was no preregistration for the analysis of six-month follow-up outcomes, reported on in Bernard et al. (2014), as this was conducted before pre-analysis plans (PAPs) were widely used in economics. However, we committed to our core hypothesis – that the role model videos could increase future-oriented investment via increasing aspirations – via the experimental design. Aspirations were the only measure we collected directly after the screening, as well as in later rounds. Working papers reviewing the theoretical literature and validating the aspirations survey measures were published in early 2011, both pointing to the experiment.²⁶

The PAP for the five-year follow-up commits to analysis of all outcomes in the six-month follow-up, plus additional ones. The hypotheses and outcome construction in this paper are

^{25.} See https://www.socialscienceregistry.org/trials/1483 for the trial registration.

^{26.} The earliest versions of these papers are at https://assets.publishing.service.gov.uk/media/ 57a08ac840f0b652dd0008d8/csae-wps-2011-03.pdf (February 2011) and https://essp.ifpri.info/ files/2011/04/ESSP_WP47_MeasuringAspirations.pdf (April 2011) respectively. The validation survey occurred in 2009.

mostly consistent with the PAP. In Appendix Section G we list a few deviations from the PAP and explain the reasons for them. Between Appendices G, H and J, we report all the main results as pre-specified. Broadly, results change little due to deviations from the PAP.

The first and main deviation from the PAP is that we report ANCOVA results for those variables for which we collected baseline values (i.e. all variables except input use and consumption). Our pre-specified main specification did not control for the baseline value of the outcome. The pre-specified approach increases variance but is consistent across all outcomes, given that for some variables we did not collect baseline data. We had pre-specified running an ANCOVA specification, but as a robustness check. Our main robustness tables in Appendix H show results are mostly robust whether or not an ANCOVA specification is used.

In these tables, we also report Equation (1) but allowing for selection of controls via postdouble-selection LASSO (PDS Lasso) (Belloni, Chernozhukov, and Hansen, 2014), where all variables in Tables A.9 and A.10 can be selected as controls. These variables include pre-specified controls X_{vi1} , other demographic variables, and baseline values of all outcomes. Any baseline variables that explain variation in a given outcome, including variables imbalanced at baseline, will be controlled for. In the text, we note a few places where results differ slightly between the ANCOVA, pre-specified and PDS Lasso specifications, but results are largely robust.

Relative to the PAP, we also make some adjustments to our multiple inference adjustment approach in line with current practice. We pre-specified the omnibus index, and the hypotheses related to the summary indices, but had not specified construction of the summary indices. We also make a few changes to our groupings of variables for multiple inference adjustment. We present the main results as pre-specified in Appendix Section G to show these changes do not affect our substantive conclusions. We discuss other minor changes relative to the PAP in Appendix Section G: we adopt a more standard trimming strategy, make small adjustments to six variables (noted in footnotes), and add seven variables (we note if variables are not pre-specified).

A final robustness check uses an alternative control group to test for spillovers, finding little evidence of these (see Section 5.5 and Appendix J).

5 Results

This section presents results five years after the intervention, based on the predictions from our conceptual framework. We then discuss results on a smaller subset of outcomes collected after six months and effects on psychological mechanisms which might explain effects.

All tables follow the same structure. They present estimates of the parameters in Equation (1): δ (Column 1), ρ (Column 2), a test for $(\delta - \rho) = 0$ (Column 3), and the mean of the dependent variable in the control group (Column 4). A significant treatment effect compared to the control group (Column 1) indicates that the intervention had an overall impact. A significant placebo effect (Column 2) indicates that attending a screening alone had an effect. We report robustness tests in Appendix Section H. We note in the main text where results differ in magnitude or significance in the robustness tests. Reported results are robust to

correction for multiple hypothesis testing unless noted.

5.1 Effects on economic outcomes five years after the screening

5.1.1 Effort, investment and productive assets

In Table 3, we show that the intervention improved effort and investment in productive activities in these communities, notably related to agriculture. We measure time allocation at individual level on a typical day in the most recent October, the time of harvest. After five years, treated household heads and spouses work significantly more than both the control and placebo group (top panel). The effect is equivalent to about 8.6 per cent of the control mean, or nearly half an hour a day per spouse. As most households have one female and one male adult member, this is roughly an hour for all adult members per day. The increased work effort comes at the expense of leisure, which decreases by 5.9 per cent or nearly 50 minutes per day per spouse.²⁷

The treatment increased investments in modern inputs, especially on the extensive margin (whether or not the household spent any resources on these inputs) (second panel, Table 3). Treated households are 10 percentage points more likely to have invested in modern agricultural inputs like improved seeds and inorganic fertilisers than the control group and 10 percentage points more likely to have invested in modern livestock inputs like feed or vet supplies. For modern agricultural inputs, this translates into a 22 per cent increase in overall spending (intensive margin) compared to the households in the control group, but we cannot reject the absence of difference with the placebo group. For livestock inputs, intensive margin effects are positive but not significant.²⁸

We find effects on labour used on the family farm consistent with our earlier results on effort. We measure days of hired and family labour used in the most recent long rains in the agricultural inputs module. Treated households report using more family labour days on agriculture than the control group, consistent with results from the time allocation module.²⁹ They are less likely to hire any non-family labour in crop cultivation activities relative to the control group at the extensive margin. There is no change in land area under cultivation, potentially because land is allocated by local authorities with no possibility to buy or sell land, while rental markets remain limited despite recent legal changes.³⁰

The third panel of Table 3 explores the intervention's impact on household productive asset holdings, defined as those that may be used in agriculture or businesses. Treated households

^{27.} Results are robust to alternative specifications (Appendix Table A.19).

^{28.} The outcomes in this panel were not measured at baseline so this is not an ANCOVA specification. The results remain robust when using PDS Lasso, except the difference in the share of households adopting modern crop inputs relative to the placebo group declines from 6 to 4 percentage points and moves from marginal significance (at the 10 per cent level) to not being statistically significant (Column 6, Appendix Table A.19). The extensive margin analysis was not pre-specified.

^{29.} However, this effect is not robust to multiple hypothesis testing, and the difference relative to the placebo is not significant. Results are positive but not significant in the PDS Lasso specification.

^{30.} For a discussion of the legal frameworks, see Deininger, Ali, and Alemu (2008). We do not display a pre-specified variable studying land area rented or sharecropped as there was very little variation. Only 14 households rented any land in and four households rented out any land at the five-year follow-up.

have higher values of productive tools compared to control and placebo households, significant relative to controls.³¹ The value of their livestock holdings are about 5 per cent higher than in the control and the placebo group, but the difference is not statistically significant.³²

We combine all outcomes from the first three panels of Table 3 into a single agricultural investment index. Treated households significantly increased this index of investment by 0.21 and 0.18 standard deviations relative to the control and placebo group, five years after exposure to the role models in the videos. Results on the index are robust across specifications. Overall, the results in Table 3 are consistent with our conceptual framework.

5.1.2 Educational investments

In poor settings like in this part of Ethiopia, investing in children as old-age security (Nugent, 1985) is still common (see Rossi and Godard, 2022, and Woldehanna et al., 2008 for a discussion on Ethiopia). It is plausible that households consider it part of their investment portfolio, especially for their long-term future, given the lack of pension options. We assess education-related investments and outcomes through enrolment and grade attainment, time in school and studying, and school-related expenses.

We examine school-related expenses at household level. For other education-related outcomes, we look at two cohorts of children. These cohorts were pre-specified and correspond to the primary and post-primary school-going ages at the time of our five-year follow-up (see Appendix Figure A.2 for a timeline). "Cohort 1" are aged 16 to 20 (post-primary school-going age) at the five-year follow-up and 11 to 15 (upper primary school-going age) at the time of the intervention. "Cohort 2" are aged 7 to 15 (primary school-going age) during the five-year follow-up and 2 to 10 during the intervention.³³ We study all households in the sample, including 71 households without children in this age range, to make sure our results are comparable with our other findings.

The intervention increased investment in children's education among children of postprimary school-going age at the five-year follow-up ("Cohort 1"). These children were all of school-going age at intervention. The first panel of Table 4 shows the treatment increases the number of children in a household aged 16 to 20 enrolled in school at endline by 35 per cent relative to the control group. In the control group, 0.17 children in this age group per household are enrolled, compared to 0.23 children per households in the treatment group. The treatment effect compared to the placebo group is not statistically significant, but of similar magnitude.³⁴ Children aged 16 to 20 in treated households spend, on average, 33 minutes

34. Results are mostly robust across the ANCOVA and pre-specified specifications, except the treatment vs

^{31.} Results are robust to alternative specifications (Appendix Table A.19).

^{32.} Results are similar but marginally significant without controlling for the baseline value of the outcome relative to the placebo, while coefficients shrink but remain positive in the PDS Lasso specification (Appendix Table A.19). We had pre-specified grouping these asset variables with other assets in Table 5 but now group them here to distinguish productive assets – consistent with future-oriented investment – from other non-productive durable assets.

^{33.} We pre-specified studying children from age 6, not age 7, but correct this to align with the age at which children start school.

more per day attending school than the control group (31 minutes relative to the placebo group) and 9 minutes more studying (although differences are only significant relative to the control group).³⁵ Most notably, there is an increase in education attainment in this group: in the control group, 0.07 children in this age group per household have completed primary school, compared to 0.16 children per households in the treatment group..³⁶ The increase in attainment is nearly a doubling relative to the placebo and control group, albeit from a very low base. At baseline, only 39 of our control group households, 7 per cent, report having children aged 11 to 15 who have completed upper primary school.

The second panel of Table 4 shows more suggestive effects on children of primary school-going age at the five-year follow-up ("Cohort 2"). Some, but not all, of these children were all of school-going age at intervention. There are no significant increases in enrolment. Primary education enrolment rates increased from 57 to 65 per cent in the control group between baseline and the five-year follow-up, partly due to local authorities tracking and enrolling primary-age children, so potentially the treatment had no additional effect. There are small increases in time spent in school (22-45 minutes) and studying (7-17 minutes), of similar magnitude to the older age groups.³⁷

In the third panel of Table 4, we show that treatment significantly increases schooling expenditures five years after the intervention. Schooling expenditures in the treatment group are 36 per cent higher than in the control group and 30 per cent higher than in the placebo group. This result is robust across specifications.

Overall, the treatment increases an index of all outcomes in Table 4 by 0.25 and 0.21 standard deviations relative to the placebo and control group respectively, five years after exposure to the video intervention. This is robust to the alternative specifications (Appendix Table A.18). This is consistent with the hypothesis that exposure to the videos led to higher investment, but in an even longer-term investment than in agriculture.

5.1.3 Consumption, durable goods and well-being

Table 5 shows the impact on indicators of the standard of living five years after the screenings. We find that after five years the intervention increased wealth in the form of consumer durables and housing, and improved some indicators of food insecurity and subjective wellbeing. It did not increase indicators of current food and basic non-food consumption as we measured them.

Treated households perceive themselves to be less at risk of hunger (top panel, Table 5).

35. Results on time in school are robust across all specifications; results on time studying are not robust in the PDS Lasso specification (Appendix Table A.20)

36. This result is robust across all specifications (Appendix Table A.20). This variable was not pre-specified.

37. Effects on minutes studying are significant relative to control in all specifications but smaller, not robust to multiple inference, and not significant relative to placebo. Effects on minutes in school are also positive but only significant in alternative specifications (Appendix Table A.20).

placebo effect on the number of children aged 16-20 in school moves from significance at the 10 per cent level in the pre-specified specification to not being statistically significant in the ANCOVA specification. Coefficients shrink slightly and lose significance in the PDS Lasso specification (Appendix Table A.20). Results on enrolment for both cohorts are also robust to an alternative specification at the individual household-member level using the probability of being enrolled as the outcome, with the individual's age as a control.

| After five years | (1) | (2) | (3) | (4) | |
|---|--------------------|-----------------|--------------------|----------------------|--|
| | Treatment | Placebo | Treat. vs. | Control mean (SD) | |
| | ireaument | 1 lacebo | placebo | Total obs. | |
| Household head and spouse's labour effort: | | | | 10001 0005. | |
| Daily minutes working | 28.01*** | 8.87 | 19.14** | 326.16 | |
| | (7.80) | (7.97) | (7.95) | (200.01) | |
| | [0.00]*** | [0.26] | [0.03]** | 1754 | |
| Daily minutes in leisure | -48.62*** | -30.06*** | -18.55* | 830.59 | |
| | (9.98) | (10.41) | (10.32) | (187.38) | |
| | [0.00]*** | [0.01]*** | $[0.07]^{*}$ | 1773 | |
| Agricultural investment: | | | | | |
| % with any spending on modern crop inputs | 0.10*** | 0.04 | 0.06* | 0.58 | |
| | (0.03) | (0.03) | (0.03) | (0.49) | |
| | $[0.01]^{**}$ | [0.58] | [0.29] | 1089 | |
| Spending on seed or fertiliser (USD) PPP | 7.33** | 3.80 | 3.53 | 33.49 | |
| | (3.07) | (3.32) | (3.31) | (43.54) | |
| | $[0.05]^{**}$ | [0.58] | [0.38] | 1078 | |
| % with any spending on feed or vet supplies | 0.10^{***} | -0.04 | 0.14^{***} | 0.45 | |
| | (0.03) | (0.03) | (0.03) | (0.50) | |
| | $[0.01]^{**}$ | [0.58] | $[0.00]^{***}$ | 1089 | |
| Spending on feed or vet supplies (USD) PPP | 2.68 | -1.84 | 4.52 | 29.30 | |
| | (4.81) | (4.81) | (4.63) | (70.92) | |
| | [0.66] | [0.91] | [0.38] | 1081 | |
| % with any spending on hired crop labour | -0.05** | -0.02 | -0.03 | 0.36 | |
| | (0.02) | (0.02) | (0.02) | (0.48) | |
| | $[0.05]^{**}$ | [0.58] | [0.38] | 1089 | |
| Spending on hired crop labour (USD) PPP | -1.30 | -4.97 | 3.67 | 54.16 | |
| | (5.45) | (5.51) | (5.42) | (93.01) | |
| | [0.81] | [0.59] | [0.50] | 1078 | |
| Value of family crop labour (USD) PPP | 33.33* | 1.27 | 32.06 | 387.81 | |
| | (19.73) | (19.39) | (20.08) | (258.03) | |
| | [0.15] | [0.95] | [0.29] | 1079 | |
| Area cultivated (nectares) | (0.02) | -0.01 | (0.02) | (0.30) | |
| | (0.02) | (0.02) | (0.02) | (0.30) | |
| Assets | [0.03] | [0.91] | [0.38] | 1071 | |
| Value of line to ale (UCD) DDD | 110.75 | 02.04 | 90 51 | 2010 22 | |
| value of livestock (USD) PPP | (122.07) | 23.24 | 89.01 (114.99) | 2018.22 | |
| | (122.97) [0.26] | [0.84] | (114.20) [0.51] | (1921.09) 1052 | |
| Value of tools (USD) PPP | [U.30] 91 97** | [0.04] 19.77 | [U.31] 8 50 | 106 09 | |
| | (10.77) | (12.0) | (12.87) | (126.02) | |
| | (10.77) [0.10]* | [0.60] | (12.07) [0.51] | (120.90) | |
| Summary index: | [0.10] | [0.00] | [0:01] | 1040 | |
| Agricultural investment index | 0.21*** | 0.03 | 0.18*** | -0.00 | |
| | (0.06) | (0.06) | (0.06) | (1.00) | |
| | [0.00]*** | [0.96] | [0.01]*** | 1082 | |

Table 3: EFFORT, INVESTMENT AND PRODUCTIVE ASSETS

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences between parameters reported in the first two columns. The comparison group is households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. Standard errors are in parentheses. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each group (panel) of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. All columns control for village fixed effects and characteristics of the household head (or the individual respondent for effort outcomes): age, years of education, an indicator for being single, and an indicator for being male. Regressions control for baseline outcomes, except for agricultural investment. Variables are defined in Appendix E. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous outcomes above the 99th percentile. For the first panel, the unit of observation is the individual household head or spouse for effort. Time in work and leisure is reported for a typical day in October by the household head and expressed per adult. For the bottom three panels, the unit of observation is the household. Heteroskedasticity-robust standard errors are reported, except for effort where they are clustered at household level. Crop inputs include seeds, fertilisers, and pesticides. Livestock inputs include animal feed and veterinary supplies. Spending on family crop labour and hired labour is the product of the average village daily wage and the number of person-days of family or hired labourers in the most recent long rains season, respectively. The agricultural investment index is a weighted average of all these outcomes, with leisure time re-coded as negative, following Anderson (2008). The q-values for the agricultural investment index are calculated across all other summary indices reported in Table 7.

| After five years | (1) | (2) | (3) | (4) |
|---|----------------|-------------|-----------------------|------------------------------------|
| | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. |
| Cohort 1: Children of post-primary school-going | g age at end | lline (aged | l 11–15 at | the time of the intervention) |
| Children aged 16-20 in school | 0.06^{*} | 0.00 | 0.06 | 0.17 |
| | (0.03) | (0.03) | (0.04) | (0.41) |
| | $[0.08]^*$ | [0.94] | [0.13] | 1078 |
| Daily minutes in school for children aged 16-20 | 33.08** | 2.40 | 30.69^{**} | 58.64 |
| | (13.48) | (11.37) | (13.82) | (149.88) |
| | $[0.03]^{**}$ | [0.94] | $[0.05]^*$ | 1070 |
| Daily minutes studying for children aged 16-20 | 9.25^{**} | 1.69 | 7.56 | 17.82 |
| | (4.68) | (4.31) | (5.05) | (52.12) |
| | $[0.06]^*$ | [0.94] | [0.13] | 1063 |
| Children aged 16-20 that attained 8th grade | 0.09*** | 0.01 | 0.07^{**} | 0.07 |
| | (0.03) | (0.02) | (0.03) | (0.26) |
| | $[0.00]^{***}$ | [0.94] | $[0.04]^{**}$ | 1078 |
| Cohort 2: Children of primary school-going age | $at \ endline$ | (aged 2–1 | 0 at the time | me of the intervention) |
| Children aged 7-15 in school | 0.03 | -0.02 | 0.05 | 1.22 |
| | (0.08) | (0.08) | (0.08) | (1.18) |
| | [0.67] | [0.82] | [0.52] | 1078 |
| Daily minutes in school for children aged 7-15 | 23.47 | -13.63 | 37.11 | 527.12 |
| | (30.16) | (29.03) | (30.22) | (437.21) |
| | [0.65] | [0.82] | [0.52] | 1060 |
| Daily minutes studying for children aged 7-15 | 17.34^{*} | 9.78 | 7.56 | 91.29 |
| | (8.91) | (8.56) | (9.14) | (115.61) |
| | [0.16] | [0.76] | [0.52] | 1061 |
| For all children | | | | |
| Schooling expenditure (USD) PPP | 6.97** | 1.17 | 5.81** | 19.17 |
| | (2.84) | (2.53) | (2.95) | (32.73) |
| | $[0.01]^{**}$ | [0.64] | $[0.05]^{**}$ | 1065 |
| Summary index: | | | | |
| Educational investment index | 0.25*** | 0.04 | 0.21*** | 0.00 |
| | (0.07) | (0.07) | (0.08) | (1.00) |
| | [0.00]*** | [0.96] | $[0.01]^{***}$ | 1082 |

Table 4: Educational investments

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences between Columns 1 and 2. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets and are calculated over each group (panel) of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. The unit of observation is the household. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 99^{th} percentile. All columns control for village fixed effects and characteristics of the household head: age, years of education, an indicator for being single, and an indicator for being male. Heteroskedasticity-robust standard errors are in parentheses. Regressions control for baseline outcomes. We label "Cohort 1" those children aged 16 to 20 at the five-year follow up, who were 11 to 15 at the time of the intervention. We label "Cohort 2" those children aged 7 to 15 at the time of the five-year follow-up, who were aged 2 to 10 at the time of the intervention, so some but not all were of school-going age. We examine all households in the sample, including 71 households who have no children in this age group in any of the rounds, to ensure the sample is comparable with other results. Variables are defined in Appendix E. Daily minutes of an activity are the sum of schooling-age household members' daily minutes. School expenditures is the amount spent on uniforms, stationery and books, textbooks, and donations to the school for the whole household. The educational investment index is an inverse-covariance-weighted average of all outcomes reported in the table, following Anderson (2008). The q-values for the educational investment index are calculated across all other summary indices reported in Table 7.

Households were asked how many months in the past year they had difficulty satisfying food needs. Treated households have had fewer of these periods: they faced 0.33 and 0.38 fewer months in the previous year with difficulty satisfying food needs relative to the control and placebo groups, respectively. The control group faced 2.71 months with these difficulties, reflecting high levels of poverty in the area.³⁸ However, there is no difference between groups on a qualitative scale, the United States Department of Agriculture's food insecurity questionnaire (Bickel et al., 2000), capturing, for example, how frequently households skip meals or run out of money to buy food. This scale may be less suited to households who consume largely from own production.³⁹

There are few effects on food or frequent non-food consumption and marginal increases in non-food infrequent expenses, such as on clothing, services or ceremonies (second panel, Table 5). Treated households also reported higher values for a self-reported measure of general economic position relative to both the control and placebo group, though this increase is only significant at the 10 per cent level.⁴⁰

As was discussed in Section 3.3 or more formally in Appendix B, how consumption is affected by the intervention, even after five years, depends on individuals' preferences. We predict that the intervention increases lifetime wealth. If income effects dominate, this might increase current consumption. On the other hand, treated individuals may continue to move spending to the future (via savings, or productive investments of various forms) if (intertemporal) substitution effects dominate, reducing current consumption. Hence, the effects on consumption of the intervention are theoretically ambiguous; the findings are consistent with the substitution effect balancing out the income effect, at least in the time frame considered and for our measures of food and frequent non-food consumption. But, importantly, none of these measures include any estimate of the service flow value from consumer durables or housing. These goods may have been accumulated since the intervention and are likely to have a higher income elasticity, where an income effect may dominate.

In fact, treated households report a higher stock of consumer durables such as furniture, kitchenware or phones, aggregated in our results as durable assets (third panel, Table 5). They report 29 per cent higher value of these assets than the control group and 32 per cent higher than the placebo group, suggestive of more spending on goods with a higher income elasticity and therefore a perceived lifetime income effect. Treated households have invested more in the quality of their housing: they report an increase in the estimated value of their house (measured as the cost of rebuilding it, in materials and labour) that is 27 and 21 per cent higher than the control and the placebo groups respectively.⁴¹ This result is consistent with direct observations

^{38.} At baseline, we measured the number of months a household relied solely on its own food production, rather than months of food insecurity. We use this measure as a baseline control.

^{39.} Results are robust in all alternative specifications (Appendix Table A.21).

^{40.} Variables in this panel were not measured at baseline, but results are robust in the PDS Lasso specification (Appendix Table A.21).

^{41.} These effects on durables and housing are robust to alternative specifications, except that in the PDS

| After five years | (1) | (2) | (3) | (4) |
|--|----------------------|----------------|----------------------|-------------------|
| | - | D 1 1 | Treat. vs. | Control mean |
| | Treatment | Placebo | placebo | (SD) Tatal aha |
| Food security: | | | | Total obs. |
| Food security index: z-score | -0.07 | -0.10 | 0.03 | 0.48 |
| | (0.06) | (0.06) | (0.06) | (0.92) |
| | [0.28] | [0.22] | [0.61] | 1076 |
| Months of food insecurity | -0.33** | 0.04 | -0.38*** | 2.71 |
| | (0.15) | (0.15) | (0.14) | (2.13) |
| | $[0.05]^*$ | [0.77] | $[0.02]^{**}$ | 1038 |
| Consumption: | | | | |
| Food consumption (USD) per ad. equiv. monthly PPP | -1.98 | -2.29 | 0.32 | 53.91 |
| | (2.05) | (1.92) | (2.07) | (29.98) |
| | [0.33] | [0.58] | [0.88] | 1076 |
| Frequent non-food (Im recall USD) per ad. equiv. PPP | 0.44 | (0.04) | (0.40) | 4.08 |
| | (0.28) | (0.28) | (0.30) | (3.09) |
| Nonfood consumption (12m recall USD) per ad equiv monthly PPD | 0.19 | [0.94] 0.54 | [0.20] 1.94** | 1070 |
| Noniood consumption (12m recan 65D) per ad. equiv. monenty 111 | (0.51) | (0.43) | (0.48) | (6.35) |
| | [0.01] | [0.40] | [0.05]* | 1079 |
| Consumption of cigarettes and tobacco (USD) PPP | 0.20 | 0.04 | 0.17 | 0.80 |
| r Guinna (12) | (0.13) | (0.12) | (0.13) | (1.66) |
| | [0.19] | [0.94] | [0.25] | 1078 |
| General economic position (scale 1 to 4) | 0.09* | 0.00 | 0.09* | 2.10 |
| | (0.05) | (0.05) | (0.05) | (0.73) |
| | [0.19] | [0.94] | [0.21] | 1088 |
| Non-productive durables and housing: | | | | |
| Value of durable assets excluding tools (USD) PPP | 20.71^{**} | -1.65 | 22.36** | 70.55 |
| | (9.97) | (8.77) | (10.17) | (127.39) |
| | [0.08]* | [0.85] | [0.06]* | 1049 |
| Value of house (USD) PPP | 384.94*** | 83.72 | 301.22**** | 1384.27 |
| | (90.70) [0.00]*** | (81.93) | (89.10) [0.00]*** | (1235.57) |
| Non organia roof | 0.00 | 0.03 | 0.00 | 1020 |
| Non-organic roor | (0.01) | (0.03) | (0.03) | (0.47) |
| | [0.59] | [0.43] | [0.64] | 1036 |
| Own toilet facility | 0.07^{*} | 0.04 | 0.02 | 0.38 |
| 5 | (0.04) | (0.03) | (0.04) | (0.49) |
| | $[0.08]^*$ | [0.43] | [0.64] | 1039 |
| Wellbeing: | | | | |
| Best life | 0.22^{**} | 0.10 | 0.12 | 4.83 |
| | (0.11) | (0.11) | (0.11) | (1.80) |
| | $[0.09]^*$ | [0.40] | [0.58] | 1901 |
| Happiest life | 0.12 | 0.12 | 0.00 | 6.05 |
| | (0.14) | (0.14) | (0.14) | (2.19) |
| Summany inder. | [0.38] | [0.40] | [0.98] | 1885 |
| | | | | |
| Welfare index | 0.15^{**} | 0.01 | 0.14^{**} | -0.00 |
| | (0.07) | (0.07) | (0.07) | (1.00) |
| | [0.03] | [0.96] | $[0.04]^{-1}$ | 1090 |

Table 5: CONSUMPTION, DURABLE GOODS AND WELL-BEING

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences between Columns 1 and 2. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each group (panel) of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. The unit of observation is the household, except for subjective well-being, which are for both household head and their spouse. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 99^{th} percentile. All columns control for village fixed effects and characteristics of the household head (or of the individual-respondent for the wellbeing outcomes): age, years of education, an indicator for being single, and an indicator for being male. Heteroskedasticity-robust standard errors are in parentheses (for the wellbeing outcomes heteroskedasticity-robust standard errors are clustered at the household-level). Regressions control for baseline outcomes, except for outcomes in the consumption panel that were not collected at baseline. Variables are defined in Appendix E. The food security index is a z-scored weighted sum of answers to the United States Department of Agriculture's food security questionnaire adapted for Ethiopia (Hadley et al., 2008). It indicates the degree of food security. Months of food insecurity is the number of months in the last 12 that the household had problems satisfying their food needs. Food consumption (7 day recall) and non-food consumption (30 day and 12 month recall) are reported per adult equivalent (PAE) and converted into monthly figures. Adult equivalents are constructed using the OECD scale. Non-food consumption with a 30 day recall is regular expenses e.g. for toiletries, transportation, mobile phone, energy, cigarettes and tobacco, repair, tailor, barber. Non-food consumption with a 12 month recall is irregular expenses e.g. for clothing, utensils, bedding, school expenses, health expenses, festivals, church and community contributions.General economic position is measured on a scale from 1 to 4, where 4 is that the household is "doing well and able to meet needs by own efforts" and 1 is that the household being "unable to meet their needs". Durable assets is the number of assets owned multiplied by the replacement value of each asset. Value of the house is how much their house would cost to build today. The roof variable is coded to reflect relative quality of the building materials and the sanitation facilities are coded to reflect the degree of privacy. Subjective well-being uses a (Cantril, 1966) ladder. Respondents are asked where they are on a ladder where 10 is the best and 1 the worst possible life, or where 10 is the happiest and 1 the least happy life. The welfare index is an inverse-covariance-weighted average of all outcomes reported above in the table, with months of food insecurity in the last year recoded to be negative, constructed following Anderson (2008). The welfare index averages over the household head's subjective well-being outcomes. The q-values for the welfare index are calculated across all other summary indices reported in Table 7.

by our enumerators: treated households are more likely to have been found to have their own toilet facility in all specifications than the control group, although differences are not significant relative to the placebo group. There are no effects on the probability of having a non-organic roof, also observed by enumerators, possibly because this is an expensive investment.

Treated households also score significantly higher on a Cantril ladder of self-reported wellbeing relative to the control group (fourth panel, Table 5). Treated participants score about a quarter of a step higher than control and placebo groups, although there is no significant effect when they are asked the same question in relation to happiness rather than life satisfaction.⁴²

In Appendix Table A.5, we report on measures of revenue generated from household economic activities. While these constructs are noisy, as in much work on subsistence farmers, we still find that total gross revenue in the treated group increased 11 per cent five years after intervention. The difference between treatment and placebo is significantly different from zero at 10 per cent, while the difference between treatment and control is not statistically significant.

Overall, these patterns suggest that treated households have (modestly) improved their standard of living, in addition to their effort and investments. The treatment effect on an index of all outcomes in Table 5 (combining the outcomes reported in the top four panels) is positive and statistically significant: the index is 0.15 and 0.14 standard deviations higher than the control and placebo group respectively, five years after exposure to the video intervention.⁴³

5.2 Where do these results come from?

5.2.1 Early impact

Our video intervention began to change household behaviour soon after the screenings. In Figure I, we report on a shorter survey, collected six months after the experiment, which shows some future-oriented behaviours had already changed.⁴⁴ To facilitate comparability, coefficients are standardised z-scores, computed by subtracting the control mean and dividing by the control standard deviation. Effects in the units of the outcomes are reported in Appendix Table A.6. The questionnaires are the same between rounds except where discussed below.

In the bottom left of Figure I, we show that treatment significantly increased the stock of savings relative to both the placebo and control groups soon after the intervention (see Appendix Table A.3 for the results in table form). Savings behaviour is a good short-run indicator of increased propensity to invest, as it is unlikely poor households could immediately

Lasso specification, the difference in the value of durable assets between treatment and placebo is smaller and marginally not significant and coefficients on housing quality shrink slightly in magnitude while remaining statistically significant (Appendix Table A.21). The housing value, non-organic roof and own toilet variables were not pre-specified.

^{42.} Effects are robust when not controlling for the baseline value of the outcome. Coefficients are still positive, but smaller and not significant, in the PDS Lasso specification (Appendix Table A.21).

^{43.} This increase is robust to removing the control for baseline outcomes, but somewhat smaller and marginally significant in the PDS Lasso specification only relative to the control group (Table A.18).

^{44.} Bernard et al. (2014) reported results from this short-run survey. This paper includes all main outcomes reported there and replaces that paper. Most outcomes are in Figure I, except hypothetical variables measuring potential demand for credit, reported in Appendix Table A.6.

make new asset purchases given limited resources. Only 39 per cent of the control group had savings, averaging 24.37 USD PPP. In the treatment group, the amount of savings nearly doubled in percentage terms six months after intervention, although there is little change in whether households have any savings.

Those effects do not persist in the long-run. This suggests households may have accumulated cash savings in the short run and invested them in relatively lumpy assets such as durables, livestock and housing, which we see in Table 3. Indeed, five years later, savings remain very modest for all groups, as compared to other assets. Control means at endline for savings are 18 USD in 2016 PPP versus livestock (2018 USD PPP), house (1384 USD PPP), durables (71 USD PPP) and tools (106 USD PPP). This suggests households ultimately seek not to hold savings in cash. There are few formal financial institutions through which they can earn returns and high inflation. They prefer to invest in assets.

We also observe an increase in loans in the short term, but this effect is smaller in percentage terms than effects on savings and also does not persist. This suggests households had repaid these loans by endline.

The treatment had already induced investments in education early on (right panel of Figure I). We first analyse Cohort 1 from Table 4: those aged 11 to 15 (upper primary school-going age) at the time of screening. We find suggestive evidence that households started investing more in their children's education six months after screening. We find a 13 per cent increase in the number of children aged 11 to 15 enrolled in school in the treatment group relative to the control group, though not statistically significant (p-value=0.13). There is a 17 per cent increase in the time spent studying (p-value=0.08) and an 8 per cent increase in time spent in school (p-value=0.32) relative to the control mean. This cohort eventually accumulates more years of education by the five-year follow-up.

We separately analyse the outcomes of children aged 7 to 10, of lower primary school-going age at the time of screening. This is a subset of Cohort 2 in the five-year follow-up in Table 4. Cohort 2 in Table 4 consists of children aged 2 to 10 at the time of screening. At the six-month follow-up we only collected educational outcomes for children of school-going age (7 and above). So those aged 2 to 6 at the time of the screening are missing education data at the six-month follow-up but are included in Cohort 2 in the five-year follow-up because they are of primary school age. In this younger cohort, we find a significant increase in enrolment relative to the control group, equivalent to about 18 per cent of the control number of children enrolled, and an increase in time spent in school (p-value=0.16) but no effect on time studying.

Treated households increased education spending after six months, with a 17 per cent increase in school-related household expenditures relative to the control group, but no significant difference relative to the placebo. This effect persists at endline.

We do not find strong evidence that households had already changed their labour supply decisions in response to the intervention. They had marginally increased time spent on the family farm. But the effects after six months are smaller than those reported in Table 3, so that differences are not significant relative to the control group but only relative to the placebo group. This is potentially because rounds cannot be perfectly compared. The time use data in the five-year follow-up survey were collected in relation to a day in October, the busy harvest season. The six-month follow-up was related to a day in March, when work on the farm is likely to be less intensive, largely land preparation for the rains in June, so we may be less likely to see strong effects at this time.⁴⁵

In sum, households had already made some changes in future-oriented behaviour after six months. We do not measure all variables we capture in the long-term survey in this shorter-term follow-up. But where we have similar variables in both rounds, we observe consistent patterns of behaviour change. Many of the initial changes households made at six months – in education investment and asset accumulation – persist and become more pronounced after five years.

5.2.2 Impact on aspirations and expectations

Next, we provide some evidence on the potential psychological mechanisms at play. We collected measures of aspirations and expectations for the household on income, wealth and children's education, from both spouses. We also collected measures of their reports of the household's current position on these dimensions, at baseline, straight after the screening, after six months and after five years. Figure II displays effects on indices of aspirations and expectations. Appendix Table A.1 reports results for indices and their individual components. Section 3.4 describes the measures.

Aspirations: Five years after the intervention, we observe statistically significant increases in the aspirations index – between 0.12 and 0.15 standard deviations – relative to both the placebo and control groups.⁴⁶ These are driven by increases in all dimensions of aspirations, though not all are individually significant (Appendix Table A.1). The magnitudes are modest: for education, the effect is half the baseline gap between the poorest and richest terciles; for income and wealth, the effects are around one-third and one-fifth of the corresponding gaps (Table 1). Crucially, effects stem from the role-model content in the documentaries, not from the screening itself: the placebo group shows no long-term changes, and differences between treatment and placebo are statistically significant, in all specifications.

We find smaller, less precise effects immediately after screening and at the six-month follow-up, although we cannot statistically reject that effects are the same in each round. This hints at some dynamic empirical pattern in the data. As alluded to in the conceptual framework in Section 3.3, aspirations may evolve gradually as people experiment with possible alternative futures (Appadurai, 2004) or learn-by-doing when demotivating beliefs start to shift (Carvalho et al., 2023). People may begin investing, observe returns, and revise their goals upward. This is consistent with the model in Dalton, Ghosal, and Mani (2016), where

^{45.} The five-year measure of labour supply also included off-farm self-employment, which we did not collect at the six-month follow-up.

^{46.} These are robust to alternative specifications in Appendix Table A.22, except that the aspirations effect relative to the control is positive but not significant in the PDS Lasso specification.

aspirations depend partly on wealth. Similarly, Orkin et al. (2023) find that aspirations of Kenyan households adapt and increase when wealth increases.

Aspirations gaps: Some models focus not on absolute aspiration levels but on the gap between aspirations and current status (Genicot and Ray, 2017). Our results using this "aspirations gap" – the difference between desired and current levels – are qualitatively similar (Appendix Table A.2).⁴⁷

Expectations: We can compare treatment effects on aspirations to effects on more standard measures of expectations in the same domains, asking what respondents think they will reach in ten years instead of what they would like to achieve over their lifetime. Empirically, as discussed in Section 3.4, levels of expectations are lower than aspirations for all rounds, but the measures are strongly correlated within respondents.

Effects on aspirations and expectations evolve in a similar way. Straight after screening, there are only small and noisy effects on expectations relative to the placebo.⁴⁸ Like aspirations, effects are small and positive after six months and large, positive and statistically significant after five years. We cannot reject that effects on aspirations and expectations indices relative to the placebo are the same in each round. This provides a sense check, showing that beliefs about the future change similarly to aspirations.⁴⁹

Overall, five-year effects on aspirations are stronger and more robust, but some impacts were visible soon after the intervention. These findings support our hypothesis: the intervention increased aspirations, which in turn influenced forward-looking behaviour – including effort and investment – with effects sustained over time.

5.2.3 Alternative mechanisms

Although our results support the aspirations hypothesis, we test alternative mechanisms that might be affected by our intervention and yield similar changes in economic behaviour. We consider three plausible alternative mechanisms, as well as social desirability bias caused by the study itself. Appendix Section E.3 details the construction of the measures used to test these alternative mechanisms.

Time and risk preferences: The intervention could have increased the discount factor (β in our model in Appendix Section B), leading to the observed increase in future-oriented behaviour, as the future is more valued.⁵⁰ But we do not find that time preferences have shifted (top panel, Table 6). Six months after intervention, there is no change in the share of patient, impatient, or very impatient respondents, categorised using the measure in Ashraf,

^{47.} For the current level of education, we use the respondents' own education level. Recall from section 3.4 that these beliefs about current position are strongly correlated with actual values reported in the survey.

^{48.} There are some effects relative to the control, but this is driven by an effect of the placebo. This disappears by six months and does not lead to any effects on behaviour).

^{49.} Our theoretical framework does not make clear predictions on behaviour of expectations, for reasons discussed in Appendix Section B.

^{50.} Alternatively, Gabaix and Laibson (2017) theorise that improving the extent to which households can visualise the future may lead to more patient behaviour.



Treatment
Difference (Treatment - Placebo)

Figure I: TREATMENT EFFECTS ON ECONOMIC OUTCOMES OBSERVED SIX MONTHS AND FIVE YEARS AFTER THE SCREENING

Notes: OLS estimates of within-village treatment effects on economic outcomes across survey rounds relative to the control and placebo groups. Outcomes are collected six months or five years after the screenings. The comparison group is households from the 64 treated villages that were not invited to any screening. The outcomes units are standardised z-scores, computed by subtracting the control mean and dividing by the control standard deviation. Effects in the outcome units are reported in Appendix Table A.6. All columns control for village fixed effects, the baseline value of the outcome, and characteristics of the respondent: age, years of education, an indicator for being single, and an indicator for being male. The unit of observation is the individual respondent (household head or their spouse) for the outcomes on the left panel, and the household for the outcomes in the right panel. Standard errors are robust to heteroskedasticity (or clustered at the household-level for the left panel). Bars are 95 per cent confidence intervals. Square-shaped markers report the difference between the treatment and placebo effects. Variables are defined in Appendix E. Time spent on work and leisure is collected for a typical day in March for the six-months follow-up, and for October for the five year survey for the head and spouse, reported by the household head and expressed per adult. Total savings is the amount saved both in and outside the home by each respondent. Credit amount records the total amount of loans taken from any formal or informal source in the past six months by each respondent. Cohort 1 are children who were 11 to 15 at the time of the screening and 16 to 20 at the five-year follow-up. Cohort 2a are children who were 7 to 10 at the time of the screenings, a subset of Cohort 2 in Table 4, children aged 7 to 15 at the time of the five-year follow-up. Cohort 2a is not directly comparable to cohort 2 in Table 4, because some of the children in cohort 2 were not of primary school-going age at the time of six-months follow-up and as they would have been between 2 and 6 years old; we did not collect data for children in this age range. We examine all households in the sample, including 71 households who have no children in this age group in any of the rounds, to ensure the sample is comparable with other results. Daily minutes of time spent studying or in school are the sum of schooling-age household members' daily minutes. School expenditures is the amount spent on uniforms, stationery and books, textbooks, and donations to the school for the whole household.



Figure II: TREATMENT EFFECTS ON THE ASPIRATIONS AND EXPECTATIONS INDICES. *Notes*: OLS estimates of within-village treatment effects on aspiration and expectations indices across survey rounds relative to the control and placebo groups. "After the screening" reports on data collected right after the screening. The remaining columns report on outcomes collected after six months and five years. The comparison group is households from the 64 treated villages that were not invited to any screening. All columns control for village fixed effects, the baseline value of the outcome, and characteristics of the respondent: age, years of education, an indicator for being single, and an indicator for being male. The unit of observation is the individual respondent. Heteroskedasticity-robust standard errors are clustered at the household level. Bars are 95 per cent confidence intervals. Square-shaped markers report the estimated difference between the treatment and placebo effects. The aspirations index is an Anderson (2008) index combining what individuals would like to achieve in their life in terms of reported income, wealth, and years of education for their eldest child. The expectations index similarly combines what individuals think they will achieve in ten years time on the same three dimensions. The aspirations and expectations aggregate Anderson (2008) index combines six dimensions of reported income, wealth and years of education for their eldest child, for aspirations and expectations. Appendix Table A.1 reports results across all three indices and their individual components.

Karlan, and Yin (2006), or of present-biased respondents. After five years, there is a small negative treatment effect on the share of time-inconsistent respondents that are patient now and impatient later, but this effect does not occur six months after intervention and is not robust to multiple hypothesis testing.

While risk does not enter explicitly the theoretical framework, it could be trivially extended. For example, if future returns are risky, then lower risk aversion would induce more effort and investment in the future. The increased salience of a plausible future through the intervention may have reduced risk aversion, leading to the observed effects. But we also find no changes in risk preferences, using a Binswanger (1980) adaptation (top panel, Table 6).⁵¹

Perceived returns to own effort and causes of success: The intervention may have

51. In fact, if anything, our risk aversion measure has increased relative to the placebo group after six months, although the effect is not robust to multiple test correction and does not persist after five years.

| | After six months | | | | After f | ive vears | | |
|---|------------------|----------|-----------------------|---------------------------|------------------|-----------|-----------------------|---------------------------|
| | (1) | (2) | (3) | (4) Control | (5) | (6) | (7) | (8) Control |
| | Treatment | Placebo | Treat. vs. placebo | mean (SD) Total obs | Treatment | Placebo | Treat. vs. placebo | mean (SD) Total obs |
| Risk and time preferences: | | | | 10000 0005. | | | | 10000 0005. |
| Bisk aversion: | 0.03 | -0.12 | 0.14* | 3 20 | 0.03 | -0.04 | 0.06 | 2.53 |
| most to least risk averse $(1 \text{ to } 5)$ | (0.08) | (0.08) | (0.08) | (1.51) | (0.09) | (0.09) | (0.09) | (1.54) |
| | [0.89] | [0.36] | [0.45] | 2025 | [0.79] | [0.83] | [0.51] | 1873 |
| % that is patient | -0.01 | -0.01 | 0.00 | 0.30 | -0.02 | -0.00 | -0.01 | 0.17 |
| * | (0.02) | (0.02) | (0.02) | (0.46) | (0.02) | (0.02) | (0.02) | (0.37) |
| | [0.89] | [0.67] | [0.96] | 2054 | [0.48] | [0.83] | [0.51] | 1899 |
| % that is somewhat impatient | -0.02 | -0.02 | 0.00 | 0.15 | -0.02 | -0.00 | -0.02 | 0.11 |
| | (0.02) | (0.02) | (0.02) | (0.36) | (0.02) | (0.02) | (0.02) | (0.31) |
| | [0.89] | [0.44] | [0.96] | 2054 | [0.43] | [0.83] | [0.51] | 1899 |
| % that is most impatient | 0.03 | 0.04 | -0.01 | 0.55 | 0.04 | 0.01 | 0.03 | 0.73 |
| | (0.03) | (0.03) | (0.03) | (0.50) | (0.03) | (0.02) | (0.02) | (0.45) |
| | [0.89] | [0.36] | [0.96] | 2054 | [0.37] | [0.83] | [0.51] | 1899 |
| % that is present biased | 0.02 | 0.04 | -0.02 | 0.34 | 0.03 | 0.05* | -0.02 | 0.53 |
| | (0.03) | (0.03) | (0.03) | (0.47) | (0.03) | (0.03) | (0.03) | (0.50) |
| | [0.89] | [0.36] | [0.96] | 1999 | [0.48] | [0.48] | [0.51] | 1872 |
| % that is patient now and impatient later | -0.00 | (0.00) | -0.00 | 0.22 | -0.05** | -0.02 | -0.03 | 0.19 |
| | (0.02) | (0.02) | (0.02) | (0.41) | (0.02) | (0.02) | (0.02) | (0.39) |
| Perceived returns of own effort | [0.98] | [0.98] | [0.96] | 1999 | [0.20] | [0.83] | [0.51] | 1872 |
| | 0.00** | 0.05 | 0.01** | 10.05 | 0.00 | 0.00 | 0.04 | 10.00 |
| Internal locus of control | 0.26^{**} | -0.05 | 0.31^{**} | 12.95 | (0.02) | 0.06 | -0.04 | 12.26 |
| | (0.12) | (0.12) | (0.12) | (2.10) | (0.11) | (0.11) | (0.11) | (1.90) |
| Crit index | [0.10] | 0.00 | 0.10 | 2055 | [0.80] | 0.05 | [0.94] | 1899 |
| Grit index | (0.04) | -0.00 | (0.10) | (1.00) | -0.05 (0.06) | -0.05 | (0.00) | (0.02) |
| | (0.00) [0.56] | (0.00) | (0.00) | (1.00) | (0.00) [0.96] | (0.00) | [0.00] | (0.99) |
| Individual causes of neverty | 0.00 | 0.22* | 0.10 | 2004 | [0.00] | 0.01 | 0.02 | 1909 |
| individual causes of poverty | (0.23) | (0.23) | (0.02) | 9.19 (9.41) | (0.03) | (0.13) | (0.03) | (2.03) |
| | (0.14) [0.11] | [0.14] | [0.14] | 2.41) | [0.14] | [0.13] | [0.13] | 1883 |
| Perceived external causes of success: | [0.11] | [0.20] | [0.01] | 2000 | [0.00] | [0.01] | [0.04] | 1000 |
| Change logue of control | 0.02 | 0.03 | 0.01 | 13.36 | 0.01 | 0.05 | 0.05 | 12.67 |
| Chance locus of control | (0.16) | (0.15) | (0.01) | (2.71) | (0.01) | (0.15) | (0.05) | (2.35) |
| | [0.10] | [0.13] | [0.10] | 2050 | [0.15] | [0.06] | (0.13) [0.74] | (2.55) |
| Fate causes of poverty | [0.32] | _0.07] | [0.35] _0.22 | 2000 7 10 | -0.03 | 0.00 | -0.03 | 6 70 |
| rate causes of poverty | (0.14) | (0.14) | (0.14) | (2.31) | (0.11) | (0.10) | (0.10) | (1.79) |
| | (0.14) [0.16] | [0.14] | [0.14] | 2039 | [0.96] | [0.10] | [0.10] | 1887 |
| Awareness and perceived returns of agricul | tural techno | ologies: | [0.22] | 2000 | [0.50] | [0.50] | [0.14] | 1001 |
| Information index | | | | | -0.02 | -0.07 | 0.04 | 0.00 |
| mormation muta | | | | | (0.02) | (0.07) | (0.04) | (1.00) |
| | | | | | [0.73] | [0.34] | [0.68] | 1104 |
| Expected fertiliser yields index | | | | | 0.08 | 0.11 | -0.03 | -0.00 |
| Inperior formier Jields index | | | | | (0.07) | (0.07) | (0.07) | (1.00) |
| | | | | | [0.52] | [0.25] | [0.68] | 1081 |

Table 6: TESTING MECHANISMS

Notes: OLS estimates of within-village treatment and placebo effects after six months (columns 1-2) and after five years (columns 5-6) of the intervention. Columns 3 and 7 test for differences in parameters obtained in previous two columns. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 and 8 display the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and respondent's age, years of education, an indicator for being single, and an indicator for being male. All regressions control for the baseline value of the outcome, except for the outcomes in the "Awareness and perceived returns of agricultural technologies panel" that were not collected at baseline. Heteroskedasticity-robust standard errors clustered at the household-level are in parentheses. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Variables are defined in Appendix E. The unit of observation is the individual respondent (household head or their spouse), except for information and fertiliser beliefs indices (which are at the household-level). The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous outcomes above the 99th percentile, though indices aggregate all non-missing outcomes. The locus of control variables are based on Levenson (1981) and capture if people see outcomes as contingent on their behaviour (internal locus of control) or as a result of chance, luck or fate (chance locus of control). The causes of poverty variables are based on the Attributions for Poverty scale (Feagin, 1975) to measure people's perceptions of the causes of poverty among people in general, rather than only in their own lives. We use survey-based instruments to calculate risk (Binswanger, 1980) and time preferences (Ashraf, Karlan, and Yin, 2006). Grit includes answers to two survey questions about how the respondent would characterise themselves. The information index is constructed from indicator variables that take value one if the household head reported having performed some of the behaviours that were described in the documentaries. Appendix Table A.4 reports treatment effects on the components of this index. The expected fertiliser yields index is based on the household heads' expected increase in output from the application of different quantities of fertiliser for maize and sorghum in an hypothetical good or bad season. These are standardised Anderson (2008) indices

altered people's broad beliefs about their ability to change their own outcomes, or, in economic terms, beliefs about the return to their own effort. Six months after the screening, we find small increases (of around 2 per cent) in internal locus of control (second panel, Table 6), and weak positive effects on a similar measure, the extent to which people believe poverty is an issue of individual agency.

However, there are no persistent effects on either measure of locus of control after five years. In the treatment group, both measures revert back to the same level as at baseline. This suggests locus of control does not account for long-term changes in investment. If this were the mechanism, we would expect short-term effects on investment as locus of control rises but then effects on investment fading away as effects on locus of control fade away. This is not the pattern we see. In contrast, aspirations and expectations and investment move together, increasing slightly in the short term, with larger effects after five years. We also find limited effects on two conceptually related measures, the extent to which people believe outcomes are caused by chance or that poverty is caused by fate.

We additionally consider an adapted measure of grit (perseverance and passion for long-term goals Duckworth et al., 2007; Alan, Boneva, and Ertac, 2019), which we had not pre-specified. These measures could broadly enter our model by increasing the actual returns to one's effort. We find no evidence of effects on our measure of grit after six months or five years.

Information and expected returns to innovation: In the bottom panel of Table 6, we investigate two additional mechanisms. Firstly, we measure if treated households adopted fifteen activities mentioned in the videos, such as purchasing pumps or using specific technologies. We find no effects on any of fifteen dummy variables capturing if households engage in an activity described in the videos (Appendix Table A.4), nor effects on a summary index across these fifteen variables (second last row of Table 6).⁵² This suggests that the households might have been aware of these activities even without the videos, as they are common within their villages.

Secondly, we explore whether exposure to the documentaries influenced households' beliefs about the returns to modern agricultural technologies, like fertilisers, despite these not directly featuring in the videos. We find no effects on households' beliefs about the returns to fertilisers.

Experimenter demand effects or reporting issues: Finally, one might worry that selfreported outcomes may be affected by biases. To avoid enumerators treating households differently, we ensured that enumerators were blind to treatment status (as mentioned in Section 2.2).

One may worry that treated households report outcomes they think would please the enumerator, or inflate reports so they appear to have made more progress to achieving aspirations. While we cannot totally rule these out, some patterns in our results suggest these issues are

^{52.} This finding helps to shed light on a potential puzzle in the results. Education investment is one key area of change due to our intervention in both the short and long term. However, the videos made little or no reference to schooling as a means of improving one's economic position. The absence of an information effect helps to square this: the intervention may have shifted goals and aspirations within the domains the treated households thought relevant for them, and not simply in areas that were mentioned during the video as used by the role models.

not driving our results. We can largely rule out experimenter demand effects simply from pleasing outsiders. While showing videos at a ticketed showing was unusual in such a remote setting, this applied both to the placebo and treated group, as the placebo screening mirrored the treatment exactly except for the video content. If demand effects occurred, it is likely we would see at least some placebo effects, but we do not.

However, it is possible that the aspirations treatment encouraged specific types of responses not suggested by the placebo, creating demand effects. Here too, some results suggest this is unlikely. First, treated households do not report higher values on some modules where the intervention might suggest particular answers might be desirable. There are no effects on savings or land under cultivation in the five-year follow-up, although both are clearly indicators of hard work. We did not find increased adoption of any of the activities or practices shown in the videos. Second, households report consistent increases in similar indicators in different modules of the survey. They report increased flows of spending on durable assets in the consumption module and increased asset stocks in the asset module. They report increased labour supply in both the time allocation module and agricultural inputs module. It is highly unlikely that households could identify that these different modules measured similar concepts to give inflated answers in both modules. Third, we see stronger effects in the endline than the midline, but it seems unlikely that experimenter demand effects would persist or increase over a five-year horizon. Finally, we examine non-pre-specified measures of living standards that were directly observed by the enumerator (Table 5). We see significant effects on the presence of a separate toilet just for the household relative to the control group (although not relative to the placebo group), suggesting improvements in living standards that cannot be accounted for by reporting biases. We also measured if enumerators found households had non-organic roofs, but find no effects here. This is not a perfect test as lack of effect does not indicate reporting bias: households may have invested, but in other types of assets, especially as these assets are lumpy and expensive. Thus, while we cannot fully rule them out, the pattern of our results suggests experimenter demand issues are unlikely.

Relatedly, the treatment may have made respondents report a more positive and optimistic description of their outcomes at present. However, we see few effects on modules which would be likely to be affected by increased optimism. We also see few effects on well-being in the endline, nor on household reports of their current position at the start of the aspirations and expectations module.

Ideally, we would have collected administrative data as a further check on the findings. But our attempts were thwarted. The smallest administrative unit for which administrative data is collected on relevant variables is the sub-district or *kebele* and not the village. While enumerators collected data on some relevant variables at this level, we did not treat enough people in a *kebele* to test for effects, with a median of only one per cent of households treated. Since 2018, this area has been inaccessible due to insecurity, preventing further data collection.

In sum, we cannot *prove* that the behavioural change is caused by the aspirations shift,
but we definitely cannot reject it as the most plausible explanation: we find little support for these alternatives, though other psychological pathways cannot be ruled out.

5.3 Discussion of main results

Overall, the results support our predictions. Our intervention showed documentaries featuring individuals who overcame poverty to achieve success telling their stories. In line with social learning theory in psychology, this has exposed the treated group to individuals whose initial life conditions they could relate to and whose success they could see as reachable.

The intervention led to positive changes in aspirations and economic outcomes. We finally combine all the summary indices (agricultural investment, educational investment, welfare, and the aspirations and expectations aggregate index) into a single omnibus index, which finds the intervention yields a positive and significant effect on outcomes of 0.26 standard deviations relative to the placebo and 0.24 standard deviations relative to the control, five years after exposure to the role models in the videos (Table 7). Results are robust in all alternative specifications. We find little or no heterogeneity in treatment effects on our summary indices across all the baseline measures of heterogeneity we had pre-specified (Appendix Figure A.4).

The patterns of results in Section 5 are consistent with households being induced to aspire to and emulate what better off households in their communities do, even though they had lived with them well before the intervention. Treated households engage more in the kind of activities and investments the top tercile in Table 1 were doing at baseline, such as investing in livestock and working more on the farm, rather than specifically doing what was portrayed in the videos. We also see increases in effort; locally common productive activities, crops and livestock; and investment into children's education that persist for five years after the intervention.⁵³

Our results counter concerns about some of the possible negative consequences of boosting aspirations and aspirations gaps. Ray (2006), Genicot and Ray (2017), and McKenzie, Mohpal, and Yang (2022) suggest boosting aspirations may give "false hope" (encouraging households to take decisions that make them worse off) or make people "frustrated" as aspirations have been raised too high, leading to less investment and effort, with negative welfare consequences. We see little evidence of this: we find more investment and higher standard of living, with no negative effects on well-being.

5.4 Cost-effectiveness analysis and benchmarking

Appendix Section K reports on intervention costs and benefit-cost ratios both at the experiment's (small) scale and at a reasonable scale of delivery. At the experiment's scale, the

^{53.} In an exploratory analysis, we also test whether effects vary by the terciles of durable assets used to categorise our sample in Table 1. We find that effects on the aspirations and expectations aggregate index are smaller among households in the lowest tercile of durable assets relative to those in the middle tercile (results available upon request). This would suggest that the intervention is not boosting aspirations amongst the poorest amongst a generally poor group. However, in Figure A.4, there is no differential effects on aspirations or expectations when those with below median assets are compared with the others, so we remain cautious about this result.

intervention costs \$62 (USD 2016 PPP) per household, driven by high fixed costs of producing documentaries for a small group of households and showing the videos to very few households in a village. But at a larger scale, it could cost \$10 (USD 2016 PPP) per household, using cost estimates from an experiment screening videos to 57,750 households via agricultural extension agents in a similar area (Bernard et al., 2019).

Back-of-the-envelope calculations suggest the intervention was highly cost-effective. Our cost-benefit ratios consider benefits only in terms of asset wealth after five years, not benefits in children's educational attainment or food security. We consider only realised benefits without assumptions about future flows from asset wealth. On most measures of asset wealth, the intervention would have more than paid for itself, even at the small scale of the experiment. If we assume the intervention is conducted at a reasonable scale, and, conservatively, we consider only benefits in terms of durable assets (where effects are highly robust across specifications), benefits per household are twice costs. If we consider benefits in durable, productive, and livestock assets, benefits are thirteen times costs, though effects on productive assets and livestock were noisily estimated and not always statistically significant. If considering increases in all assets including housing, the benefit-cost ratio is even higher.

These effects are in percentage terms similar to other studies of psychosocial interventions. Appendix Figure A.5, with further explanation in Table A.26, provides a comparison with some recent papers that tested similar psychosocial interventions, without trying to conduct a complete meta-analysis, as interventions and time-scales are difficult to compare. An evaluation in Kenya is close to our paper but only involves short-term outcomes (Orkin et al., 2023); others involve psycho-social interventions but that are more involved than our light-touch intervention such as Batista and Seither (2019), Blattman, Jamison, and Sheridan (2017), Blattman et al. (2023), Baranov et al. (2020), Cecchi et al. (2022), John and Orkin (2022) and Lubega et al. (2021). These usually occur over multiple weeks or months and target a number of psychological constructs other than or in addition to aspirations, such as patience/delayed gratification, planning and self-efficacy. We also included the BRAC-inspired graduation programme, involving a large asset transfer and other intensive support (Banerjee et al., 2015). This clearly has larger impacts on assets and living standards than ours. Finally, we include estimates of effects on aspirations from studies of exposure to role models (Beaman et al., 2012; Macours and Vakis, 2014; Rojas Valdes, Wydick, and Lybbert, 2021).

For outcomes for which we could compare the impacts in percentage terms (aspirations and expectations, effort in terms of time worked, assets and investments, and educational expenditure), our effects remain within the confidence intervals of most of these interventions, while the asset transfer programme has considerably larger effects. The novelty of our finding remains that this intervention is more light-touch and is evaluated over a longer time horizon than most of these papers (except for Blattman et al., 2023 and Baranov et al., 2020) and yet we find long-term persistence across different economic outcomes.

| After five years | (1) | (2) | (3) | (4) |
|--|----------------|---------|-----------------------|------------------------------------|
| | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. |
| Agricultural investment index | 0.21*** | 0.03 | 0.18^{***} | -0.00 |
| | (0.06) | (0.06) | (0.06) | (1.00) |
| | $[0.00]^{***}$ | [0.96] | $[0.01]^{***}$ | 1082 |
| Educational investment index | 0.25^{***} | 0.04 | 0.21^{***} | 0.00 |
| | (0.07) | (0.07) | (0.08) | (1.00) |
| | $[0.00]^{***}$ | [0.96] | $[0.01]^{***}$ | 1082 |
| Welfare index | 0.15^{**} | 0.01 | 0.14^{**} | -0.00 |
| | (0.07) | (0.07) | (0.07) | (1.00) |
| | $[0.03]^{**}$ | [0.96] | $[0.04]^{**}$ | 1090 |
| Aspiration index | 0.12^{**} | -0.03 | 0.15^{**} | 0.02 |
| | (0.06) | (0.06) | (0.06) | (1.00) |
| | $[0.04]^{**}$ | [0.96] | $[0.01]^{**}$ | 1904 |
| Expectations index | 0.21^{***} | -0.00 | 0.22^{***} | 0.01 |
| | (0.06) | (0.06) | (0.06) | (1.00) |
| | $[0.00]^{***}$ | [0.96] | $[0.00]^{***}$ | 1900 |
| Aspirations and expectations aggregate index | 0.19^{***} | -0.03 | 0.22^{***} | 0.01 |
| | (0.06) | (0.06) | (0.06) | (1.00) |
| | $[0.00]^{***}$ | [0.96] | $[0.00]^{***}$ | 1904 |
| Omnibus index | 0.26^{***} | 0.02 | 0.24^{***} | -0.00 |
| | (0.06) | (0.06) | (0.07) | (1.00) |
| | $[0.00]^{***}$ | [0.96] | $[0.00]^{***}$ | 1091 |

Table 7: SUMMARY INDICES IN WITHIN-VILLAGE ANALYSIS

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences in parameters obtained in first two columns. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and characteristics of the household head (or spouse for individual-level outcomes): age, years of education, an indicator for being single, and an indicator for being male. Regressions control for the baseline value of the outcome, except for the regressions on the agricultural investment index, welfare index, and omnibus index, which control for the respective index constructed from all the available components available at baseline. Heteroskedasticity-robust standard errors (or clustered at the household-level for individual-level outcomes) are in parentheses. Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. The unit of observation is the household, except for the aspirations and expectations indices (which are observed for both household head and their spouse). The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous outcomes above the 99^{th} percentile, though the indices aggregate all non-missing outcomes. The outcomes are inverse-covariance-weighted averages standardised relative to the within-village control group, following Anderson (2008). The agricultural investment index includes all outcomes reported in Table 3, with household's daily minutes in leisure being recoded to be negative. The educational investment index includes all outcomes reported in Table 4. The welfare index includes all outcomes reported in Table 5, with months of food insecurity in the last year recoded to be negative. The welfare index averages over the household head's subjective well-being outcomes. The aspirations and expectations indices are made of the reported aspirations (expectations) for income, wealth and years of education for children. The omnibus index aggregates the agricultural investment, educational investment, welfare, and aspirations and expectations aggregate standardised indices into a single index, following Kling, Liebman, and Katz (2007). We use the household head's aspirations and expectations aggregate index for the omnibus index.

5.5 Lack of spillovers

Here we test for potential spillover effects between treated and other households. Social learning theory and economic models of how aspirations change suggest that treated respondents who improve their economic position may become role models for others, changing their aspirations and/or future-oriented behaviour (Bandura, 1977; Genicot and Ray, 2017). Indeed, our main results suggest aspirations are influenced by observational learning from the experiences of other people, in the videos. So treated respondents may influence others who observe them or share content with non-participants. Less positively, untreated people may be resentful or discouraged if not selected for an intervention.

We construct a group of 10 pure control villages to test for spillover effects. In these villages, we collect data with 18 randomly selected households per village in the five-year follow-up. The pure control villages are not randomly selected but we argue they provide a plausible alternative counterfactual for the treatment villages. They come from the list of 84 villages randomly selected from our sampling frame of 189 villages with 50-100 households in the district, as was described in Section 2.2. As discussed there, our treatment villages were chosen based on logistical considerations: we found 16 large screening venues and chose the 64 villages closest to those screening venues such that we had four villages within walking distance of the 16 screening sites. The next 10 closest villages make up the pure control group. They would have been selected for the experiment, had we formed quintuplets or sextuplets instead of quadruplets of villages per screening site (see also Appendix Figure A.1). While not randomly selected, these villages appear similar to villages included in the experiment (Appendix J reports balance tests). The design is shown in Appendix Figure A.3.

Our test for spillovers compares the within-village control group to the pure control group (the specification is also in Appendix J). The within-village control group live in the same villages as the treated group and could be affected by any spillovers. We focus on the summary indices we use to summarise our main results, but standardise relative to the pure control group.

We find little systematic evidence of spillovers.⁵⁴ There are significant differences between the within-village control group and pure control group on only one of five indices (Column 3, Appendix Table A.23). If anything, we find some evidence of positive spillovers for the aspirations and welfare index, which could potentially attenuate our main estimates. The estimates of all other treatment effects are very close to those in our main specifications although the control group is different (Columns 1 and 4, Appendix Table A.23).

Spillover effects may take longer than five years to occur or are small in magnitude, requiring a larger and longer-term experiment to detect them. But in this setting, we do not have strong evidence that changes in aspirations are shaped by day-to-day experiences of what others do or think, as implied by Carvalho et al. (2023) or Genicot and Ray (2020). Maybe this is not sur-

^{54.} We pre-specified further potentially less well-powered tests for spillovers, reported in our working paper at https://cepr.org/publications/dp18492. These show little systematic evidence of spillovers.

prising: as Table 1 showed, the respondents have always lived with better-off people displaying similar behaviours to those in the videos. The videos may be a particularly resonant and memorable "vicarious" experience not provided by encountering successful people in one's daily life.

6 Conclusion

We randomly invited individuals in a poor and isolated area to a one-hour video documentary in which four people from similar backgrounds to the audience tell their life story of getting out of poverty. After five years, we find persistent effects on whether households invest for the future, and some indicators of their standard of living. These results are meaningful. The size of the effects are not very large – a few dollars more spending on education, some more durable assets – but in percentage terms they are not insubstantial. Something is triggered that affected forward-oriented behaviour. We find evidence consistent with a change in aspirations being the main psychological mechanism.

A back-of-the-envelope calculation suggests the intervention was cost-effective. Counting only the robust increase in durable assets, the benefit-cost ratio after five years is about 2-to-1 for a reasonable scale of implementation. In percentage terms, our effects align with those of similar psycho-social interventions – though they remain below those of more intensive anti-poverty programs – yet exhibit remarkable persistence.

Is this intervention giving false hope? We cannot fully judge this. But treated households see positive effects on assets and standard of living, suggesting overall positive impacts. And we did not suggest to individuals – rightly or wrongly – what path would lead them out of poverty, unlike most interventions that offer 'solutions' in microcredit, health or education. We only invited our treatment group to listen to stories told by individuals from similar backgrounds.

We want to be cautious. This is a remote and isolated area, and it is not clear that these effects would be present or persistent elsewhere. At least this study offers a proof of concept, that such light-touch easily scalable interventions may have persistent positive effects.

References

Alamayo, D. 2021. "Safuu: The Indigenous Oromo Moral Thought." *Journal of Philosophy and Culture* 9 (1): 1–10.

Alan, S., T. Boneva, and S. Ertac. 2019. "Ever Failed, Try Again, Succeed Better: Results from a Randomized Educational Intervention on Grit." *The Quarterly Journal of Economics* 134 (3): 1121–1162.

Alan, S. and S. Ertac. 2018. "Fostering Patience in the Classroom: Results from a Randomised Educational Intervention." *Journal of Political Economy* 126 (5): 1865–1911.

Anderson, M. L. 2008. "Multiple Inference and Gender Differences in the Effects of Early Intervention: A Reevaluation of the Abecedarian, Perry Preschool, and Early Training Projects." *Journal of the American Statistical Association* 103 (484): 1481–1495.

Angelucci, M. and D. Bennett. 2024. "The Economic Impact of Depression Treatment in India." *American Economic Review* 114 (1): 169–198. Angrist, J. and A. Krueger. 1999. "Empirical Strategies in Labor Economics." In *Handbook of Labor Economics*, vol. 3. Elsevier, 1277–1366.

Appadurai, A. 2004. "The Capacity to Aspire: Culture and the Terms of Recognition." In *Culture and Public Action*, edited by Vijayendra Rao and M. Walton. Palo Alto, California: Stanford University Press, 59–84.

——. 2013. The Future as a Cultural Fact: Essays on the Global Condition. London, UK: Verso Books.

Ashraf, N., G. Bryan, A. Delfino, E. Holmes, L. Iacovone, and A. Pople. 2022. "Learning to See the World's Opportunities: The Impact of Imagery on Entrepreneurial Success." *London School of Economics & Political Science Working Paper*: 1–80.

Ashraf, N., D. S. Karlan, and W. Yin. 2006. "Tying Odysseus to the Mast: Evidence from a Commitment Savings Product in the Philippines." *Quarterly Journal of Economics* 121 (2): 635–672.

Bandura, A. 1977. "Self-Efficacy: Toward a Unifying Theory of Behavioural Change." *Psychological Review* 84 (2): 191–215.

——. 2001. "Social Cognitive Theory: An Agentic Perspective." Annual Review of Psychology 52 (1): 1–26.

Banerjee, A., E. Duflo, N. Goldberg, D. Karlan, R. Osei, W. Parienté, J. Shapiro, B. Thuysbaert, and C. Udry. 2015. "A Multifaceted Program Causes Lasting Progress for the Very Poor: Evidence from Six Countries." *Science* 348 (6236): 1260799.

Baranov, V., S. Bhalotra, P. Biroli, and J. Maselko. 2020. "Maternal Depression, Women's Empowerment, and Parental Investment: Evidence from a Randomized Controlled Trial." *American Economic Review* 110 (3): 824–59.

Barker, N., G. Bryan, D. Karlan, A. Ofori-Atta, and C. Udry. 2022. "Cognitive Behavioral Therapy among Ghana's Rural Poor Is Effective Regardless of Baseline Mental Distress." *American Economic Review: Insights* 4 (4): 527–45.

Batista, C. and J. Seither. 2019. "Aspirations, Expectations, Identities: Behavioral Constraints of Micro-entrepreneurs." NOVAFRICA Working Paper Series 1906: 1–28.

Beaman, L., E. Duflo, R. Pande, and P. Topalova. 2012. "Female Leadership Raises Aspirations and Educational Attainment for Girls: A Policy Experiment in India." *Science* 335 (6068): 582–586.

Belloni, A., V. Chernozhukov, and C. Hansen. 2014. "Inference on Treatment Effects after Selection among High-Dimensional Controls." *The Review of Economic Studies* 81 (2): 608–650.

Benabou, R. 1996. "Equity and Efficiency in Human Capital Investment: The Local Connection." *The Review of Economic Studies* 63 (2): 237–264.

Benjamini, Y., A. M. Krieger, and D. Yekutieli. 2006. "Adaptive Linear Step-up Procedures that Control the False Discovery Rate." *Biometrika* 93 (3): 491–507.

Bernard, T., S. Dercon, K. Orkin, and A. S. Taffesse. 2014. "The Future in Mind: Aspirations and Forward-Looking Behaviour in Rural Ethiopia." *Bureau for Research and Economic Analysis of Development Working Paper* 429: 1–42.

Bernard, T., S. Makhija, D. J. Spielman, and G. T. Abate. 2019. "The (Marginal) Cost of Technology Adoption: A Cost-Effectiveness Analysis of Digital Green's Video-Mediated Agricultural Extension Approach in Ethiopia." *International Food Policy Research Institute Project Note* : 1–6.

Bernard, T. and A. S. Taffesse. 2014. "Aspirations: An Approach to Measurement with Validation Using Ethiopian Data." *Journal of African Economies* 23 (2): 189–224.

Besley, T. 2016. "Aspirations and the Political Economy of Inequality." Oxford Economic Papers 69 (1): 1–35.

Bhat, B., J. de Quidt, J. Haushofer, V. Patel, G. Rao, F. Schilbach, and P. Vautrey. 2022. "The Long-Run Effects of Psychotherapy on Depression, Beliefs, and Economic Outcomes." *National Bureau of Economic Research Working Paper* 30011: 1–76.

Bickel, G. W., M. Nord, C. Price, W. Hamilton, and J. Cook. 2000. *Guide to Measuring Household Food Security*. Alexandria, Virginia: Food and Nutrition Service, United States Department of Agriculture.

Binswanger, H. P. 1980. "Attitudes Toward Risk: Experimental Measurement in Rural India." *American Journal of Agricultural Economics* 62 (3): 395–407.

Bisin, A. and T. Verdier. 2001. "The Economics of Cultural Transmission and the Dynamics of Preferences." *Journal of Economic Theory* 97 (2): 298–319.

Blattman, C., J. Jamison, and M. Sheridan. 2017. "Reducing Crime and Violence: Experimental Evidence from Cognitive Behavioral Therapy in Liberia." *American Economic Review* 107 (4): 1165–1206.

Blattman, C., S. Chaskel, J. C. Jamison, and M. Sheridan. 2023. "Cognitive Behavioral Therapy Reduces Crime and Violence over Ten Years: Experimental Evidence." *American Economic Review: Insights* 5 (4): 527–45.

Bossuroy, T., M. Goldstein, B. Karimou, D. Karlan, H. Kazianga, W. Parienté, P. Premand, C. Thomas, C. Udry, J. Vaillant, and K. Wright. 2022. "Tackling Psychosocial and Capital Constraints to Alleviate Poverty." *Nature* 605 (7909): 291–297.

Bouguen, A., Y. Huang, M. Kremer, and E. Miguel. 2019. "Using Randomized Controlled Trials to Estimate Long-Run Impacts in Development Economics." *Annual Review of Economics* 11 (1): annurev–economics–080218–030333.

Campos, F., M. Frese, M. Goldstein, L. Iacovone, H. Johnson, D. McKenzie, and M. Mensmann. 2017. "Teaching Personal Initiative Beats Traditional Training in Boosting Small Business in West Africa." *Science* 357 (6357): 1287–1290.

Cantril, H. 1966. *The Pattern of Human Concern*. New Brunswick, New Jersey: Rutgers University Press.

Carvalho, J.-P., A. Bergeron, J. Henrich, N. Nunn, and J. Weigel. 2023. "Zero-Sum Thinking, the Evolution of Effort-Suppressing Beliefs, and Economic Development." *National Bureau of Economic Research Working Paper* 31663: 1–56.

Cecchi, F., A. Garcia, R. Lensink, and B. Wydick. 2022. "Aspirational Hope, Dairy Farming Practices, and Milk Production: Evidence from a Randomized Controlled Trial in Bolivia." *World Development* 160: 106087.

Chinoy, S., N. Nunn, S. Sequeira, and S. Stantcheva. 2025. "Zero-Sum Thinking and the Roots of US Political Divides." 31688: 1–135.

Chong, A., S. Duryea, and E. La Ferrara. 2012. "Soap Operas and Fertility: Evidence from Brazil." *American Economic Journal: Applied Economics* 4 (4): 1–31.

Cohen, J. 2001. "Defining Identification: A Theoretical Look at the Identification of Audiences with Media Characters." *Mass Communication and Society* 4: 245–264.

Crivello, G. and N. van der Gaag. 2016. "Between Hope and a Hard Place: Boys and Young Men Negotiating Gender, Poverty and Social Worth in Ethiopia." Young Lives Working Paper 160.

Dalton, P., S. Ghosal, and A. Mani. 2016. "Poverty and Aspirations Failure." *Economic Journal* 126: 165–188.

Deininger, K., D. A. Ali, and T. Alemu. 2008. "Assessing the Functioning of Land Rental Markets in Ethiopia." *Economic Development and Cultural Change* 57 (1): 67–100.

Díaz, D., A. Blanco, M. Bajo, and M. Stavraki. 2015. "Fatalism and Well-being across Hispanic Cultures: The Social Fatalism Scales (SFS)." Social Indicators Research 124: 929–945.

Duckworth, A. L., C. Peterson, M. D. Matthews, and D. R. Kelly. 2007. "Grit: Perseverance and Passion for Long-Term Goals." *Journal of Personality and Social Psychology* 92 (6): 1087–1101.

Durkheim, E. 1893. *The Division of Labour in Society, Contemporary Social Theory*. London: Macmillan.

Durlauf, S. 1996. "A Theory of Persistent Income Inequality." *Journal of Economic Growth* 1 (1): 75–93.

Eble, A. and M. Escueta. 2022. "Demand, Supply, and Learning in a Very Low-income Context." *EdWorkingPaper No. 21-473* (20): 1–74.

Elder, J. W. 1966. "Fatalism in India: A Comparison between Hindus and Muslims." *Anthropological Quarterly* 39 (3): 227–243.

Fairlie, R., F. Hoffmann, and P. Oreopoulos. 2014. "A Community College Instructor Like me: Race and Ethnicity Interactions in the Classroom." *American Economic Review* 104 (8): 2567–2591.

Feagin, J. R. 1975. *Subordinating the Poor: Welfare and American Beliefs*. Englewood Cliffs, New Jersey: Prentice Hall.

Fishbach, A. and M. J. Ferguson. 2007. "The Goal Construct in Social Psychology." In *Social Psychology: Handbook of Basic Principles*, edited by A. W. Kruglanski and E. T. Higgins. New York: Guilford, 490–515.

Foster, G. M. 1965. "Peasant Society and the Image of Limited Good." *American Anthropologist* 67 (2): 293–315.

Fruttero, A., N. Muller, and Ó. Calvo-González. 2025. "The Power and Roots of Aspirations." *The World Bank Research Observer* 40 (1): 58–103.

Gabaix, X. and D. Laibson. 2017. "Myopia and Discounting." National Bureau of Economic Research Working Paper 23254: 1–51.

Gemeda, A. B. and K. Rajani. 2018. "Indigenous Religion and Being Human: The Case of "Waaqeffannaa" Religion of the Oromo People." *International Journal of Academic Research and Development* 3 (2): 566–574.

Genicot, G. and D. Ray. 2017. "Aspirations and Inequality." Econometrica 85 (2): 489–519.

———. 2020. "Aspirations and Economic Behavior." Annual Review of Economics 12: 715–746.

Ghosal, S., S. Jana, A. Mani, S. Mitra, and S. Roy. 2022. "Sex Workers, Stigma, and Self-Image: Evidence from Kolkata Brothels." *The Review of Economics and Statistics* 104 (3): 431–448.

Green, M. and T. Brock. 2000. "The Role of Transportation in the Persuasiveness of Public Narratives." *Journal of Personality and Social Psychology* 79 (5): 701–721.

Greene, A. L., H. J. Sullivan, and K. Beyard-Tyler. 1982. "Attitudinal Effects of the Use of Role Models in Information About Sex-Typed Careers." *Journal of Educational Psychology* 74 (3): 393–398.

Hadley, C., D. Lindstrom, F. Tessema, and T. Belachew. 2008. "Gender Bias in the Food Insecurity Experience of Ethiopian Adolescents." *Social Science & Medicine* 66 (2): 427–438.

Haushofer, J., R. Mudida, and J. Shapiro. 2020. "The Comparative Impact of Cash Transfers and a Psychotherapy Program on Psychological and Economic Well-Being." *National Bureau of Economic Research* 28106.

Heller, S. B., A. K. Shah, J. Guryan, J. Ludwig, S. Mullainathan, and H. A. Pollack. 2016. "Thinking, Fast and Slow? Some Field Experiments to Reduce Crime and Dropout in Chicago." *The Quarterly Journal of Economics* 132 (1): 1–54. Hoff, K. and J. E. Stiglitz. 2016. "Striving for Balance in Economics: Towards a Theory of the Social Determination of Behaviour." *Journal of Economic Behavior & Organization* 126: 25–57.

Hunt, M. 2004. "Race/Ethnicity and Beliefs about Wealth and Poverty." Social Science Quarterly 85 (3): 827–853.

Ingersoll, J. 1966. "Fatalism in Village Thailand." Anthropological Quarterly 39 (3): 200–225.

Janzen, S. A., N. Magnan, S. Sharma, and W. M. Thompson. 2017. "Aspirations Failure and Formation in Rural Nepal." *Journal of Economic Behavior and Organization* 139: 1–25.

Jensen, R. and E. Oster. 2009. "The Power of TV: Cable Television and Women's Status in India." *Quarterly Journal of Economics* 124 (3): 1057–1094.

John, A. and K. Orkin. 2022. "Can Simple Psychological Interventions Increase Preventive Health Investment?" *Journal of the European Economic Association* 20 (3): 1001–1047.

Kahneman, D., J. L. Knetsch, and R. Thaler. 1986. "Fairness as a Constraint on Profit Seeking: Entitlements in the Market." *American Economic Review* 76 (4): 728–741.

Kearney, M. S. and P. B. Levine. 2020. "Role Models, Mentors, and Media Influences." *The Future of Children* 30 (1): 83–106.

Kebede, A. 1999. The Role of Cultural/Historical Contingencies, Decenteredness, and Movement Culture in the Endurance of the Rastafari. Oklahoma State University.

Kerwin, J. T., N. Rostom, and O. Sterck. 2024. "Striking the Right Balance: Why Standard Balance Tests Over-reject the Null, and How to Fix It." University of Washington Working Paper : 1–56.

Kling, J. R., J. B. Liebman, and L. F. Katz. 2007. "Experimental Analysis of Neighborhood Effects." *Econometrica* 75 (1): 83–119.

Kofoed, M. and E. McGovney. 2019. "The Effect of Same-Gender and Same-Race Role Models on Occupation Choice: Evidence from Randomly Assigned Mentors at West Point." *Journal of Human Resources* 54: 430–467.

Kőszegi, B. 2010. "Introduction to Reference-Dependent Preferences." https://neuroeconomics.org/documents/Koszegi_Workshop2010.pdf.

Kremer, M., G. Rao, and F. Schilbach. 2019. "Behavioral Development Economics." In *Handbook* of *Behavioral Economics: Applications and Foundations 1*, vol. 2. Elsevier, 345–458.

La Ferrara, E. 2016. "Mass Media and Social Change: Can We Use Television to Fight Poverty?" *Journal of the European Economic Association* 14 (4): 791–827.

——. 2019. "Presidential Address: Aspirations, Social Norms, and Development." Journal of the European Economic Association 17 (6): 1687–1722.

Lee, D. S. 2009. "Training, Wages, and Sample Selection: Estimating Sharp Bounds on Treatment Effects." *The Review of Economic Studies* 76 (3): 1071–1102.

Leight, J., D. O. Gilligan, M. Mulford, A. Seyoum Taffesse, and H. Tambet. 2024. "Aspiring to More? New Evidence on the Effect of Light-touch Aspirations Interventions in Rural Ethiopia." *Journal of Development Effectiveness* 16: 485–498.

Levenson, H. 1981. "Differentiating Among Internality, Powerful Others, and Chance." In *Research with the Locus of Control Construct Vol.1*, edited by H. M. Lefcourt. New York: Academic Press, 15–63.

Levine, D. N. 1965. Wax & Gold: Tradition and Innovation in Ethiopian Culture. University of Chicago Press.

Locke, E. A. and G. P. Latham. 1990. A Theory of Goal Setting and Task Performance. Englewood Cliffs, New Jersey: Prentice Hall.

Lockwood, P. and Z. Kunda. 1997. "Superstars and Me: Predicting the Impact of Role Models on the Self." *Journal of Personality and Social Psychology* 73 (1): 91.

Lubega, P., F. Nakakawa, G. Narciso, C. Newman, A. N. Kaaya, C. Kityo, and G. A. Tumuhimbise. 2021. "Body and Mind: Experimental Evidence from Women Living with HIV." *Journal of Development Economics* 150: 102613.

Lybbert, T. and B. Wydick. 2018. "Poverty, Aspirations and the Economics of Hope." *Economic Development and Cultural Change* 66 (4): 709–753.

Macours, K. and R. Vakis. 2014. "Changing Households' Investment Behaviour Through Social Interactions with Local Leaders: Evidence from a Randomised Transfer Programme." *The Economic Journal* 124 (576): 607–633.

Mani, A. and E. Riley. 2021. "Social Networks, Role Models, Peer Effects, and Aspirations." In *Social Mobility in Developing Countries: Concepts, Methods, and Determinants*, edited by V. Iversen, A. Krishna, and K. Sen, chap. 17. Oxford, United Kingdom: Oxford University Press, 424–447.

Marx, D. M. and S. J. Ko. 2012. "Superstars "Like" Me: The Effect of Role Model Similarity on Performance Under Threat." *European Journal of Social Psychology* 42 (7): 807–812.

McKelway, M. 2021. "Women's Employment in India: Intra-Household and Intra-Personal Constraints." *Dartmouth College Working Paper* : 1–102.

McKenzie, D., A. Mohpal, and D. Yang. 2022. "Aspirations and Financial Decisions: Experimental Evidence from the Philippines." *Journal of Development Economics* 156: 102846.

Mokyr, J. 2016. A Culture of Growth: The Origins of the Modern Economy. Princeton University Press.

Nugent, J. 1985. "The Old-Age Security Motive for Fertility." *Population and Development Review* : 75–97.

Orkin, K., R. Garlick, M. Mahmud, R. Sedlmayr, J. Haushofer, and S. Dercon. 2023. "Aspiring to a Better Future: Can a Simple Psychological Intervention Reduce Poverty?" *National Bureau of Economic Research Working Paper* 31735: 1–102.

Porter, C. and D. Serra. 2020. "Gender Differences in the Choice of Major: The Importance of Female Role Models." *American Economic Journal: Applied Economics* 12 (3): 226–54.

Rahmato, D. and A. Kidani. 1999. Consultations with the Poor: A Study to Inform the World Development Report 2000/01 on Poverty and Development - National Report Ethiopia.

Ray, D. 2006. "Aspirations, Poverty and Economic Change." In *Understanding Poverty*, edited by Banerjee, Abhijit and Benabou, Roland and Mookherjee, Dilip. Oxford: Oxford University Press, 409–422.

Riley, E. 2022. "Role Models in Movies: The Impact of Queen of Katwe on Students' Educational Attainment." *The Review of Economics and Statistics* 106 (2): 334–351.

Rojas Valdes, R. I., B. Wydick, and T. J. Lybbert. 2021. "Can Hope Elevate Microfinance? Evidence from Oaxaca, Mexico." Oxford Economic Papers 74 (1): 236–264.

Ross, P. H. 2019. "Occupation Aspirations, Education Investment, and Cognitive Outcomes: Evidence from Indian Adolescents." *World Development* 123: 104613.

Rossi, P. and M. Godard. 2022. "The Old-Age Security Motive for Fertility: Evidence from the Extension of Social Pensions in Namibia." *American Economic Journal: Economic Policy* 14 (4): 488–518.

Serneels, P. and S. Dercon. 2021. "Aspirations, Poverty, and Education. Evidence from India." *The Journal of Development Studies* 57 (1): 163–183.

Stout, J. G., N. Dasgupta, M. Hunsinger, and M. A. McManus. 2011. "STEMing the Tide: Using Ingroup Experts to Inoculate Women's Self-Concept in Science, Technology, Engineering and Mathematics (STEM)." *Journal of Personality and Social Psychology* 100 (2): 255–270.

Whelan, C. T. 1996. "Marginalization, Deprivation, and Fatalism in the Republic of Ireland: Class and Underclass Perspectives." *European Sociological Review* 12 (1): 33–51.

Willis, P. 1977. Learning to Labour. How Working Class Kids Get Working Class Jobs. New York: Columbia University Press.

Wilson, W. J. 1987. The Truly Disadvantaged: The Inner City, the Underclass, and Public Policy. Chicago: University of Chicago Press.

Woldehanna, T., A. Mekonnen, N. Jones, B. Tefera, J. Seager, T. Alemu, and G. Asgedom. 2008. "Education Choices in Ethiopia: What Determines Whether Poor Households Send Their Children to School?" *Ethiopian Journal of Economics* 17 (1): 1–38.

Wollie, C. W. 2009. "Causal Attributions for Poverty among Youths in Bahir Dar, Amhara Region, Ethiopia." *Journal of Social, Evolutionary, and Cultural Psychology* 3 (3): 251.

World Bank. 2015. World Development Report 2015: Mind, Society, and Behavior. Washington, DC: World Bank.

For Online Publication: Appendix

Contents

| Α | Map and timeline | 50 |
|--------------|--|--|
| В | Theoretical appendix B.1 Aspirations in a reference-dependent model B.2 Model predictions from a change in aspirations | 52 52 53 |
| С | Summary of documentaries and placebo | 54 |
| D | Additional results | 56 |
| Е | Data and measures E.1 Conversions from Ethiopian birr (ETB) to USD PPP | 62 62 62 63 67 67 |
| F | Experimental integrityF.1Design, sample, and compliance< | 69 69 71 73 |
| G | Deviations from the Pre-Analysis Plan (PAP) G.1 Changes to pre-specified outcome variables G.2 Multiple inference adjustment | 75 76 78 |
| н | Robustness tests | 86 |
| Ι | Heterogeneity | 92 |
| J | Spillovers | 93 |
| K | Cost effectiveness analysis K.1 Cost of the intervention | 96 96 96 |
| \mathbf{L} | Comparison of effect sizes with other interventions | 98 |

Appendix roadmap

Section A provides a timeline of the study and additional descriptive statistics. Section B presents a formal model to conceptualise aspirations and the effect of raising them on economic behaviour. Section C describes two of the individual stories featured in the documentaries alongside with a description of the content of the placebo screenings. Section D shows detailed estimates on the aspirations and expectations measures across rounds and additional results. Section E provides details on variable construction. Section F provides details about the study design, compliance, balance, and attrition. Section G outlines any of the ways in which the analysis presented departs from our original Pre-Analysis Plan. Section I describes the specifications used to test for the robustness of the main results and Section I reports heterogenous treatment effects across pre-specified dimensions. Section J presents tests for spillovers. Section K provides details of a back-of-the-envelope estimate of the intervention's cost-effectiveness, and finally Section L compares the magnitude of our estimates to related studies.

A Map and timeline



Figure A.1: VILLAGES IN THE STUDY

Notes: Black dots correspond to the treated villages. White and black dots corresponds to the pure control villages used in the spillover analysis. White dots correspond to the villages that were part of the original sampling frame but are not included in the analysis.



Panel B: Schooling-age cohorts at the time of the surveys:

| | Age in | Age in |
|--------|---------|---------|
| Cohort | 2010/11 | 2015/16 |
| 1 | 11-15 | 16-20 |
| 2 | 2-10 | 7-15 |
| 2(a) | 7-10 | 12-15 |

Figure A.2: TIMELINE OF THE STUDY

Notes: Panel A shows the overall study timeline. Grey horizontal bars denote the periods where a survey or the screening intervention took place. Panel B shows the cohort ages of children between baseline and the five-year follow-up. These cohorts are used to define educational outcomes.

B Theoretical appendix

This appendix section provides a formal model to derive prediction on how an intervention targeting an exogenously induced change in aspirations might affect economic decisions.

B.1 Aspirations in a reference-dependent model

We explore the effect of a change in aspirations, or the reference point, on effort and investment, in a simple multi-period model of allocating effort and resources for future benefit versus consuming more or enjoying more leisure now.

Existing economic models of aspirations formation and its consequences capture the idea that achieving goals may yield utility (Heath, Larrick, and Wu, 1999; Dalton, Ghosal, and Mani, 2016; Genicot and Ray, 2017). These models use reference-dependent utility (Kőszegi and Rabin, 2006) and interpret goals as reference points.

Aspirations enter our model as a reference point: instantaneous utility $v(c_t, l_t; a_t)$ is defined not just over consumption c_t and leisure l_t , but is anchored by the aspirations one has for one's economic position a_t . More specifically, we assume that $v(c_t, l_t; a_t) = u(c_t, l_t) + z(c_t - a_t)$, with $u_{c_t}, u_{l_t} > 0$ and $u_{c_tc_t}, u_{l_tl_t} < 0.^{55}$ The function z can be seen as a loss-gain function: not fulfilling one's aspirations reduces welfare, so $z(c_t - a_t) \leq 0$ if $c_t \leq a_t$. Or equivalently, starting from below and getting closer to one's goal increases one's utility. Overachieving, when $c_t > a_t$, is assumed to be adding utility or $z(c_t - a_t) > 0$. This loss-gain function is assumed to be increasing and concave in c_t , i.e. $z_{c_t} > 0$, $z_{c_tc_t} \leq 0$.

We consider a unitary household, with an infinite time horizon, maximising discounted lifetime utility at each moment t, $W_t = \sum_{s=0}^{\infty} \beta^s v(c_{t+s}, l_{t+s}; a_{t+s})$, with the discount factor being $0 < \beta \leq 1$. At the start of each period t, the household has revenue y_t and assets A_t available, based on decisions at t-1. Total resources $A_t + y_t$ in each period t can be allocated to either consumption or used to produce future revenue. Revenue at t+1 is obtained from allocating both effort $e_t = 1 - l_t$ and capital $k_t = A_t + y_t - c_t$ in period t. The transition equation for future revenue is $y_{t+1} = f(k_t, e_t)$, with $f_{k_t}, f_{e_t} > 0$ and $f_{k_t k_t}, f_{e_t e_t} < 0$. Allowing for some depreciation δ from using capital, the transition equation for assets is $A_{t+1} = (1-\delta).k_t$.

Maximising W_t , subject to the two transition equations for revenue and assets defined for each period t+s, allows us to derive the following Euler equations from the first order conditions governing decisions about consumption c_t and leisure l_t :

(2)
$$u_{ct} + z_{ct} = \beta \cdot (1 + f_{kt} - \delta) \cdot (u_{c_{t+1}} + z_{c_{t+1}})$$

(3)
$$u_{l_t} = \beta f_{e_t} (u_{c_{t+1}} + z_{c_{t+1}})$$

55. We use throughout the notation $\frac{\partial g(x_t)}{\partial x_t} = g_{x_t}$ for any function g.

Equation (2) governs choices between consumption today versus saving and investing for future consumption; Equation (3) governs taking leisure today or putting in effort with a return tomorrow. These are familiar Euler equations, except for the terms defined by the loss function. Without the loss function, the model yields the standard intertemporal results, whereby the marginal utility of present consumption (or leisure) will equal the discounted marginal utility of future consumption generated from returns to savings (or effort).

B.2 Model predictions from a change in aspirations

The model predicts that a change in aspirations can affect decisions about consumption and leisure. If aspirations for the future (a_{t+1}) increase at t, current effort and/or investment will increase. The intuition is captured by considering how an increase in a_{t+1} affects the Euler equations. z is a concave function in its argument $(c_{t+1}-a_{t+1})$ for a given a_{t+1} . Thus $\frac{\partial z_{c_{t+1}}}{\partial a_{t+1}} > 0$. For a given past level of aspirations, a_t , an increase in aspirations for the future, a_{t+1} , will boost the right-hand side of both Equation (2) and (3). For both equations to hold simultaneously after this change, the left-hand side of each equation needs to go up too and/or the other terms on the right hand side need to go down. To restore equality in Equation (3), a reduction in leisure today is required: investment in the future through effort will increase u_{l_t} and reduce the marginal product of labour f_{e_t} . To restore equality in Equation (2) the household will need to consume less, and save and invest more at t so that future consumption increases. In turn, this decrease in present consumption will increase the left-hand side of Equation (2), as consuming less at t will increase marginal utility u_{c_t} , as well as z_{c_t} . More savings will also reduce the marginal product of capital f_{k_t} on the right side of (2) and reduce $u_{c_{t+1}}$ until equality across both Equation (2) and (3) is restored.

It follows that someone with lower aspirations for the future will limit investment and effort relative to someone otherwise identical in all other characteristics but with higher aspirations.

The model yields a more ambiguous prediction on how an upward shift in aspirations would affect consumption. Equation (3) offers a rule for the path of consumption, not for the level in each period. Boosting aspirations will boost future wealth, as there is more incentive to shift resources to the future for a given discount rate. In turn, increased future wealth will boost consumption at some point in the future. Given the stronger incentives to save and invest, whether higher aspirations will also lead to higher levels of consumption in the near future will depend on individual preferences, in particular the inter-temporal substitution elasticity and other features of the underlying utility function (Deaton, 1992). In particular, the change in aspirations for the future increases the opportunity cost of consuming today. This generates both a *income effect* – the value of lifetime assets increases because they yield higher returns in the future – and a *price effect* – the opportunity cost of consumption at any moment in time increases as well. The income effect allows for more consumption at any moment in time, but the price effect will encourage moving consumption to the future. Preferences will determine when the former will outweigh the latter across the consumption path. There are some implications of the assumptions of our model. First, we remain agnostic about where the reference point comes from, beyond that it is not a decision variable, in line with standard treatments in behavioural economics. The main text offered links to the psychological and anthropological literature – suggesting links to past experiences, as well as norms and values. If the reference point could be set as part of the optimisation problem, then it follows that if there is a gain from overachieving, then, to maximise utility, the reference point would be set to be as low as possible, which would be a trivial result. We also abstract from any endogenous revision of aspirations within the model, such as in response to past attainment. Second, our assumptions imply a loss from underachieving, with marginal losses increasing for higher levels of underachievement. This setup is consistent with Dalton, Ghosal, and Mani (2016)'s assumptions for underachieving, while Genicot and Ray (2017) assume a gain from overachieving, i.e. when $c_t > a_t$, but no effect from underachieving (i.e. frustration does not come at a cost). Our assumption and these other formulations of utility around the reference point yield the same underlying intuition: if aspirations are low relative to what could be achieved, boosting aspirations will provide incentives to put in more effort.

Finally, we note the model does not make predictions on the behaviour of expectations. The close empirical relationship between the measures of aspirations and expectations in our data makes it difficult to distinguish these concepts from one another or draw any conclusions about how they relate. Theoretically, predictions about the effects of the videos on expectations are complex. Expectations may simply capture beliefs about the outcomes of economic choices. They then likely change only gradually in response to the videos: the videos affect aspirations, which drive changes in economic choices and outcomes, and these changes in choices and outcomes indirectly change expectations. On the other hand, expectations may be another proxy for reference points (Kőszegi and Rabin, 2006). They then have a dual role: they may be affected by the videos and determine economic choices via reference points, and also reflect beliefs about the outcomes of economic choices, leading to complex equilibrium concepts when expectations are formed endogenously. Our limited measures of expectations mean studying this is beyond the scope of this paper.

C Summary of documentaries and placebo

The treatment consisted of four documentaries about two men and two women. Two documentaries are described below. Two documentaries are not summarised here, "Immortal Treasure", about Ayelech Fikre, and "The Exemplary Achievement", about Waki Feyyera. The four documentaries and an example of the placebo segments are available at: https://www.youtube.com/channel/UCqfoNjCzt8YPjTRWQaMQfAg.

Beshir Malim Yisak, in the video "The Fast Journey"

Beshir Malim Yisak is a 27 year old farmer, married, with two children. He has no formal education but is considered a model farmer in the area for his achievements in a short period.

Five years ago, in an area where most of the inhabitants usually breed cattle, Beshir started crop production. He consulted an agricultural expert in a local NGO about good farming practices and implemented everything he learned. He planted vegetables and sold them at the market, and bought a pair of oxen after a good harvest. Three years later, Beshir used savings to purchase a water pump from Addis Ababa, with the help of the agricultural extension agent. He rented additional land to expand his farm as he could water a larger area. He planted papaya, sugar cane and maize and increased his productivity by improving his soil fertility. He gradually built up a large herd of cattle. He started keeping bees for honey. During 2007, when tree planting was encouraged by village administrations, he produced and distributed seedlings to seven peasant associations and a local NGO in the area. Extension agents and fellow farmers speak of him as an innovator and hard worker.

Teyiba Abdella, in the video "The Life-Transforming Flour Trade"

Most people in Teyiba Abdella's district are involved in cultivating crops and livestock and in trade. Teyiba is engaged in both trade and farming. Her parents refused to bless her marriage, so Teyiba and her husband, Aliya Yousuf, started their married life with hardly any income or assets. Their fellow villagers contributed one birr each to help them start their life together. Using the neighbours' contributions as seed money, Teyiba began trading wheat flour. She used to walk three hours to market carrying 50 kilograms of flour on her back. A woman who owns a flour mill in the market town observed her efforts and offered her credit to purchase flour. After selling the flour she obtained on credit, Teyiba paid back her debt and saved her profits. Because she paid her debts on time, the miller started giving her up to 100 kilograms of flour on credit. Teyiba also began trading eggs and chickens and bought a donkey to carry loads to the market. Then she and her husband opened their own shop. They built themselves a house and bought land in the nearby village to build another house. Teyiba's husband does most household chores while she runs the businesses. Other villagers used to criticise Teyiba for being the main breadwinner, but she rejected their criticisms. People in the village now have a high regard for her hard work and commitment. Teyiba's husband admires her strength and believes she is a great role model for people in their village.

Example segment from placebo treatment

The clip's title "Boru Bari", literally meaning "Tomorrow Morning", is meant to suggest the idea that "tomorrow is another day". It is a humorous take on rural life. The main character describes his current life to a journalist. He says everything is great but he looks unhappy. When pushed, he explains the reason with great hesitation, albeit humorously: his wife is having an extra-marital affair. Like the documentaries, the segment is in Oromiffa.

D Additional results

| | | After th | ne screening | 5 | | After | six months | | | After | five years | |
|--|----------------|----------------|-----------------------|------------------------------------|-------------|-----------|-----------------------|------------------------------------|----------------|-----------|-----------------------|------------------------------------|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | (9) | (10) | (11) | (12) |
| | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. |
| Summary indices: | | | | | | | | | | | | |
| Aspirations index | 0.04 | -0.04 | 0.09^{**} | 0.00 | 0.03 | 0.02 | 0.02 | 0.00 | 0.12^{**} | -0.03 | 0.15^{**} | 0.02 |
| | (0.05) | (0.05) | (0.04) | (1.00) | (0.05) | (0.05) | (0.05) | (1.01) | (0.06) | (0.06) | (0.06) | (1.00) |
| | [0.38] | [0.33] | [0.10] | 2003 | [0.53] | [0.92] | [0.74] | 2059 | $[0.04]^{**}$ | [0.96] | $[0.01]^{**}$ | 1904 |
| Expectations index | 0.23^{***} | 0.17^{***} | 0.06 | 0.00 | 0.07 | 0.01 | 0.06 | -0.00 | 0.21^{***} | -0.00 | 0.22^{***} | 0.01 |
| | (0.05) | (0.05) | (0.05) | (1.00) | (0.05) | (0.05) | (0.05) | (1.00) | (0.06) | (0.06) | (0.06) | (1.00) |
| | $[0.00]^{***}$ | $[0.00]^{***}$ | [0.22] | 2000 | [0.51] | [0.92] | [0.63] | 2055 | $[0.00]^{***}$ | [0.96] | $[0.00]^{***}$ | 1900 |
| Aspirations and expectations aggregate index | 0.13^{***} | 0.05 | 0.08^{*} | 0.00 | 0.05 | 0.01 | 0.04 | 0.00 | 0.19^{***} | -0.03 | 0.22^{***} | 0.01 |
| | (0.05) | (0.04) | (0.04) | (1.00) | (0.05) | (0.05) | (0.05) | (1.01) | (0.06) | (0.06) | (0.06) | (1.00) |
| | $[0.00]^{***}$ | [0.32] | [0.10] | 2003 | [0.51] | [0.92] | [0.67] | 2059 | $[0.00]^{***}$ | [0.96] | $[0.00]^{***}$ | 1904 |
| Aspirations: what would you like to achieve? | | | | | | | | | | | | |
| Aspired income (USD) PPP | 1769.95 | -3091.51 | 4861.46^{*} | 23993.62 | 804.84 | 1973.81 | -1168.97 | 21697.51 | 2505.32 | -643.61 | 3148.93^{*} | 15446.71 |
| | (2855.82) | (2432.07) | (2578.11) | (57202.10) | (2342.38) | (2335.89) | (2471.39) | (45062.36) | (1773.81) | (1593.19) | (1701.71) | (27746.43) |
| | [0.80] | [0.31] | $[0.09]^*$ | 1969 | [0.73] | [0.40] | [0.83] | 2022 | [0.22] | [0.69] | $[0.10]^*$ | 1864 |
| Aspired wealth (USD) PPP | -267.69 | -3839.81** | 3572.12^{**} | 13717.86 | -1455.44 | -1775.59 | 320.15 | 14542.17 | 1584.05 | 919.14 | 664.91 | 11978.72 |
| | (1926.34) | (1784.27) | (1561.63) | (38805.20) | (1649.39) | (1558.37) | (1545.68) | (31209.03) | (1301.17) | (1338.82) | (1395.59) | (21518.47) |
| | [0.89] | $[0.09]^*$ | $[0.07]^*$ | 1975 | [0.57] | [0.40] | [0.83] | 2034 | [0.22] | [0.69] | [0.63] | 1867 |
| Aspired education (years) | 0.16 | 0.07 | 0.08 | 14.12 | 0.27^{*} | 0.13 | 0.15 | 14.05 | 0.29^{*} | -0.10 | 0.39^{**} | 14.26 |
| | (0.12) | (0.12) | (0.12) | (2.39) | (0.14) | (0.14) | (0.14) | (2.61) | (0.16) | (0.16) | (0.16) | (2.60) |
| | [0.59] | [0.54] | [0.48] | 1943 | [0.18] | [0.40] | [0.83] | 1936 | [0.19] | [0.69] | $[0.05]^{**}$ | 1769 |
| Expectations: what do you expect in ten years? |) | | | | | | | | | | | |
| Expected income (USD) PPP | 1186.56*** | 523.91* | 662.66** | 5091.18 | 296.96 | 107.04 | 189.92 | 5156.43 | 309.11 | 70.89 | 238.22 | 3409.97 |
| | (336.89) | (308.54) | (334.04) | (5972.24) | (435.83) | (414.40) | (427.25) | (8652.46) | (190.28) | (173.05) | (189.95) | (2820.92) |
| | $[0.00]^{***}$ | [0.13] | $[0.08]^*$ | 1952 | [0.67] | [0.80] | [0.66] | 2007 | [0.10] | [0.68] | [0.21] | 1861 |
| Expected wealth (USD) PPP | 1019.14*** | 366.57 | 652.58^{*} | 4743.43 | 135.32 | -350.48 | 485.80 | 4781.23 | 596.06** | 226.46 | 369.61 | 4009.69 |
| | (319.44) | (310.29) | (336.56) | (5865.62) | (317.35) | (283.69) | (305.14) | (5803.38) | (250.19) | (242.98) | (247.91) | (3997.53) |
| | $[0.00]^{***}$ | [0.24] | $[0.08]^*$ | 1962 | [0.67] | [0.65] | [0.33] | 1993 | $[0.04]^{**}$ | [0.53] | [0.20] | 1853 |
| Expected education (years) | 0.58^{***} | 0.62^{***} | -0.04 | 13.33 | 0.33^{**} | 0.14 | 0.20 | 13.47 | 0.58^{**} | -0.32 | 0.90*** | 12.31 |
| | (0.17) | (0.17) | (0.15) | (3.61) | (0.16) | (0.17) | (0.17) | (3.05) | (0.26) | (0.26) | (0.27) | (3.88) |
| | $[0.00]^{***}$ | [0.00]*** | [0.78] | 1814 | [0.13] | [0.65] | [0.36] | 1791 | $[0.04]^{**}$ | [0.53] | [0.00]*** | 1706 |

Table A.1: ASPIRATIONS AND EXPECTATIONS AFTER THE SCREENING, AFTER SIX MONTHS, AND AFTER FIVE YEARS

Notes: OLS estimates of within-village treatment and placebo effects right after the video screenings (columns 1-2), after six months (columns 5-6), and after five years (columns 9-10). All columns control for village fixed effects and characteristics of the respondent: age, years of education, an indicator for being single, and an indicator for being male. Regressions control for baseline outcomes. Columns 3, 7, and 11 test for differences in parameters obtained in previous two columns. Column 4, 8, and 12 displays the control mean, standard deviation, and number of observations across rounds. The comparison group is households from the 64 treated villages that were not invited to any screening. After six months we have more observations than after the screening because we could not complete the post-screening surveys with 22 individuals that misued the screening and 81 individuals that attended them but left before the end of the videos. Heteroskedasticity-robust standard errors are clustered at the household-level in parentheses. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Variables are defined in Appendix E. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1. The unit of observations is the individual respondent (household head or their spouse). The number of observations and expectations are observations were in the individual respondent (household head or their spouse). The number of observations aris

| Table A.2: | ASPIRATIONS | AND | EXPECTATIONS | GAPS | AFTER | THE | SCREENING, | AFTER | SIX | MONTHS, | AND | AFTER | FIVE | YEAR |
|------------|-------------|-----|--------------|------|-------|-----|------------|-------|-----|---------|-----|-------|------|------|
|------------|-------------|-----|--------------|------|-------|-----|------------|-------|-----|---------|-----|-------|------|------|

| | | After th | ne screening | 5 | | After s | six months | | | After | five years | |
|--|----------------|----------------|-----------------------|------------------------------------|------------|-----------|-----------------------|------------------------------------|---------------|-----------|-----------------------|------------------------------------|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | (9) | (10) | (11) | (12) |
| | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. |
| Summary indices of gaps: | | | | | | | | | | | | |
| Aspirations (minus current at baseline) gap index | 0.04 | -0.05 | 0.09^{**} | 0.00 | 0.02 | 0.01 | 0.01 | -0.01 | 0.09 | -0.03 | 0.11^{**} | 0.01 |
| | (0.05) | (0.04) | (0.04) | (1.00) | (0.05) | (0.05) | (0.05) | (0.99) | (0.06) | (0.06) | (0.06) | (0.98) |
| | [0.35] | [0.28] | $[0.05]^{**}$ | 2003 | [0.69] | [0.90] | [0.81] | 2059 | [0.12] | [0.92] | $[0.04]^{**}$ | 1903 |
| Expectations (minus current at baseline) gap index | 0.27^{***} | 0.19^{***} | 0.08 | 0.00 | 0.06 | -0.01 | 0.07 | -0.01 | 0.18^{***} | -0.01 | 0.19^{***} | -0.00 |
| | (0.05) | (0.05) | (0.05) | (1.00) | (0.05) | (0.05) | (0.05) | (1.00) | (0.06) | (0.06) | (0.06) | (0.99) |
| | $[0.00]^{***}$ | $[0.00]^{***}$ | [0.14] | 2000 | [0.69] | [0.90] | [0.56] | 2054 | $[0.01]^{**}$ | [0.92] | $[0.01]^{***}$ | 1899 |
| Aspirations and expectations gap aggregate index | 0.16^{***} | 0.06 | 0.10^{**} | 0.00 | 0.03 | -0.01 | 0.04 | -0.02 | 0.15^{**} | -0.03 | 0.17^{***} | 0.01 |
| | (0.05) | (0.05) | (0.05) | (1.00) | (0.05) | (0.05) | (0.05) | (0.99) | (0.06) | (0.06) | (0.06) | (0.98) |
| | $[0.00]^{***}$ | [0.23] | $[0.05]^{**}$ | 2003 | [0.69] | [0.90] | [0.68] | 2059 | $[0.02]^{**}$ | [0.92] | $[0.01]^{***}$ | 1903 |
| Aspirations minus current level (at baseline) | | | | | | | | | | | | |
| Aspired income gap (USD) PPP | 1633.47 | -2947.78 | 4581.25^{*} | 22564.40 | 536.30 | 2086.97 | -1550.67 | 20395.07 | 2285.27 | -636.50 | 2921.77^{*} | 14399.79 |
| | (2869.83) | (2443.90) | (2592.40) | (56840.75) | (2354.50) | (2352.40) | (2489.08) | (44754.25) | (1765.05) | (1600.78) | (1697.30) | (27868.79) |
| | [0.85] | [0.34] | [0.12] | 1951 | [0.82] | [0.38] | [0.80] | 2003 | [0.29] | 0.69 | [0.13] | 1848 |
| Aspired wealth gap (USD) PPP | 113.91 | -3828.69** | 3942.60*** | 11728.91 | -1272.92 | -1502.58 | 229.65 | 12607.96 | 1093.34 | 781.82 | 311.53 | 10627.57 |
| | (1824.53) | (1666.66) | (1476.21) | (37173.77) | (1599.14) | (1511.07) | (1511.44) | (30017.37) | (1284.52) | (1331.10) | (1377.82) | (21451.27) |
| | [0.95] | [0.06]* | [0.02]** | 1957 | [0.64] | [0.38] | 0.88] | 2011 | [0.39] | 0.69 | [0.82] | 1850 |
| Aspired education gap (years) | 0.16 | 0.07 | 0.08 | 12.85 | 0.27^{*} | 0.13 | 0.15 | 12.84 | 0.29* | -0.10 | 0.39** | 13.00 |
| | (0.12) | (0.12) | (0.12) | (2.90) | (0.14) | (0.14) | (0.14) | (3.06) | (0.16) | (0.16) | (0.16) | (3.04) |
| | [0.59] | [0.54] | [0.48] | 1943 | [0.18] | [0.38] | [0.80] | 1936 | [0.19] | [0.69] | [0.05]** | 1769 |
| Expectations minus current level (at baseline) | | | | | | | | | | | . , | |
| Expected income gap (USD) PPP | 1215.91*** | 611.15** | 604.76* | 3912.53 | 239.97 | 129.13 | 110.84 | 4046.39 | 203.74 | 49.29 | 154.45 | 2337.89 |
| 1 01() | (333.87) | (304.66) | (336.68) | (5598.73) | (440.52) | (417.23) | (429.14) | (8651.84) | (187.27) | (168.90) | (185.86) | (2765.03) |
| | [0.00]*** | [0.07]* | [0.11] | 1931 | [0.75] | [0.76] | [0.80] | 1986 | [0.28] | [0.77] | [0.41] | 1843 |
| Expected wealth gap (USD) PPP | 949.06*** | 309.63 | 639.43* | 3196.99 | 97.35 | -330.60 | 427.95 | 3285.77 | 462.89* | 220.36 | 242.53 | 2611.47 |
| | (309.25) | (295.88) | (326.19) | (5065.44) | (308.26) | (279.20) | (297.98) | (5372.90) | (255.21) | (251.48) | (257.96) | (3947.68) |
| | 0.00]*** | [0.29] | [0.11] | 1950 | [0.75] | [0.65] | [0.36] | 1981 | [0.10] | [0.57] | [0.41] | 1840 |
| Expected education gap (years) | 0.58*** | 0.62*** | -0.04 | 12.06 | 0.33** | 0.14 | 0.20 | 12.29 | 0.58** | -0.32 | 0.90*** | 11.05 |
| - O • (v / | (0.17) | (0.17) | (0.15) | (3.92) | (0.16) | (0.17) | (0.17) | (3.51) | (0.26) | (0.26) | (0.27) | (4.24) |
| | 0.00]*** | [0.00]*** | [0.78] | 1814 | [0.13] | [0.65] | [0.36] | 1791 | [0.08]* | [0.57] | [0.00]*** | 1706 |

Notes: OLS estimates of within-village treatment and placebo effects right after the video screenings (columns 1-2), after six months (columns 5-6), and after five years (columns 9-10). Columns 3, 7, and 11 test for differences in parameters obtained in previous two columns. Column 4, 8, and 12 displays the control mean, standard deviation, and number of observations across rounds. The comparison group comprises households from the 64 treated villages that were not invited to any screening. After six months we have more observations than after the screening because we could not complete the post-screening surveys with 22 individuals that missed the screening and 81 individuals that attended them but left before the end of the videos. Heteroskedasticity-robust standard errors are clustered at the household-level in parentheses. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Variables are defined in Appendix E. All monetary values are in PPP-adjusted USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1. The unit of observations is the individual stepade or their spouse). The number of observations are defined in the note to Figure II. The aspirations (or expectations) gaps take the measure of aspirations (or expectations) and subtract the current level at baseline elicited for that same dimension. For the current level of education, we use the respondents' own elucation level. The aspirations and expectations gap aggregate index is made up of six outcomes: the reported gaps in income, wealth and years of education for the respondents' oldest child, for both aspirations and expectations.

| | | After | six months | | | five years | | | |
|--------------------------------------|---------------|----------|-------------|--------------|--------------|----------------|-----------|--------------|--|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | (8) | |
| | | | Treat re | Control mean | | | Treat rea | Control mean | |
| | Treatment | Placebo | nlacobo | (SD) | Treatment | Placebo | nlacobo | (SD) | |
| | | | placebo | Total obs. | | | placebo | Total obs. | |
| Has any savings | 0.03 | 0.01 | 0.02 | 0.39 | -0.02 | -0.01 | -0.01 | 0.21 | |
| | (0.02) | (0.02) | (0.02) | (0.49) | (0.02) | (0.02) | (0.02) | (0.41) | |
| | [0.31] | [0.82] | [0.58] | 2064 | [0.76] | [0.80] | [0.86] | 1909 | |
| Has outside savings | 0.03 | 0.03 | 0.00 | 0.26 | -0.01 | -0.01 | -0.00 | 0.20 | |
| | (0.02) | (0.02) | (0.02) | (0.44) | (0.02) | (0.02) | (0.02) | (0.40) | |
| | [0.31] | [0.68] | [0.95] | 2064 | [0.76] | [0.80] | [0.86] | 1909 | |
| Has any credit | 0.03 | -0.02 | 0.05^{*} | 0.34 | -0.03 | 0.02 | -0.06** | 0.33 | |
| | (0.03) | (0.03) | (0.03) | (0.47) | (0.03) | (0.03) | (0.03) | (0.47) | |
| | [0.31] | [0.79] | [0.29] | 2052 | [0.76] | [0.80] | [0.42] | 1897 | |
| Has any agricultural credit | | | | | 0.02^{***} | 0.01^{*} | 0.01 | 0.00 | |
| | | | | | (0.01) | (0.01) | (0.01) | (0.07) | |
| | | | | | $[0.06]^*$ | [0.34] | [0.73] | 1908 | |
| Total savings (USD) PPP | 19.82^{***} | 5.10 | 14.72^{*} | 24.37 | -0.21 | 0.69 | -0.90 | 17.87 | |
| | (6.98) | (5.37) | (7.68) | (80.68) | (3.82) | (3.67) | (3.72) | (64.07) | |
| | $[0.04]^{**}$ | [0.77] | [0.29] | 2002 | [0.96] | [0.89] | [0.86] | 1865 | |
| Total outside savings (USD) PPP | 2.19 | 1.78 | 0.41 | 8.88 | -1.05 | 2.19 | -3.24 | 16.50 | |
| | (1.46) | (1.47) | (1.59) | (25.07) | (3.61) | (3.65) | (3.74) | (59.48) | |
| | [0.31] | [0.68] | [0.90] | 2002 | [0.96] | [0.80] | [0.73] | 1863 | |
| Total credit (USD) PPP | 3.98^{*} | 0.56 | 3.42 | 19.65 | -8.93* | -0.69 | -8.24 | 39.65 | |
| | (2.29) | (2.27) | (2.45) | (40.09) | (5.03) | (5.15) | (5.04) | (87.98) | |
| | [0.31] | [0.84] | [0.49] | 2044 | [0.38] | [0.89] | [0.47] | 1897 | |
| Hypothetical loan (1 year USD) PPP | 76.54 | 312.51 | -235.97 | 2051.78 | -91.33 | -277.41^{**} | 186.07 | 1579.91 | |
| | (222.17) | (233.55) | (235.01) | (3135.83) | (138.69) | (138.08) | (133.83) | (1974.66) | |
| | [0.82] | [0.68] | [0.58] | 1101 | [0.76] | [0.34] | [0.47] | 1056 | |
| Hypothetical loan (5 years USD) PPP | 282.76 | -83.75 | 366.51 | 4406.63 | -24.04 | -329.73 | 305.69 | 3037.61 | |
| | (416.15) | (414.02) | (423.73) | (6112.10) | (250.30) | (239.31) | (232.71) | (3518.50) | |
| | [0.64] | [0.84] | [0.58] | 1086 | [0.96] | [0.56] | [0.47] | 1041 | |
| Hypothetical loan (10 years USD) PPP | 93.08 | 655.33 | -562.24 | 7552.52 | -414.45 | -644.33 | 229.88 | 5663.60 | |
| | (873.91) | (912.53) | (987.41) | (11830.81) | (633.47) | (597.53) | (613.18) | (7780.67) | |
| | [0.92] | [0.79] | [0.73] | 1104 | [0.76] | [0.70] | [0.86] | 1002 | |

Table A.3: SAVINGS AND CREDIT

Notes: OLS estimates of within-village treatment and placebo effects after six months (columns 1-2) and after five years (columns 5-6) of the intervention. Columns 3 and 7 test for differences in parameters obtained in previous two columns. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 and 8 display the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and characteristics of the respondent (or household head for household-level outcomes): age, years of education, an indicator for being single, and an indicator for being male. Regressions control for the baseline value of the outcome. Heteroskedaticity-robust standard errors are clustered at the household-level in parentheses. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are acclulated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Variables are defined in Appendix E. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1. Regressions control for baseline outcomes, except for agricultural credit that was not collected at baseline. The unit of observations varies slightly across rows because some respondent do not answer all questions and because we trim continuous outcomes above the 99th percentile. Has savings is an indicator set to one if the respondent has any savings, while Has outside savings is one if savings are held outside the home. Has credit indicates if the respondent borrowed from any formal or informal source in the past six months. Agricultural credit records the total amount of loans taken. Hypothetical loan repayable in 1, 5, or 10 years measures the amount the respondent would borrow, interest-free, repayable in those time-frames.

| | After 5 years | | | |
|---|---------------|---------|------------|--------------|
| | (1) | (2) | (3) | (4) |
| | | | (-) | Control mean |
| | Treatment | Placebo | Treat. vs. | (SD) |
| | | | placebo | Total obs. |
| Information index | -0.02 | -0.07 | 0.04 | 0.00 |
| | (0.07) | (0.07) | (0.07) | (1.00) |
| | [0.97] | 0.60 | [0.75] | 1104 |
| Income from trading | 0.01 | -0.03 | 0.04 | 0.23 |
| Ũ | (0.03) | (0.03) | (0.03) | (0.42) |
| | [0.97] | [0.60] | [0.48] | 1104 |
| Attended community meeting to discuss ag issues | 0.02 | -0.03 | 0.04 | 0.68 |
| | (0.03) | (0.03) | (0.03) | (0.47) |
| | [0.97] | [0.60] | [0.48] | 1089 |
| Requested or visited by ag expert | 0.05 | 0.01 | 0.04 | 0.43 |
| | (0.03) | (0.03) | (0.04) | (0.50) |
| | [0.97] | 0.85 | [0.48] | 1089 |
| Uses irrigation | -0.02 | -0.05* | 0.03 | 0.24 |
| 0 | (0.03) | (0.03) | (0.03) | (0.43) |
| | [0.97] | 0.60 | [0.52] | 1089 |
| Sought advice on fertilisers | 0.00 | 0.01 | -0.01 | 0.42 |
| | (0.03) | (0.03) | (0.03) | (0.49) |
| | [1.00] | [0.75] | [0.79] | 1023 |
| Sought advice on land preparation | 0.02 | 0.03 | -0.01 | 0.36 |
| and a second second best and the second | (0.03) | (0.03) | (0.03) | (0.48) |
| | [0.97] | [0.60] | [0.80] | 1079 |
| Sought advice on seeding | 0.01 | 0.02 | -0.01 | 0.42 |
| 0 | (0.03) | (0.03) | (0.03) | (0.49) |
| | [0.97] | [0.75] | [0.86] | 1078 |
| Number of visits by ag extension agent | -0.03 | 0.11 | -0.14* | 0.79 |
| | (0.08) | (0.08) | (0.08) | (1.26) |
| | [0.97] | 0.60 | [0.48] | 1089 |
| Visited demonstration plot in last year | 0.02 | 0.00 | 0.02 | 0.15 |
| * U | (0.02) | (0.02) | (0.03) | (0.36) |
| | [0.97] | [0.85] | [0.75] | 1089 |
| Produces cash crops | -0.02 | 0.04 | -0.06 | 0.61 |
| * | (0.03) | (0.03) | (0.04) | (0.49) |
| | [0.97] | [0.60] | [0.48] | 1089 |
| Uses pump | 0.00 | 0.01 | -0.01 | 0.01 |
| | (0.01) | (0.01) | (0.01) | (0.10) |
| | [1.00] | [0.60] | [0.48] | 1088 |
| Uses stone bands | -0.03 | -0.02 | -0.02 | 0.69 |
| | (0.03) | (0.03) | (0.03) | (0.46) |
| | [0.97] | [0.75] | [0.76] | 1104 |
| Applies water harvesting | 0.00 | -0.02 | 0.02 | 0.07 |
| | (0.02) | (0.02) | (0.02) | (0.26) |
| | [1.00] | [0.60] | [0.52] | 1104 |
| Uses crop rotation | -0.03 | -0.05 | 0.02 | 0.53 |
| - | (0.03) | (0.03) | (0.03) | (0.50) |
| | [0.97] | [0.60] | [0.76] | 1089 |
| Uses cattle | 0.01 | -0.04 | 0.05 | 0.70 |
| | (0.03) | (0.03) | (0.03) | (0.46) |
| | [0.97] | 0.60 | 0.48 | 1088 |

Table A.4: INFORMATION INDEX COMPONENTS

Notes: OLS estimates of within-village treatment effects after five years (columns 1-2). Column 3 tests for differences in parameters obtained in first two columns. All columns control for village fixed effects and characteristics of the household head: age, years of education, an indicator for being single, and an indicator for being male. Regressions control for the baseline value of the outcome. Outcome variables are listed on the left, and described in detail in the Appendix E.3.4. The unit of observation is the household. Regressions do not control for baseline outcomes as these outcomes were only collected at the five-year follow-up. Heteroskedasticity-robust standard errors are in parentheses. Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Column 4 displays the control mean, standard deviation, and total number of observations. The number of observations varies slightly across rows because some respondents do not answer all questions, though the indices aggregate all non-missing outcomes. The information index is an inverse-covariance-weighted average of the components, constructed following Anderson (2008), standardised relative to the within-village control group and also reported in Table 6. The components of the index are indicators equal to 1 if the household does the activity or action mentioned, except the number of visits received by an agricultural extension worker (the only non-binary component of the index).

| After five years | (1) | (2) | (3) | (4) |
|--|-----------|---------|-----------------------|------------------------------------|
| | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. |
| Gross revenue (USD PPP) | 104.37 | -56.11 | 160.48^{*} | 1468.82 |
| | (95.87) | (94.65) | (94.90) | (1273.97) |
| | [0.66] | [0.62] | [0.55] | 1061 |
| Revenue from crop production (USD PPP) | 21.93 | 21.16 | 0.77 | 383.70 |
| | (22.43) | (22.56) | (23.97) | (300.60) |
| | [0.66] | [0.62] | [0.97] | 1077 |
| Revenue from livestock rearing and produce (USD PPP) | -3.96 | -99.66 | 95.70 | 740.53 |
| | (73.81) | (71.95) | (71.74) | (1002.83) |
| | [0.96] | [0.62] | [0.55] | 1087 |
| Revenue from wage labour (USD PPP) | -3.06 | 6.08 | -9.14 | 25.86 |
| | (8.04) | (9.18) | (9.07) | (111.32) |
| | [0.84] | [0.62] | [0.63] | 1080 |
| Revenue from non-farm enterprises (USD PPP) | 21.34 | 14.93 | 6.41 | 159.94 |
| | (28.79) | (29.86) | (31.79) | (353.37) |
| | [0.69] | [0.62] | [0.97] | 1076 |
| Transfers and remittances (USD PPP) | 18.62 | 7.26 | 11.36 | 114.03 |
| | (16.86) | (14.76) | (18.06) | (180.99) |
| | [0.66] | [0.62] | [0.79] | 1079 |

Table A.5:REVENUE

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences between Columns 1 and 2. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and characteristics of the household head: age, years of education, an indicator for being single, and an indicator for being male. Heteroskedasticity-robust standard errors are in parentheses. The unit of observation is the household. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 99^{th} percentile. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1. Gross revenue includes revenues received from wage labour, livestock and poultry, crops, non-farm activities and transfers and remittances. Revenue from crops is computed as price times quantity sold in the last agricultural season. We capture production in both the short and long rains seasons. We use Central Statistical Agency data from the Oromiya region for December 2015 and January 2016 to compute the price of crops. Revenue from livestock rearing and produce includes all revenues from sale of livestock and poultry and the production of livestock goods in the last 12 months. We use the self-reported sale prices to construct the livestock revenue variable. Livestock revenue includes also own-consumption of animals, valued at sales prices. Revenue from wage labour is the sum of agricultural and non-agricultural wages received by the household over the past 12 months. Non-farm revenue is the number of months an activity was operated for times the monthly revenue from this activity, for a list of common activities. Transfer revenues are revenues from all sources of transfers: pension, remittances, public works and other assistance programs, and other transfers e.g. burial associations, in the last 12 months.

| $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | After six months | (1) | (2) | (3) | (4) |
|---|---|-----------------------|------------------|-----------------------|----------------------|
| Household head and spouse's labour effort: Daily minutes on family farm 7.16 -6.04 13.20* 299.07 (7.14) (7.40) (7.43) (216.87) [0.56] [0.83] [0.15] 1880 Daily minutes in leisure 6.27 0.45 5.83 796.25 (10.90) (11.14) (10.92) (185.84) [0.56] [0.57] 16.61 Savings and credit: Total savings (USD) PPP 19.82*** 5.10 14.72* 24.37 (6.98) (5.37) (7.68) (80.68) [0.02]** [0.63] [0.22] 2002 Total credit (USD) PPP 3.98* 0.56 3.42 19.65 [3.20]*4 Hypothetical loan (1 year USD) PPP 76.54 312.51 -235.97 2051.78 (222.17) (233.55) (235.01) (3135.83) [0.92] [0.63] [0.42] 1101 Hypothetical loan (10 years USD) PPP 9.36 655.33 -562.24 7552.52 (873.91) (912.53) | | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) |
| Daily minutes on family farm 7.16 -6.04 13.20* 299.07 (7.14) (7.40) (7.43) (216.87) $[0.56]$ $[0.83]$ $[0.15]$ 1880 Daily minutes in leisure 6.27 0.45 5.83 796.25 Savings and credit: $[0.56]$ $[0.59]$ $[0.59]$ 1861 Savings and credit: $[0.56]$ $[0.57]$ (7.68) (80.68) $[0.52]^{***}$ 5.10 14.72^* 24.37 (6.39) (5.37) (7.68) (80.68) $[0.02]^{**}$ $[0.63]$ $[0.22]$ 2002 Total credit (USD) PPP 3.98^* 0.56 3.42 19.65 (229) (2.27) (2.45) (40.09) $[0.16]$ $[0.30]$ 0.32 2004 Hypothetical loan (1 year USD) PPP 76.54 312.51 -235.97 2051.78 (222.17) (233.55) (235.01) (3135.83) $[0.92]$ $[0.63]$ $[0.42]$ 1101 | Household head and spouse's labour effort: | | | | 10001 005. |
| (7.14) (7.40) (7.43) (216.87) Daily minutes in leisure (6.27) 0.45 5.83 760.25 Savings and credit: (10.90) (11.14) (10.92) (185.84) Savings and credit: (6.98) (6.37) (7.68) (80.68) Savings and credit: (6.98) (6.37) (7.68) (80.68) Savings and credit: (11.14) (10.92) (10.92) (10.92) Total savings (USD) PPP 19.82*** 5.10 14.72* 24.37 (6.98) (5.37) (7.68) (80.68) (0.02]** (0.63] (0.22] 2002 Total credit (USD) PPP 3.98* 0.56 3.42 19.05) (313.83) [0.92] [0.63] [0.42] 1101 (1183.08) (0.92] (0.63] (0.77) (1183.08) (0.92] [0.63] [0.57] 1104 (1183.08) (0.92) (0.63] (0.57) 1104 Cohort 1: Children of post-primary school-going age at the five-year follow-up (aged 11-15 in school (0.05) (0.05) (0.073) (0.29) [0.71] [0.56] | Daily minutes on family farm | 7.16 | -6.04 | 13.20* | 299.07 |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | | (7.14) | (7.40) | (7.43) | (216.87) |
| $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | | [0.56] | [0.83] | [0.15] | 1880 |
| $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | Daily minutes in leisure | 6.27 | 0.45 | 5.83 | 796.25 |
| $ \begin{bmatrix} [0.56] & [0.97] & [0.59] & 1861 \\ \hline Savings and credit: \\ \hline Total savings (USD) PPP & 19.82*** 5.10 & 14.72* & 24.37 \\ (6.98) & (5.37) & (7.68) & (80.68) \\ [0.02]** & [0.63] & [0.22] & 2002 \\ \hline Total credit (USD) PPP & 3.98* & 0.56 & 3.42 & 19.65 \\ (2.29) & (2.27) & (2.45) & (40.09) \\ [0.16] & [0.80] & [0.32] & 2044 \\ \hline Hypothetical loan (1 year USD) PPP & 76.54 & 312.51 & -235.97 & 2051.78 \\ (222.17) & (233.55) & (235.01) & (3135.83) \\ [0.92] & [0.63] & [0.42] & 1101 \\ \hline Hypothetical loan (10 years USD) PPP & 93.08 & 655.33 & -562.24 & 7552.52 \\ (873.91) & (912.53) & (987.41) & (11830.81) \\ [0.92] & [0.63] & [0.57] & 1104 \\ \hline Cohort 1: Children of post-primary school-going age at the five-year follow-up (aged 11-15 at the time of the intervention) \\ \hline Children aged 11-15 in school & 0.07 & 0.03 & 0.04 & 0.56 \\ (0.05) & (0.04) & (0.05) & (0.73) \\ [0.20] & [0.71] & [0.56] & 1126 \\ \hline Daily minutes in school for children aged 11-15 & 15.66 & 6.33 & 9.33 & 188.71 \\ (15.85) & (15.29) & (15.94) & (248.36) \\ [0.32] & [0.71] & [0.56] & 1109 \\ \hline Daily minutes studying for children aged 11-15 & 9.95* & 2.16 & 7.78 & 58.11 \\ (5.72) & (5.79) & (6.17) & (86.58) \\ [0.20] & [0.71] & [0.56] & 1109 \\ \hline Cohort 2^{(a)}: Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) \\ \hline Children aged 7-10 in school & 0.11** & 0.05 & 0.06 & 0.60 \\ (0.05) & (0.05) & (0.05) & (0.73) \\ [0.8]^* & [0.53] & [0.48] & 1126 \\ \hline Daily minutes in school for children aged 7-10 & 23.52 & 11.56 & 11.96 & 198.10 \\ (16.61) & (16.63) & (16.79) & (250.25) \\ [0.23] & [0.53] & [0.48] & 1112 \\ \hline Daily minutes in school for children aged 7-10 & 1.51 & -2.96 & 4.47 & 450.8 \\ (0.75) & [0.53] & [0.48] & 1108 \\ \hline For all children \\ \hline For $ | U U | (10.90) | (11.14) | (10.92) | (185.84) |
| Savings and credit: 1 1 1 1 1 Total savings (USD) PPP 19.82*** 5.10 14.72* 24.37 (6.98) (5.37) (7.68) (80.68) [0.02]** [0.63] [0.22] 2002 Total credit (USD) PPP 3.08 6.56 3.42 19.65 (2.29) (2.27) (2.45) (40.09) [0.16] [0.80] [0.32] 2044 Hypothetical loan (1 year USD) PPP 7.65 3.32 (222.17) (23.355) (235.01) (3135.83) [0.92] [0.63] [0.42] 1101 Hypothetical loan (10 years USD) PPP 93.08 655.33 -562.24 7552.52 (873.91) 912.53) (987.41) (11830.81) [0.92] [0.63] [0.57] 1104 Cohort 1: Children of post-primary school-going age at the five-year follow-up (aged 11-15 at the time of the intervention) [0.20] [0.71] [0.56] 1109 Daily minutes in school for children aged 11-15 15.66 6.33 9.33 188.71 (15.72) (5.79) (6.17) (86.58) [0.20] | | [0.56] | [0.97] | [0.59] | 1861 |
| Total savings (USD) PPP 19.82*** 5.10 14.72^* 24.37 (6.98) (5.37) (7.68) (80.68) Total credit (USD) PPP 3.98* 0.56 3.42 19.65 (2.29) (2.27) (2.45) (40.09) [0.16] [0.80] [0.32] 2044 Hypothetical loan (1 year USD) PPP 76.54 312.51 -235.97 2051.78 (22.217) (233.55) (235.01) (3135.83) [0.92] [0.63] [0.42] 1101 Hypothetical loan (10 years USD) PPP 93.08 655.33 -562.24 7552.52 (873.91) (912.53) (987.41) (11830.81) [0.92] [0.63] [0.57] 1104 Cohort 1: Children of post-primary school-going age at the five-year follow-up (aged 11-15 at the time of the intervention) 0.07 0.03 0.04 0.56 Children aged 11-15 in school 0.07 0.03 0.04 0.56 1109 Daily minutes in school for children aged 11-15 15.59 (15.29) (15.94) (248.36) [0.22] [0.71] [0.56] 1109 1109 11 | Savings and credit: | | | | |
| (6.98) (5.37) (7.68) (80.68) Total credit (USD) PPP $(0.63]$ $(0.22]$ 2002 Total credit (USD) PPP (2.29) (2.27) (2.45) (40.09) (2.29) (2.27) (2.45) (40.09) $(0.16]$ (0.80) $(0.32]$ 2044 Hypothetical loan (1 year USD) PPP 76.54 312.51 -235.97 2051.78 (222.17) (233.55) (233.55) (235.01) (3135.83) $(0.92]$ $[0.63]$ $[0.42]$ 1101 Hypothetical loan (10 years USD) PPP 93.08 655.33 -562.24 7552.52 (873.91) (912.53) (987.41) (11830.81) $[0.92]$ $[0.63]$ $[0.57]$ 1104 Choidren of post-primary school-going age at the five-year follow-up $(aged 11-15$ at the time of the intervention) $[0.20]$ $[0.71]$ $[0.56]$ 1126 Daily minutes in school for children aged 11-15 9.56^{*} 2.16 7.78 58.11 (5.72) (5.73) (6.17) (86.58) $[0.20]$ $[0.71]$ | Total savings (USD) PPP | 19.82*** | 5.10 | 14.72^{*} | 24.37 |
| Image: bit is the standard state in the standard state is the s | | (6.98) | (5.37) | (7.68) | (80.68) |
| Total credit (USD) PPP 3.98^* 0.56 3.42 19.65 (2.29) (2.27) (2.45) (40.09) Hypothetical loan (1 year USD) PPP 76.54 312.51 -235.97 2051.78 (222.17) (233.55) (235.01) (3135.83) $[0.92]$ $[0.63]$ $[0.42]$ 1101 Hypothetical loan (10 years USD) PPP 93.08 655.33 -562.24 7552.52 (873.91) (912.53) (987.41) (11830.81) $[0.92]$ $[0.63]$ $[0.57]$ 1104 Cohort 1: Children of post-primary school-going age at the five-year follow-up (aged 11-15 in school 0.07 0.03 0.04 0.56 Children aged 11-15 in school 0.07 0.03 0.04 0.56 Daily minutes in school for children aged 11-15 15.66 6.33 9.33 188.71 (5.72) (5.79) (6.17) (86.58) $[0.20]$ $[0.71]$ $[0.56]$ 1109 Daily minutes studying for children aged 11-15 9.55^* 2.16 7.78 58.11 (5.72) (5.79) (6.17) $(86$ | | [0.02]** | [0.63] | [0.22] | 2002 |
| $\begin{array}{cccccccccccccccccccccccccccccccccccc$ | Total credit (USD) PPP | 3.98* | 0.56 | 3.42 | 19.65 |
| Image: height of the intervention of the interven | | (2.29) | (2.27) | (2.45) | (40.09) |
| Hypothetical loan (1 year USD) PPP 76.54 312.51 -235.97 2051.78 (222.17) (233.55) (235.01) (3135.83) Hypothetical loan (10 years USD) PPP 93.08 655.33 -562.24 752.52 (873.91) (912.53) (987.41) (11830.81) $[0.92]$ $[0.63]$ $[0.57]$ 1104 Cohort 1: Children of post-primary school-going age at the five-year follow-up (aged 11-15 in school 0.07 0.03 0.04 0.56 Children aged 11-15 in school 0.07 0.03 0.04 0.56 Daily minutes in school for children aged 11-15 15.66 6.33 9.33 188.71 (15.85) (15.29) (15.94) (248.36) $[0.32]$ $[0.71]$ $[0.56]$ 1109 Daily minutes studying for children aged 11-15 9.95^* 2.16 7.78 58.11 (5.72) (5.79) (6.17) (86.58) $[0.20]$ $[0.71]$ $[0.56]$ 1109 Daily minutes studying for children aged 7-10 25.22 11.56 11.96 198.10 Dail | | [0.16] | [0.80] | [0.32] | 2044 |
| $\begin{array}{c ccccccccccccccccccccccccccccccccccc$ | Hypothetical loan (1 year USD) PPP | 76.54 | 312.51 | -235.97 | 2051.78 |
| $ \begin{bmatrix} 0.92 \\ 0.63 \\ 0.42 \\ 0.42 \\ 0.42 \\ 0.52 \\ 0.53 \\ 0.57$ | | (222.17) | (233.55) | (235.01) | (3135.83) |
| Hypothetical loan (10 years USD) PPP 93.08 655.33 -562.24 7552.52 (873.91) (912.53) (987.41) (11830.81) [0.92] [0.63] [0.57] 1104 Cohort 1: Children of post-primary school-going age at the five-year follow-up (aged 11-15 at the time of the intervention) Children aged 11-15 in school 0.07 0.03 0.04 0.56 (0.20) [0.71] [0.56] 1126 Daily minutes in school for children aged 11-15 15.66 6.33 9.33 188.71 (15.85) (15.29) (15.94) (248.36) [0.32] [0.71] [0.56] 1109 Daily minutes studying for children aged 11-15 9.95* 2.16 7.78 58.11 (5.72) (5.79) (6.17) (86.58) [0.20] [0.71] [0.56] 1109 Cohort $2^{(a)}$: Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) Children aged 7-10 0.55 0.06 0.60 (0.05) (0.05) <td< td=""><td></td><td>[0.92]</td><td>[0.63]</td><td>[0.42]</td><td>1101</td></td<> | | [0.92] | [0.63] | [0.42] | 1101 |
| $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$ | Hypothetical loan (10 years USD) PPP | 93.08 | 655.33 | -562.24 | 7552.52 |
| $\begin{bmatrix} [0.92] & [0.63] & [0.57] & 1104 \\ 1104 \\ \hline Cohort 1: Children of post-primary school-going age at the five-year follow-up (aged 11-15 at the time of the intervention) \\ \hline Children aged 11-15 at the time of the intervention) \\ \hline Children aged 11-15 in school & 0.07 & 0.03 & 0.04 & 0.56 \\ (0.05) & (0.04) & (0.05) & (0.73) \\ [0.20] & [0.71] & [0.56] & 1126 \\ \hline Daily minutes in school for children aged 11-15 & 15.66 & 6.33 & 9.33 & 188.71 \\ (15.85) & (15.29) & (15.94) & (248.36) \\ [0.32] & [0.71] & [0.56] & 1109 \\ \hline Daily minutes studying for children aged 11-15 & 9.95 & 2.16 & 7.78 & 58.11 \\ (5.72) & (5.79) & (6.17) & (86.58) \\ [0.20] & [0.71] & [0.56] & 1109 \\ \hline Cohort 2^{(a)}: Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) \\ \hline Children aged 7-10 in school & 0.11^{**} & 0.05 & 0.06 & 0.60 \\ (0.05) & (0.05) & (0.05) & (0.73) \\ [0.08]^{*} & [0.53] & [0.48] & 1126 \\ \hline Daily minutes in school for children aged 7-10 & 23.52 & 11.56 & 11.96 & 198.10 \\ (16.61) & (16.63) & (16.79) & (250.25) \\ [0.23] & [0.53] & [0.48] & 1112 \\ \hline Daily minutes studying for children aged 7-10 & 1.51 & -2.96 & 4.47 & 45.08 \\ (4.83) & (4.69) & (4.72) & (70.78) \\ [0.75] & [0.53] & [0.48] & 11108 \\ \hline For all children \\ \hline Schooling expenditure (USD) PPP & 6.47^{*} & 4.37 & 2.09 & 37.75 \\ \hline (2.50) & (2.50) & (2.50) \\ \hline \end{tabular}$ | | (873.91) | (912.53) | (987.41) | (11830.81) |
| Cohort 1: Children of post-primary school-going age at the five-year follow-up (aged 11-15 at the time of the intervention) Children aged 11-15 at the time of the intervention) Children aged 11-15 at the time of the intervention) Children aged 11-15 in school 0.07 0.03 0.04 0.56 (0.05) (0.04) (0.05) (0.73) [0.20] $[0.71]$ $[0.56]$ 1126 Daily minutes in school for children aged 11-15 15.66 6.33 9.33 188.71 Daily minutes studying for children aged 11-15 15.66 6.33 9.33 188.71 Daily minutes studying for children aged 11-15 9.95^* 2.16 7.78 58.11 Daily minutes in school 0.11^{**} 0.05 0.60 0.060 Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) Children aged 7-10 at the time of the intervention Children aged 7-10 0.11^{**} 0.05 0.60 0.60 Daily minutes in school for children aged 7-10 23.52 11.56 <td></td> <td>[0.92]</td> <td>[0.63]</td> <td>[0.57]</td> <td>1104</td> | | [0.92] | [0.63] | [0.57] | 1104 |
| Children aged 11-15 in school 0.07 0.03 0.04 0.56 (0.05) (0.04) (0.05) (0.73) [0.20] [0.71] [0.56] 1126 Daily minutes in school for children aged 11-15 15.66 6.33 9.33 188.71 (15.85) (15.29) (15.94) (248.36) [0.32] [0.71] [0.56] 1109 Daily minutes studying for children aged 11-15 9.95* 2.16 7.78 58.11 (5.72) (5.79) (6.17) (86.58) [0.20] [0.71] [0.56] 1109 0.05 (0.05) (0.05) (0.73) Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) [0.20] [0.71] [0.56] 1109 Children aged 7-10 in school 0.11^{**} 0.05 (0.05) (0.73) [0.68]* [0.53] [0.48] 1126 Daily minutes in school for children aged 7-10 23.52 11.56 11.96 198.10 (16.61) (16.63) (16.79) (250.25) [0.73] [0.53] [0.48] 1112 Daily minutes studying for c | Cohort 1: Children of post-primary school-going (aged 11–15 at the time of the interve | age at the ention) | five-year | follow-up | |
| Children aged 11-15 in school $(0.01 - 0.05 - 0.04 - 0.05)$ (0.05) - 0.04 - 0.05 - 0.04 - 0.05 (0.05) - 0.04 - 0.05 $(0.73)[0.20] - [0.71] - [0.56] - 1126Daily minutes in school for children aged 11-15 - 15.66 - 6.33 - 9.33 - 188.71(15.85) - (15.29) - (15.94) - (248.36)[0.32] - [0.71] - [0.56] - 1109Daily minutes studying for children aged 11-15 - 9.95* - 2.16 - 7.78 - 58.11(5.72) - (5.79) - (6.17) - (86.58)[0.20] - [0.71] - [0.56] - 1109Cohort 2^{(a)}: Children of primary school-going age at the five-year follow-up(aged 7-10 at the time of the intervention)Children aged 7-10 in school 0.11^{**} - 0.05 - 0.06 - 0.60(0.05) - (0.05) - (0.05) - (0.73)[0.08]^* - [0.53] - [0.48] - 1126Daily minutes in school for children aged 7-10 - 23.52 - 11.56 - 11.96 - 198.10(16.61) - (16.63) - (16.79) - (250.25)[0.23] - [0.53] - [0.48] - 1112Daily minutes studying for children aged 7-10 - 1.51 - 2.96 - 4.47 - 45.08(4.83) - (4.69) - (4.72) - (70.78)[0.75] - [0.53] - [0.48] - 1108For all childrenSchooling expenditure (USD) PPP - 6.47^* - 4.37 - 2.09 - 37.75 - (2.59) - (2.06) - (51.30)$ | Children aged 11 15 in school | 0.07 | 0.03 | 0.04 | 0.56 |
| $ \begin{bmatrix} (0.03) & (0.04) & (0.05) & (0.73) \\ [0.20] & [0.71] & [0.56] & 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1126 \\ 1109 \\ 1232 & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.32] & [0.71] & [0.56] & 1109 \\ 10.33] & [0.48] & 1126 \\ 126 \\ 126$ | Children aged 11-15 in school | (0.07) | (0.03) | (0.04) | (0.72) |
| Daily minutes in school for children aged 11-15 $[0.20]$ $[0.71]$ $[0.53]$ 1120 Daily minutes in school for children aged 11-15 15.66 6.33 9.33 188.71 (15.85) (15.29) (15.94) (248.36) $[0.32]$ $[0.71]$ $[0.56]$ 1109 Daily minutes studying for children aged 11-15 9.95^* 2.16 7.78 58.11 (5.72) (5.79) (6.17) (86.58) $[0.20]$ $[0.71]$ $[0.56]$ 1109 Cohort $2^{(a)}$: Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) (0.05) (0.05) (0.05) (0.73) Children aged 7-10 in school 0.11^{**} 0.05 0.06 0.60 $(0.08]^*$ $[0.53]$ $[0.48]$ 1126 Daily minutes in school for children aged 7-10 23.52 11.56 11.96 198.10 (16.61) (16.63) (16.79) (250.25) $[0.23]$ $[0.53]$ $[0.48]$ 1112 Daily minutes studying for children aged 7-10 1.51 -2.96 4.47 | | [0.00] | (0.04) [0.71] | [0.05] | (0.75) |
| $\begin{array}{c} \text{(15.85)} & (15.29) & (15.94) & (248.36) \\ & [0.32] & [0.71] & [0.56] & 1109 \\ & [0.32] & [0.71] & [0.56] & 1109 \\ & [0.32] & [0.71] & [0.56] & 1109 \\ & [0.32] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & [0.20] & [0.71] & [0.56] & 1109 \\ & \\ & (aged 7-10 at the time of the intervention) \\ \hline \\ Children aged 7-10 at the time of the intervention \\ & (0.05) & (0.05) & (0.05) & (0.73) \\ & [0.08]^* & [0.53] & [0.48] & 1126 \\ & [0.23] & [0.53] & [0.48] & 1112 \\ & \\ Daily minutes in school for children aged 7-10 & 23.52 & 11.56 & 11.96 & 198.10 \\ & (16.61) & (16.63) & (16.79) & (250.25) \\ & [0.23] & [0.53] & [0.48] & 1112 \\ & Daily minutes studying for children aged 7-10 & 1.51 & -2.96 & 4.47 & 45.08 \\ & (4.83) & (4.69) & (4.72) & (70.78) \\ & [0.75] & [0.53] & [0.48] & 1108 \\ \hline For all children \\ \hline \\ For all children \\ \hline \\ $ | Daily minutes in school for children aged 11-15 | [0.20] 15.66 | 6 33 | 0.33 | 188 71 |
| $\begin{bmatrix} (10.30) & (10.32) & (10.33) & (10.33) \\ [0.32] & [0.71] & [0.56] & 1109 \\ 9.95^* & 2.16 & 7.78 & 58.11 \\ (5.72) & (5.79) & (6.17) & (86.58) \\ [0.20] & [0.71] & [0.56] & 1109 \\ \hline \\ Cohort \mathcal{I}^{(a)}: Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention)Children aged 7-10 in school 0.11^{**} & 0.05 & 0.06 & 0.60 \\ (0.05) & (0.05) & (0.05) & (0.73) \\ [0.08]^* & [0.53] & [0.48] & 1126 \\ 1126 \\ Daily minutes in school for children aged 7-10 & 23.52 & 11.56 & 11.96 & 198.10 \\ (16.61) & (16.63) & (16.79) & (250.25) \\ [0.23] & [0.53] & [0.48] & 1112 \\ Daily minutes studying for children aged 7-10 & 1.51 & -2.96 & 4.47 & 45.08 \\ (4.83) & (4.69) & (4.72) & (70.78) \\ [0.75] & [0.53] & [0.48] & 1108 \\ For all children \\ \hline \\ Schooling expenditure (USD) PPP & 6.47^* & 4.37 & 2.09 & 37.75 \\ (2.52) & (2.62) & (2.66) & (51.20) \\ \hline \\$ | Daily minutes in school for emitten aged 11-19 | (15.00) | (15.29) | (15.94) | (248.36) |
| Daily minutes studying for children aged 11-15 9.95^* 2.16 7.78 58.11 (5.72) (5.79) (6.17) (86.58) $[0.20]$ $[0.71]$ $[0.56]$ 1109 Cohort $2^{(a)}$: Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) Children aged 7-10 at the time of the intervention Children aged 7-10 in school 0.11^{**} 0.05 0.06 0.60 0.05 (0.05) (0.05) (0.05) (0.73) $[0.08]^*$ $[0.53]$ $[0.48]$ 1126 Daily minutes in school for children aged 7-10 23.52 11.56 11.96 198.10 Daily minutes studying for children aged 7-10 1.51 -2.96 4.47 45.08 (4.83) (4.69) (4.72) (70.78) $[0.75]$ $[0.53]$ $[0.48]$ 1108 For all children Schooling expenditure (USD) PPP 6.47^* 4.37 2.09 37.75 (2.52) (2.62) (2.60) (51.30) | | [0.32] | [0.71] | [0.56] | 1109 |
| Early limited studying for children (5.72) (5.79) (6.17) (86.58) $(0.20]$ $[0.71]$ $[0.56]$ 1109 Cohort $2^{(a)}$: Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) $(0.11^{**}$ 0.05 0.06 0.60 Children aged 7-10 in school 0.11^{**} 0.05 0.06 0.60 Daily minutes in school for children aged 7-10 23.52 11.56 11.96 198.10 Daily minutes studying for children aged 7-10 23.52 11.56 11.96 198.10 Daily minutes studying for children aged 7-10 1.51 -2.96 4.47 45.08 (4.83) (4.69) (4.72) (70.78) $[0.75]$ $[0.53]$ $[0.48]$ 1108 For all children Schooling expenditure (USD) PPP 6.47^* 4.37 2.09 37.75 (2.52) (2.62) (2.66) (51.30) | Daily minutes studying for children aged 11-15 | [0.02] 9.95* | 2.16 | 7.78 | 58.11 |
| $\begin{bmatrix} (0.112) & (0.112) & (0.112) \\ [0.20] & [0.71] & [0.56] & 1109 \\ \hline \\ [0.20] & [0.71] & [0.56] & 1109 \\ \hline \\ \\ [0.20] & [0.71] & [0.56] & 1109 \\ \hline \\ \\ \\ [0.20] & [0.71] & [0.56] & 1109 \\ \hline \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ $ | Daily minutes studying for aniaron aged 11 15 | (5.72) | (5.79) | (6.17) | (86.58) |
| Cohort $\mathscr{Z}^{(a)}$: Children of primary school-going age at the five-year follow-up (aged 7-10 at the time of the intervention) Image: Children aged 7-10 at the time of the intervention) Children aged 7-10 in school 0.11^{**} 0.05 0.06 0.60 Children aged 7-10 in school 0.11^{**} 0.05 0.05 (0.73) Daily minutes in school for children aged 7-10 23.52 11.56 11.96 198.10 Daily minutes studying for children aged 7-10 23.52 11.56 11.96 198.10 Daily minutes studying for children aged 7-10 1.51 -2.96 4.47 45.08 (4.83) (4.69) (4.72) (70.78) $[0.75]$ $[0.53]$ $[0.48]$ 1108 For all children Schooling expenditure (USD) PPP 6.47^* 4.37 2.09 37.75 (2.52) (2.62) (2.62) (51.20) (51.20) | | [0.20] | [0.71] | [0.56] | 1109 |
| $\begin{array}{c} (aged \ 7-10 \ at the time of the intervention) \\ \hline \\ Children aged \ 7-10 \ in school \\ (0.05) \\ (0.13) \\ [0.23] \\ [0.53] \\ [0.48] \\ 1112 \\ 0.48] \\ 1112 \\ 0.48] \\ 1112 \\ 0.48] \\ 1112 \\ 0.48] \\ 1112 \\ 0.48] \\ (16.61) \\ (16.63) \\ (16.79) \\ (250.25) \\ [0.23] \\ [0.23] \\ [0.53] \\ [0.48] \\ 1112 \\ 0.48] \\ 1112 \\ 0.48] \\ 1112 \\ 0.48] \\ 1112 \\ 0.48] \\ 1108 \\ For all children \\ \hline \\ Schooling expenditure (USD) PPP \\ \hline \\ (2.52) \\ (2.52) \\ (2.62$ | Cohort $\mathcal{Z}^{(a)}$: Children of primary school-going as | ge at the fit | ve-year fol | low-up | |
| Children aged 7-10 in school 0.11^{**} 0.05 0.06 0.60 (0.05) (0.05) (0.05) (0.05) (0.73) Daily minutes in school for children aged 7-10 23.52 11.56 11.96 198.10 Daily minutes studying for children aged 7-10 23.52 11.56 11.96 198.10 Daily minutes studying for children aged 7-10 1.51 -2.96 4.47 45.08 (4.83) (4.69) (4.72) (70.78) $[0.75]$ $[0.53]$ $[0.48]$ 1108 | (aged 7–10 at the time of the inter | vention) | | * | |
| $\begin{array}{c} \text{(0.05)} & (0.05) & (0.05) & (0.73) \\ (0.05) & (0.05) & (0.05) & (0.73) \\ [0.08]^* & [0.53] & [0.48] & 1126 \\ 23.52 & 11.56 & 11.96 & 198.10 \\ (16.61) & (16.63) & (16.79) & (250.25) \\ [0.23] & [0.53] & [0.48] & 1112 \\ 1.51 & -2.96 & 4.47 & 45.08 \\ (4.83) & (4.69) & (4.72) & (70.78) \\ [0.75] & [0.53] & [0.48] & 1108 \end{array}$ | Children aged 7-10 in school | 0.11** | 0.05 | 0.06 | 0.60 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | | (0.05) | (0.05) | (0.05) | (0.73) |
| Daily minutes in school for children aged 7-10 $\begin{bmatrix} 10.57 \\ 23.52 \end{bmatrix}$ $\begin{bmatrix} 10.57 \\ 11.56 \end{bmatrix}$ 11.96 \\ 198.10 \\ (16.61) (16.63) (16.79) (250.25) \\ \begin{bmatrix} 0.23 \\ 250.25 \end{bmatrix} Daily minutes studying for children aged 7-10 $\begin{bmatrix} 1.51 \\ 1.56 \end{bmatrix}$ $\begin{bmatrix} 0.48 \\ 1112 \end{bmatrix}$ Daily minutes studying for children aged 7-10 $\begin{bmatrix} 1.51 \\ 1.51 \end{bmatrix}$ $\begin{bmatrix} -2.96 \\ 4.47 \end{bmatrix}$ $\begin{bmatrix} 4.47 \\ 45.08 \\ (4.83) \end{bmatrix}$ <i>For all children</i> $\begin{bmatrix} 0.75 \\ 0.53 \end{bmatrix}$ $\begin{bmatrix} 0.48 \\ 0.48 \end{bmatrix}$ 1108 Schooling expenditure (USD) PPP $\begin{bmatrix} 6.47^* \\ 4.37 \end{bmatrix}$ $\begin{bmatrix} 2.09 \\ 2.62 \end{bmatrix}$ $\begin{bmatrix} 7.75 \\ 2.98 \end{bmatrix}$ | | [0.08]* | [0.53] | [0.48] | 1126 |
| $\begin{array}{c} (16.61) & (16.63) & (16.79) & (250.25) \\ [0.23] & [0.53] & [0.48] & 1112 \\ 1.51 & -2.96 & 4.47 & 45.08 \\ (4.83) & (4.69) & (4.72) & (70.78) \\ [0.75] & [0.53] & [0.48] & 1108 \end{array}$ For all children Schooling expenditure (USD) PPP $\begin{array}{c} 6.47^* & 4.37 & 2.09 & 37.75 \\ (2.52) & (2.62) & (2.96) & (51.20) \end{array}$ | Daily minutes in school for children aged 7-10 | 23.52 | 11.56 | 11.96 | 198.10 |
| Daily minutes studying for children aged 7-10 $\begin{bmatrix} 0.23 \\ .53 \end{bmatrix}$ $\begin{bmatrix} 0.48 \\ .1112 \end{bmatrix}$ Daily minutes studying for children aged 7-10 1.51 -2.96 4.47 45.08 (4.83) (4.69) (4.72) (70.78) $[0.75]$ $[0.53]$ $[0.48]$ 1108 For all children Schooling expenditure (USD) PPP 6.47^* 4.37 2.09 37.75 (2.52) (2.62) (2.06) (51.20) | | (16.61) | (16.63) | (16.79) | (250.25) |
| Daily minutes studying for children aged 7-10 1.51 -2.96 4.47 45.08 (4.83) (4.69) (4.72) (70.78) $[0.75]$ $[0.53]$ $[0.48]$ 1108 For all children Schooling expenditure (USD) PPP 6.47^* 4.37 2.09 37.75 (2.52) (2.62) (2.06) (51.20) | | [0.23] | [0.53] | [0.48] | 1112 |
| $ \begin{array}{c} (4.83) & (4.69) & (4.72) & (70.78) \\ \hline [0.75] & [0.53] & [0.48] & 1108 \end{array} \\ \hline \\ Schooling expenditure (USD) PPP & 6.47^* & 4.37 & 2.09 & 37.75 \\ (2.52) & (2.62) & (2.06) & (51.20) \end{array} $ | Daily minutes studying for children aged 7-10 | 1.51 | -2.96 | 4.47 | 45.08 |
| For all children $[0.75]$ $[0.53]$ $[0.48]$ 1108 Schooling expenditure (USD) PPP 6.47^* 4.37 2.09 37.75 (2.52) (2.62) (2.06) (51.20) | | (4.83) | (4.69) | (4.72) | (70.78) |
| For all children 5 Schooling expenditure (USD) PPP 6.47* 4.37 2.09 37.75 (2.52) (2.62) (2.66) (51.20) | | [0.75] | [0.53] | [0.48] | 1108 |
| Schooling expenditure (USD) PPP 6.47^* 4.37 2.09 37.75 (2.52) (2.62) (2.06) (51.20) | For all children | | | | |
| (2.52) (2.62) (2.62) (2.100) | Schooling expenditure (USD) PPP | 6.47* | 4.37 | 2.09 | 37.75 |
| (0.02) (0.02) (0.90) (01.09) | | (3.52) | (3.62) | (3.96) | (51.39) |
| $[0.07]^*$ $[0.23]$ $[0.60]$ 1107 | | $[0.07]^{*}$ | [0.23] | [0.60] | 1107 |

Table A.6: Economic changes after six months

Notes: OLS estimates of within-village treatment and placebo effects six months after the intervention (columns 1-2). Column 3 tests for differences in parameters obtained in first two columns. The comparison group is households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and characteristics of the household head (or of the individual-respondent for the effort outcomes): age, years of education, an indicator for being single, and an indicator for being male. All regressions control for the baseline value of the outcome. Heteroskedasticity-robust standard errors are in parentheses (or clustered at the householdlevel for the effort outcomes). Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., * ** at 5 pct., and *** at 1 pct. level. Variables are defined in the note to Figure I and Appendix E. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1. The unit of observation is the individual household head or spouse for effort, total savings and total credit and the household for other variables. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous outcomes above the 99^{th} percentile. We examine all households in the sample, including 71 households who have no children in this age group in any of the rounds, to ensure the sample is comparable with other results. Cohort $2^{(a)}$ is not directly comparable to cohort 2 in Table 4, because some of the children in cohort 2 were not of primary school-going age at the time of six-months follow-up and as they would have been between 2 and 6 years old; we did not collect data for children in this age range.

E Data and measures

This section provides additional details on the construction of variables used in the paper.

E.1 Conversions from Ethiopian birr (ETB) to USD PPP

The survey collected data in Ethiopian birr (ETB) at the time of each survey. All monetary values in the tables and figures are displayed in 2016 USD PPP, to match the time of the five year endline. To convert baseline and six month follow-up values to 2016 prices, we divide the reported values in ETB by the monthly non-food national consumer price index (CPI) series (averaged over the months in which our survey took place and rebased so that it was equal to 1 in January 2016, the midpoint in our endline survey). We use the Central Statistical Authority publicly-available CPI reports.⁵⁶ For baseline, we divide the monetary values by 0.514. For midline, we divide the monetary values by 0.592.

To convert 2016 values to USD PPP, we use an exchange rate of 8.67 ETB per 1 USD PPP, the World Bank PPP conversion factor for private consumption in 2016. The price level ratio of PPP conversion factor (GDP) to ETB market exchange rate for 2016 was 0.41.⁵⁷

E.2 Agricultural investment

E.2.1 Modern crop and livestock inputs

Spending on crop inputs includes expenditure on seeds (bartered or purchased), fertiliser, herbicides, tractor hire and other non-labour inputs in the last long rains season.⁵⁸

To capture labour inputs in agriculture, we record the number of person-days of family and hired labour for crop agriculture in the last long rains season. This is collected by plot and crop and summed. To value this labour, we multiply by the median male wage for each village across all crop-related activities (i.e. seeding, planting, weeding, harvesting). Female wages are rarely measured, reflecting that most wage labour in agriculture is male. If a village wage is not available, we use the *kebele*-level (district-level) wage. If there is no *kebele*-level wage reported, we use the sample median of 50 ETB per day (about \$5.76 PPP per day) for that *kebele*.

Spending on livestock and poultry inputs includes expenditure on the purchase of inputs required for livestock in the past 12 months: feed, veterinary supplies, and hired labour.

E.2.2 Land

Total land area under cultivation is the area cultivated by the household across all plots in the last long rainy season. It excludes land rented out but includes land rented or sharecropped in. Areas are given in local units and converted into hectares.

^{56.}https://web.archive.org/web/20191115152931/http://www.csa.gov.et/price-indices/ consumer-price-index/category/109-cpi-2016?limitstart=0, accessed 17/08/2021.

^{57.} https://data.worldbank.org/topic/economy-and-growth?view=chart, accessed 27/08/2019.

^{58.} We only have prices for seed purchases by some households. We use either the household-level purchase price or, if not available, the sample level median of seed price. We use the price of white teff seeds for tikur teff, grass pea for cow peas, zengada for oats, and an average of wheat and barley seeds for wasira.

E.2.3 Labour supply

Labour supply and time allocation were measured using a household roster, completed by the household head for all household members for a typical day. Minutes worked includes work on the family farm, looking after livestock or other family business, as well as paid work for someone not in the household. Minutes of leisure include play time or general leisure (including time taken eating, drinking, bathing, sleeping). The category not reported involves all domestic chores (such as fetching water, firewood, cleaning, cooking, washing, shopping) and caring for others, including children and ill household members. For endline, for example, the household head was asked for each household member: "Typically, during October 2015 (*Tikimt* 2008 Ethiopian Calendar), how many hours per day did [household member] spend on the following activities?". We collect fractions of an hour, so express the total time in minutes per adult per day. As we expect the household head to be most knowledgeable about themselves and their spouse, we only include their data. Labour supply effects are robust to including all adult members of the household (available upon request).

Time allocation is collected for a period close to the survey time to minimise recall bias. It is collected for October for the five-year follow-up. This is the harvest period, a busy time on the farm. It is collected in March for the six-month follow-up. This is during the short rains, when there are activities on the farm (such as land preparation and pasture regeneration) but likely less intense than during the harvest period.

E.2.4 Livestock

The value of livestock and poultry is the sum of the value of all livestock varieties owned by the household. We construct prices using the sale prices reported by households for each variety of livestock. If the household has not sold the type of livestock it owns in the last 12 months, we use the first available median of sale prices from other households at the village, *kebele* or screening site level. If a price is above the 99th percentile, we replace it with the first available of the median price at the village, *kebele* or screening site level.

E.3 Beliefs, preferences, and information

E.3.1 Time preferences

We use the measurement tool from Ashraf, Karlan, and Yin (2006), in a similar context with participants with low literacy levels. We ask individuals to consider a situation in which they were about to receive a gift. They are first asked three questions in the "near-term" frame:

- 1. Would they prefer the gift of 100 ETB today or could instead choose to receive a gift of 125 ETB in one month?
- 2. If they answer 100 ETB to question 1, they are asked if they prefer 100 ETB today or 150 ETB in one month.
- 3. Individuals are then asked how much they would need to receive to wait one month for the payment instead of receiving 100 ETB today, with a ceiling of 1,000 ETB, implying

a discount factor of at least 0.1.

We create three indicator variables as crude measures of an individual's discount rate, the extent to which they discount rewards when they are in the future:

- Patient: Individuals who select 125 ETB over 100 ETB in Question 1.
- Slightly Impatient: Individuals who select 150 ETB over 100 ETB in Question 2 (but did not select 125 ETB over 100 ETB).
- Very Impatient: Individuals who need to receive over 150 ETB.⁵⁹

We also capture whether individuals' choices are consistent with quasi-hyperbolic discounting (rather than exponential discounting). We ask the first two questions, but over a more *distant* time frame (one vs two months). As in Ashraf, Karlan, and Yin (2006), we create two indicators. Those who are coded as "Present-biased" or "Hyperbolic" choose the immediate reward in the near term frame and the delayed reward in the distant frame. Those who are coded as "Patient now and impatient later" choose the delayed reward in the near term frame and the immediate reward in the distant frame. This could arise if individuals have funds now, but think it is likely they will be liquidity-constrained in two months time (for example, due to seasonality). The proportion of our sample (in the control group) who are present biased is 34 per cent six months after the baseline, compared to 28 per cent in Ashraf, Karlan, and Yin (2006), and the proportion who are "Patient now and impatient later" is 22 per cent, compared to 20 per cent. The remaining 44 per cent of our sample gave responses consistent with exponential discounting.

We note increases in the portion of the sample who are impatient over time (it increases from 68 to 80 per cent over five years) and who are present-biased (from 34 to 53 per cent). This could be because we neglected to alter our measures to account for inflation. The increase in impatience is consistent with the monetary reward for waiting being worth less in real terms at endline than at the six-months follow-up. This would likely affect all treatment groups similarly, so should not jeopardise estimation of treatment effects.

While measures of risk and time preferences are based on hypothetical questions, not incentivised measures, recent work suggests this does not affect answers (Ubfal, 2016; Falk et al., 2018).

E.3.2 Risk preferences

We use a survey-based measure of risk preferences based on Binswanger (1980). In the main measure presented in text, we ask participants about a hypothetical maize sale. We ask which of five hypothetical payments respondents would choose for this maize, if the payout was determined by a coin toss. In the first payout, they would be certain to be paid 300 ETB for one 50kg bag of maize. In the second, they would have an equal chance of receiving 200 ETB or 400 ETB. After that, there are three more payouts, which increase in both mean and variance, as shown in Table A.7. We treat this choice as a categorical variable, with values of 1 for those who made the

^{59.} We recode 47 observations over the three rounds who give inconsistent answers as missing. They prefer 100 ETB in the first two questions but choose less than 150 ETB in one month for the third question. We view them as misunderstanding the question.

most risk averse choice and 5 for those who chose the most risk neutral to risk loving option.⁶⁰

| (1) | (2) | (3) | (4) | (5) | (6) | (7) |
|--------|---------|-------|------------|-----------|------------------------|----------------------|
| Choice | Payouts | | Exp. value | Std. dev. | $\Delta E / \Delta SD$ | Risk aversion |
| | Heads | Tails | | | | |
| 1 | 2.5 | 2.5 | 2.5 | 0.00 | 0.35 | Severe |
| 2 | 2 | 4 | 3 | 1.41 | 0.35 | Intermediate |
| 3 | 1.5 | 5.5 | 3.5 | 2.83 | 0.35 | Moderate |
| 4 | 1 | 7 | 4 | 4.24 | 0.35 | Slight-to-neutral |
| 5 | 0 | 10 | 5 | 7.07 | | Neutral-to-preferred |

Table A.7: RISK AVERSION FOR EACH PAYOUT

Notes: Column 1 gives the choice number. Columns 2 and 3 give the payout options of the hypothetical lotteries. Columns 4 and 5 give the mean and variance of each lottery. The successive lotteries offered increase in both mean and variance, with payouts ordered from most to least risk averse.

E.3.3 Locus of Control, Perceptions of Poverty, and Grit

All measures take values from 1 ("Strongly Disagree") to 4 ("Strongly Agree"). Scores are the sum of items in the scale, standardised and reported as z scores. For all scales, we assess the reliability of each item. Items that met any of the following criteria were removed: low corrected item-total correlation (0.25); increased Cronbach's α if item removed; low item variation (80 per cent identical responses on the item). If respondents did not answer all items in a sub-scale, we code the items they do not answer as missing and adjust their score to generate a homogeneous score range using an appropriate multiplier. However, if a respondent is missing over 60 per cent of the items of a sub-scale or has given the same answer to all items on the scale, we replace the sub-scale score as missing.

Locus of control: We construct two sub-scales from a subset of items from the Internal Powerful Others Chance (IPC) scale (Levenson, 1981). Higher values on the Internal scale indicate that respondents see outcomes as contingent on individual behaviour. Higher values on the Chance scale indicate that respondents see outcomes as a result of chance, luck or fate (chance locus of control).

Causes of poverty: Similarly, we construct two sub-scales of the Attributions for Poverty scale (Feagin, 1975) which capture if individuals use Individualistic or Fatalistic explanations for poverty. Higher values on the Individualistic scale mean individuals attribute poverty to individual characteristics. Higher values on the Fatalistic scale mean individuals attribute poverty to chance or luck.

^{60.} We prefer this to estimating risk preference parameters assuming a specific functional form for the utility function, as this relies on all households making decisions under uncertainty in the same way.

Grit: We construct a standardised index of grit from two measures in the vein of Alan, Boneva, and Ertac (2019). The first asks level of agreement with the statement "I do a thorough job". The second asks level of agreement with the statement "I make plans and follow through with them".

E.3.4 Information

We explore if farmers take the same actions shown by subjects of the documentaries. We construct an inverse-covariance-weighted average following Anderson (2008), standardised relative to the within-village control group. This index comprises the following variables, each equal to 1 (and 0 otherwise) if the household:

- 1. earns any income from trading
- 2. attends community meetings to discuss agricultural issues
- 3. receives or seeks visits by an agricultural expert
- 4. takes advice on land preparation from agricultural expert
- 5. takes advice on seeds from agricultural expert
- 6. takes advice on fertilisers from agricultural expert
- 7. number of visits received by an agricultural extension worker (the only non-binary component of the index)
- 8. visited a demonstration plot in last year
- 9. grows cash crops
- 10. uses a water pump
- 11. builds stone bands and terracing
- 12. applies water conservation/water harvesting practices
- 13. applies crop rotation
- 14. uses any irrigation technique
- 15. uses cattle in crop activities

E.3.5 Expected fertiliser yields:

We elicit expectations about the increase in output from the use of modern (phosphatebased) fertilisers. We asked a list of questions to the household head to elicit how many kilograms of output they would expect to produce on an hectare of their land if 0, 50, 100, 150 kilograms of fertiliser were applied. We varied whether the hypothetical season was good or bad (in terms of agronomic conditions) and whether the crop produced was maize or sorghum. To combine these answers, we first estimate the elasticity of expected output relative to fertiliser by regressing the answers to these questions on the four quantities of fertiliser for each respondent, by crop and hypothetical season, to generate four expected yields per respondent (i.e. expected yield from an extra kilo of fertiliser in a good/bad season for sorghum/maize). Next, we combine these four expected yield estimates into a single index, following Anderson (2008).

E.4 Consumption and well-being

E.4.1 Consumption

All consumption variables are constructed in USD PPP and transformed into adult equivalent units, where adult equivalent household members are constructed using the OECD scale. The household head is 1, each other adult is weighted 0.7 and each child under 16 is weighted 0.5.

Food consumption is the sum of the value of food consumed from various sources over the past 7 days divided by 7 and multiplied by 30 to obtain a monthly estimate. This includes food purchased, received via barter, gifts, loans, wages in kind and self-production. Following Beegle et al. (2012), for purchased food items, we use reported prices, and for food received via barter, gifts, loans and wages in kind and self-produced food items, we construct prices using the first available level of price of purchased food from the following: household-level price, screening site level median, median from the neighbouring *kebele*, sample level median.

Non-food small-item consumption is the sum of frequent non-food consumption, with a recall period of one month. Items included are: toiletries, transportation costs, mobile phone costs, energy, cigarettes and tobacco, repair, tailor, barber, other services and other small purchases (less than 100 ETB, or \$11.5 PPP).

Non-food lumpy consumption is the sum of expenses made over the past 12 months (divided by 12 to obtain monthly estimates), from the following list of items: clothing and footwear, utensils, beddings, school expenses, health expenses, funerals, weddings, religious expenses, contribution to community projects, land taxes and other large purchases (more than 100 ETB, or \$11.5 PPP).

E.4.2 Subjective well-being

The subjective well-being is measured using two items indicating best and happiest life. Best life is measured by showing respondents a picture of a ladder with 10 steps (Cantril, 1966). They are told the top of the ladder represents the best possible life for them and the bottom step represents the worst possible. They are then asked, "Where on the ladder do you feel you personally stand at present?" The above question was repeated to measure happiest life, with the top and bottom of the ladder representing the happiest and least happy possible life.

E.5 Savings, credit and revenue

Savings and credit

We define an individual's savings as the total cash savings each respondent has in all possible savings places: banks, co-operatives, voluntary savings and loan groups, traditional *iqqub* (a version of a rotating credit and savings association), or with a friend or relative. Savings were captured individually for the household head and their spouse and are analysed at individual level, as men and women commonly save some amounts separately both within the household and in savings groups. Results are robust to aggregating all savings across the household and conducting analysis at household level.

To measure credit, we ask about the total value of loans larger than 15 ETB in nominal values (1.7 USD PPP), including from co-operatives, banks or micro-finance institutions, money-lenders, iqqub and friends or family. We consider only the value of the loan when it was first taken out.

Revenue

The revenue aggregate includes revenues received from wage labour, livestock and poultry, crops, non-farm activities and transfers and remittances.

Revenue from crops is computed as price times quantity sold in the last agricultural season. We capture production in both the short and long rains seasons. Quantities produced were replaced to missing if the yield per crop was higher than designated cut-off values.⁶¹ We use Central Statistical Agency (CSA) data from the Oromiya region for December 2015 and January 2016 to compute the price of crops. The process for computing prices is:

- 1. Of the 42 crops in our dataset, we were able to find exact matches for 25 crops: White teff, maize, zengada, horse beans, groundnuts, sesame, lentils, chat, bananas, potatoes, sugar cane, pineapple, avocado, onion, spinach, garlic, chickpeas, cow peas, orange, godere, sweet potato, tomato, haricot beans, fenugreek, green pepper
- 2. For six crops, there were more varieties in the CSA data than in our questionnaire, hence we take an average of the prices for the various varieties for each crop: sergegna teff, barley, wheat, sorghum, linseed and coffee.
- 3. For the following seven crops, we could not find exact matches or varieties in the CSA data, and used the best available approximation for these crops (listed in brackets): sinar/gerima (oats), adenguare (chick peas), red pepper (green pepper), wasira (average of wheat and barley), guaya/grass peas (white peas), finger millet (African millets) and tobacco (chat).
- 4. We could not find price observations in the Oromiya region for the following crops, and hence used the price data points from other regions: mango (Tigray region) and oats and sunflower seeds (Amhara region).

Revenue from livestock rearing and produce includes all revenues from sale of livestock and poultry and the production of livestock goods in the last 12 months. We use the self-reported sale prices to construct the livestock revenue variable. We include rental out of livestock, dairy, wool, egg sales.

Revenue from wage labour is the sum of agricultural and non-agricultural wages received by the household over the past 12 months. Non-farm revenue is the number of months an activity was operated for times the monthly revenue from this activity, for a list of common activities. Transfer revenues are revenues from all sources of transfers: pension, remittances, public works and other assistance programs, and other transfers e.g. burial associations, in the last 12 months. It is computed as the number of months the household received transfers from this source times the average monthly transfer from this source.

^{61.} Cut-off values are listed in brackets for each crop: wheat (2 tons/ha), bananas(2 tons/ha), white teff (2 tons/ha), horse beans (1.6 tons/ha), coffee (1.6 tons/ha), barley (1.8 tons/ha), sorghum (3.2 tons/ha), maize (4 tons/ha), sesame (1.5 tons/ha), lentils (1.5 tons/ha), chat (2.5 tons/ha), haricot beans (3.5 tons/ha), onions (10 tons/ha), sugar cane (37 tons/ha).

F Experimental integrity

F.1 Design, sample, and compliance





Notes: Diagram of the sampling and randomisation into different experimental groups. Rectangles indicated villages, whereas the circles indicate households. Numbers inside the circle represent the number of households in each experimental group per village.

| | All groups | Treatment | Placebo | Within-village control | Pure control |
|-----------------------------|------------|-----------|---------|---------------------------|--------------|
| Number of villages | 74 | | 64 | | 10 |
| Individuals: | | | | | |
| In sample | 2434 | 690 | 717 | 705 | 322 |
| Given tickets | 2112 | 690 | 717 | 705 | 0 |
| Compliers | 2070 | 673 | 698 | 699 | 0 |
| Non-compliers | 42 | 17 | 19 | 6 | 0 |
| of which | | | | | |
| At wrong screening | 20 | 3 | 11 | 6 | 0 |
| Missed screening | 22 | 14 | 8 | 0 | 0 |
| Proportion of non-compliers | .017 | .025 | .026 | .009 | 0 |
| Households: | | | | | |
| In sample | 1322 | 383 | 378 | 381 | 180 |
| Given tickets | 1142 | 383 | 378 | 381 | 0 |
| Compliers | 1116 | 371 | 368 | 377 | 0 |
| Non-compliers | 26 | 12 | 10 | 4 | 0 |
| of which | | | | | |
| At wrong screening | 11 | 2 | 5 | 4 | 0 |
| Missed screening | 15 | 10 | 5 | 0 | 0 |
| Proportion of non-compliers | .02 | .031 | .029 | .01 | 0 |

Table A.8: SAMPLE AND COMPLIANCE

Notes: Observations for individuals and households by treatment and compliance to treatment.

F.2 Balance

| | Baseline Sample Means | | | | Endline Sample Means | | | | |
|--|--------------------------|---------|-----------|---------|--------------------------|---------|-----------|---------|--|
| Variable | $\operatorname{Control}$ | Placebo | Treatment | p-value | $\operatorname{Control}$ | Placebo | Treatment | p-value | |
| Male | 0.491 | 0.484 | 0.493 | 0.49 | 0.495 | 0.477 | 0.490 | 0.14 | |
| Age | 36.6 | 37.1 | 37.0 | 0.86 | 36.7 | 37.0 | 37.0 | 0.89 | |
| Years of education | 1.23 | 1.28 | 1.49 | 0.13 | 1.25 | 1.28 | 1.50 | 0.15 | |
| Single | 0.074 | 0.068 | 0.093 | 0.24 | 0.070 | 0.065 | 0.095 | 0.13 | |
| Watches television at least once a week | 0.110 | 0.094 | 0.104 | 0.63 | 0.108 | 0.093 | 0.113 | 0.47 | |
| Listens to radio at least once a week | 0.627 | 0.588 | 0.620 | 0.37 | 0.623 | 0.581 | 0.627 | 0.25 | |
| Travels outside the district at least once a week | 0.142 | 0.121 | 0.148 | 0.38 | 0.144 | 0.127 | 0.159 | 0.34 | |
| Ever lived outside of current district six months | 0.104 | 0.078 | 0.110 | 0.12 | 0.106 | 0.080 | 0.116 | 0.11 | |
| Aspirations index | 0.000 | -0.018 | 0.044 | 0.55 | -0.001 | -0.018 | 0.070 | 0.31 | |
| Expectations index | 0.000 | -0.072 | 0.054 | 0.07 | -0.006 | -0.057 | 0.065 | 0.09 | |
| Daily minutes working | 350 | 351 | 355 | 0.85 | 354 | 351 | 356 | 0.88 | |
| Daily minutes in leisure | 739 | 729 | 735 | 0.63 | 739 | 727 | 734 | 0.53 | |
| Internal locus of control | 12.7 | 12.5 | 12.6 | 0.61 | 12.7 | 12.6 | 12.6 | 0.75 | |
| Individual causes of poverty | 8.91 | 8.83 | 8.93 | 0.76 | 8.88 | 8.86 | 8.96 | 0.74 | |
| Chance locus of control | 12.3 | 12.4 | 12.5 | 0.61 | 12.3 | 12.4 | 12.5 | 0.63 | |
| Fate causes of poverty | 6.71 | 6.81 | 6.84 | 0.60 | 6.69 | 6.79 | 6.83 | 0.56 | |
| Risk aversion: most to least risk averse $(1 \text{ to } 5)$ | 3.23 | 3.15 | 3.24 | 0.56 | 3.22 | 3.16 | 3.25 | 0.60 | |
| % that is patient | 0.333 | 0.308 | 0.341 | 0.45 | 0.336 | 0.309 | 0.341 | 0.46 | |
| % that is somewhat impatient | 0.119 | 0.146 | 0.120 | 0.31 | 0.114 | 0.146 | 0.120 | 0.23 | |
| % that is most impatient | 0.548 | 0.546 | 0.539 | 0.96 | 0.550 | 0.545 | 0.538 | 0.93 | |
| % that is present biased | 0.323 | 0.362 | 0.358 | 0.32 | 0.324 | 0.355 | 0.360 | 0.41 | |
| % that is patient now and impatient later | 0.229 | 0.213 | 0.230 | 0.73 | 0.231 | 0.220 | 0.235 | 0.83 | |
| Best life | 4.27 | 4.10 | 4.31 | 0.16 | 4.27 | 4.10 | 4.34 | 0.13 | |
| Happiest life | 6.64 | 6.51 | 6.61 | 0.68 | 6.68 | 6.50 | 6.65 | 0.48 | |
| Omnibus F-test | 0.62 | 0.23 | 0.71 | 0.51 | 0.70 | 0.19 | 0.57 | 0.42 | |

Table A.9: BALANCE TESTS – INDIVIDUAL-LEVEL VARIABLES

Table shows means of selected baseline variables for each treatment group in the baselined sample (columns 1-3) and endlined sample (columns 5-7). Column 4 shows *p*-values for equal means in the baselined sample, evaluating balanced treatment assignments. Column 8 shows *p*-values for equal means in the endlined sample, evaluating balanced attrition. Hypothesis tests are based on heteroskedasticity-robust standard errors clustered at the household-level. The bottom row reports *p*-values from an omnibus *F*-test regressing all covariates on a treatment group indicator. In the fourth and eight column, we report randomization-inference *p*-values from a chi-squared statistics test from a multinomial logit regression using treatment categories as the dependent variable, following Kerwin, Rostom, and Sterck (2024).

| | Baseline Sample Means | | | | Endline Sample Means | | | | |
|---|--------------------------|---------|-----------|---------|--------------------------|---------|-----------|---------|--|
| Variable | $\operatorname{Control}$ | Placebo | Treatment | p-value | $\operatorname{Control}$ | Placebo | Treatment | p-value | |
| Number of individuals aged 0-6 | 1.42 | 1.44 | 1.33 | 0.32 | 1.42 | 1.45 | 1.32 | 0.28 | |
| Number of individuals aged 11-15 | 0.610 | 0.692 | 0.774 | 0.03 | 0.618 | 0.710 | 0.782 | 0.03 | |
| Number of individuals aged 7-10 | 0.908 | 0.921 | 1.071 | 0.02 | 0.901 | 0.948 | 1.075 | 0.03 | |
| Household size | 5.53 | 5.58 | 5.73 | 0.43 | 5.52 | 5.66 | 5.73 | 0.43 | |
| Children aged 11-15 in school | 0.455 | 0.526 | 0.592 | 0.04 | 0.460 | 0.539 | 0.599 | 0.04 | |
| Daily minutes in school for children aged 11-15 | 148 | 180 | 193 | 0.03 | 150 | 184 | 196 | 0.03 | |
| Daily minutes studying for children aged 11-15 | 53.0 | 63.9 | 69.0 | 0.06 | 53.3 | 65.7 | 69.4 | 0.05 | |
| Children aged 7-10 in school | 0.469 | 0.397 | 0.516 | 0.05 | 0.462 | 0.403 | 0.514 | 0.08 | |
| Daily minutes in school for children aged 7-10 | 156 | 131 | 167 | 0.06 | 154 | 132 | 167 | 0.11 | |
| Daily minutes studying for children aged 7-10 | 38.3 | 29.7 | 41.5 | 0.02 | 37.3 | 29.5 | 41.4 | 0.03 | |
| Schooling expenditure (USD) PPP | 38.7 | 44.2 | 50.8 | 0.03 | 38.3 | 44.6 | 51.2 | 0.02 | |
| Value of livestock (USD) PPP | 1341 | 1220 | 1402 | 0.09 | 1346 | 1226 | 1432 | 0.06 | |
| Value of tools (USD) PPP | 70.8 | 72.7 | 76.2 | 0.65 | 71.3 | 71.5 | 77.3 | 0.55 | |
| Food security index: z-score | 0.044 | -0.061 | 0.017 | 0.31 | 0.051 | -0.051 | 0.036 | 0.32 | |
| Months of food insecurity | 5.59 | 5.73 | 5.66 | 0.80 | 5.61 | 5.77 | 5.67 | 0.74 | |
| Value of durable assets excluding tools (USD) PPP | 52.5 | 53.7 | 62.6 | 0.48 | 53.6 | 55.1 | 62.1 | 0.62 | |
| Value of house (USD) PPP | 1186 | 1077 | 1215 | 0.21 | 1185 | 1093 | 1218 | 0.31 | |
| Non-organic roof | 0.485 | 0.486 | 0.570 | 0.03 | 0.487 | 0.503 | 0.566 | 0.09 | |
| Own toilet | 0.760 | 0.765 | 0.741 | 0.73 | 0.765 | 0.764 | 0.740 | 0.69 | |
| Omnibus F-test | 0.284 | 0.478 | 0.310 | 0.244 | 0.274 | 0.414 | 0.206 | 0.186 | |
| Omnibus F-test (across both Table A.9 and A.10) | 0.442 | 0.352 | 0.578 | 0.378 | 0.396 | 0.108 | 0.458 | 0.218 | |

Table A.10: BALANCE TESTS – HOUSEHOLD-LEVEL VARIABLES

Table shows means of selected baseline variables for each treatment group in the baselined sample (columns 1-3) and endlined sample (columns 5-7). Column 4 shows p-values for equal means in the baselined sample, evaluating balanced treatment assignments. Column 8 shows p-values for equal means in the endlined sample, evaluating balanced attrition. Hypothesis tests are based on heteroskedasticity-robust standard errors clustered at the household-level. The second last row show p-values from an F-test of all covariates in this table regressed on the control, placebo, or treatment indicators. In the fourth and eight column, we report a nested test from a multinomial logit regression using treatment categories as the dependent variable. The bottom row of p-values comes from household-level regressions of all covariates on the respective indicators or multinomial logits, spanning all covariates in Table A.9 and A.10, following Kerwin, Rostom, and Sterck (2024).
F.3 Attrition

| | (1) | (2) | (3) | (4) | (5) | (6) |
|--|---------|-------------|------------|------------|-------------|-------------|
| Attrited in | | | | | | |
| | Any | Any | After | After | After | After |
| | round | round | six months | six months | five years | five years |
| Treatment | 0.021 | 0.020 | 0.000 | 0.100 | 0.026^{*} | 0.021 |
| | (0.015) | (0.116) | (0.006) | (0.086) | (0.014) | (0.109) |
| Placebo | 0.018 | 0.057 | -0.003 | 0.018 | 0.021 | 0.042 |
| | (0.015) | (0.121) | (0.006) | (0.044) | (0.014) | (0.115) |
| % male | | -0.045 | | 0.018 | | -0.059 |
| | | (0.046) | | (0.028) | | (0.038) |
| Age | | -0.000 | | -0.000 | | 0.000 |
| | | (0.001) | | (0.000) | | (0.001) |
| Years of education | | 0.004 | | -0.002 | | 0.005 |
| | | (0.005) | | (0.002) | | (0.005) |
| Single | | 0.068 | | 0.054 | | 0.013 |
| | | (0.049) | | (0.038) | | (0.033) |
| % that watches television at least once a week | | -0.010 | | -0.015 | | 0.006 |
| | | (0.030) | | (0.014) | | (0.028) |
| % that listens to radio at least once a week | | -0.022 | | -0.019 | | -0.004 |
| | | (0.026) | | (0.013) | | (0.022) |
| % that travels outside the district at least once a week | | -0.007 | | 0.026 | | -0.032 |
| | | (0.031) | | (0.026) | | (0.020) |
| % that ever lived outside of current district six months | | 0.034 | | 0.044 | | -0.015 |
| | | (0.039) | | (0.033) | | (0.022) |
| Number of individuals aged 0-6 | | 0.009 | | 0.007 | | 0.002 |
| | | (0.010) | | (0.007) | | (0.008) |
| Number of individuals aged 11-15 | | -0.008 | | 0.010 | | -0.018 |
| | | (0.013) | | (0.006) | | (0.011) |
| Number of individuals aged 7-10 | | 0.021^{*} | | 0.004 | | 0.018^{*} |
| | | (0.012) | | (0.007) | | (0.010) |
| Total assets (USD) PPP | | -0.000 | | -0.000 | | -0.000 |
| | | (0.000) | | (0.000) | | (0.000) |
| % male*treatment | | 0.028 | | -0.063 | | 0.032 |
| | | (0.073) | | (0.051) | | (0.069) |
| % male*placebo | | -0.025 | | -0.012 | | -0.017 |
| | | (0.087) | | (0.032) | | (0.082) |
| Age*treatment | | 0.000 | | -0.000 | | 0.000 |
| | | (0.002) | | (0.001) | | (0.002) |
| Age*placebo | | 0.000 | | 0.000 | | -0.000 |
| | | (0.001) | | (0.000) | | (0.001) |
| Years of education*treatment | | 0.002 | | 0.001 | | 0.001 |
| | | (0.008) | | (0.002) | | (0.007) |
| Years of education*placebo | | -0.003 | | 0.001 | | -0.003 |
| | | (0.007) | | (0.002) | | (0.006) |
| Single*treatment | | -0.069 | | -0.072 | | -0.017 |
| | | (0.072) | | (0.045) | | (0.063) |
| Single*placebo | | -0.006 | | -0.058 | | 0.051 |
| | | (0.088) | | (0.040) | | (0.080) |

| Table A 11. | Determinants | OF | HOUSEHOLD-LEVEL | ATTRITION |
|-------------|--------------|------------------------|-----------------|-----------|
| 14010 1.11. | DETERMINANTS | $\mathbf{O}\mathbf{r}$ | HOUSEHOLD-LEVEL | ALIMITON |

Continues on next page.

| | | | | | | · · · |
|--|-------|----------|------------|--|------------|------------|
| | (1) | (2) | (3) | (4) | (5) | (6) |
| Attrited in | | | | | | |
| | Any | Any | After | After | After | After |
| | round | round | six months | six months $% \left({{{\left({{{\left({{{\left({{{\left({{{\left({{{{}}}}} \right)}} \right.}$ | five years | five years |
| Continued from previous page | | | | | | |
| % that watches television at least once a week*treatment | | -0.009 | | 0.002 | | -0.015 |
| ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, | | (0.044) | | (0.015) | | (0.042) |
| % that watches television at least once a week*placebo | | 0.030 | | 0.016 | | 0.014 |
| 1 | | (0.057) | | (0.016) | | (0.055) |
| % that listens to radio at least once a week*treatment | | 0.022 | | 0.009 | | 0.001 |
| | | (0.036) | | (0.015) | | (0.034) |
| % that listens to radio at least once a week *placebo | | 0.046 | | 0.012 | | 0.032 |
| | | (0.035) | | (0.015) | | (0.031) |
| % that travels outside the district at least once a week*treatment | | -0.040 | | -0.030 | | -0.019 |
| | | (0.036) | | (0.026) | | (0.028) |
| % that travels outside the district at least once a week*placebo | | -0.035 | | -0.030 | | -0.004 |
| | | (0.041) | | (0.027) | | (0.032) |
| % that ever lived outside of current district six months*treatment | | -0.073 | | -0.035 | | -0.020 |
| | | (0.047) | | (0.038) | | (0.035) |
| % that ever lived outside of current district six months *placebo | | -0.024 | | -0.049 | | 0.028 |
| | | (0.057) | | (0.032) | | (0.048) |
| Number of individuals aged 0-6*treatment | | -0.001 | | -0.007 | | 0.002 |
| | | (0.015) | | (0.008) | | (0.013) |
| Number of individuals aged 0-6*placebo | | -0.009 | | -0.008 | | -0.001 |
| | | (0.014) | | (0.007) | | (0.013) |
| Number of individuals aged 11-15 [*] treatment | | 0.002 | | -0.007 | | 0.014 |
| | | (0.022) | | (0.009) | | (0.021) |
| Number of individuals aged 11-15*placebo | | -0.009 | | -0.010 | | 0.000 |
| | | (0.020) | | (0.008) | | (0.019) |
| Number of individuals aged 7-10*treatment | | -0.020 | | -0.003 | | -0.017 |
| | | (0.017) | | (0.007) | | (0.015) |
| Number of individuals aged 7-10*placebo | | -0.043** | | -0.004 | | -0.039** |
| | | (0.018) | | (0.007) | | (0.017) |
| Total assets (USD) PPP*treatment | | -0.000 | | -0.000 | | -0.000 |
| | | (0.000) | | (0.000) | | (0.000) |
| Iotal assets (USD) PPP*placebo | | (0.000) | | -0.000 | | 0.000 |
| | | (0.000) | | (0.000) | | (0.000) |
| Control mean | .03 | .03 | .01 | .01 | .03 | .03 |
| F-test p -value | .28 | .34 | .87 | 1 | .11 | .36 |

Notes: Each column reports coefficients from regressions where the outcome is an indicator for individual attrition across different survey rounds. Columns labeled "Any round" indicate attrition at any point during the study period; "After six months" and "After five years" indicate attrition by those specific time-points. If a baseline covariate is missing, we replace the missing values with the sample mean and include a missing data indicator. All regressions include village fixed effects. Standard errors are robust to heteroskedasticity and are reported in parentheses. *; **; and *** denote significance at the 10; 5; and 1 percent levels respectively. At the bottom we report the mean attrition rate in the control group and a *p*-value from an *F*-test testing that all coefficients on the covariates reported in the column are equal to zero. The number of observations is 1,152 households interviewed at baseline.

G Deviations from the Pre-Analysis Plan (PAP)

This study was pre-registered on the AEA RCT Registry (ID: AEARCTR-0001483) under the title "The Future in Mind: Aspirations and Forward-Looking Behaviour in the Short and Long Run in Rural Ethiopia." Pre-registration occurred on February 15, 2017. Pre-registration was only for the analysis of the long-run data.

We conducted the treatment and six-month follow-up in 2010-11 (see timeline in Figure A.2). Analysis in the working paper reporting on the short run data (Bernard et al., 2014) was not pre-specified, as pre-analysis plans were not yet common. The AER registry was set up in 2013 and papers like Olken (2015) or Casey, Glennerster, and Miguel (2012) were not yet published and only just circulating. So the intervention's effects on aspirations and outcomes measured in the short run were already known and publicly available when prespecifying analysis in the long run. We then conducted data collection for the five-year follow-up data (from December 2015 to January 2016). Pre-registration occurred after data collection but before analysis of the five-year follow-up data.

We have made some changes in analysis relative to the pre-analysis plan. We show all results as pre-specified, either in this appendix or Appendix H or J, and changes do not materially affect our results.

First, the main tables now report ANCOVA results for most outcome variables, those for which we collected baseline values. The pre-specified specification omitted baseline outcomes as controls. We report robustness tests for this change in Appendix H. As discussed in the main text, there are only four coefficients where the ANCOVA and PAP specifications differ: the treatment vs placebo effect (i) on the value of livestock and (ii) on the number of children aged 16-20 in school, and the treatment vs control effect on (iii) time spent in school for children aged 7-15 and (iv) on having an organic roof. These all lose significance in the ANCOVA specification relative to the pre-specified specification, where they are marginally significant.

Second, we now apply the same specification for the six-month and five-year follow-up analysis, an ANCOVA specification comparing treatment, placebo and within-village control groups. We had pre-specified different specifications for the two follow-ups. For the six-month follow-up, we pre-specified a specification comparing treatment, placebo and within-village control groups, although without controlling for the baseline value of the outcome. This is shown in Appendix Section H, as discussed above. For the five-year follow-up, we specified the regression we run to test for spillovers (Equation (4) in Appendix J) as the main specification. This compares treatment, placebo and within-village control groups to the pure control group. We have reduced the emphasis on tests for spillovers in the main paper so do not use this as our main specification. The results with this alternative control group are shown in Appendix Section J: the main treatment effects and (lack of) placebo effects are robust regardless of which control group is used.⁶²

^{62.} We note that the PAP inaccurately stated that the pure control villages were randomly selected. In fact, these villages were selected based on logistics, as described in Section 5.5.

Third, the PAP pre-specified trimming observations four standard deviations above the sample mean for continuous outcome variables, a non-standard approach used in the short run working paper (Bernard et al., 2014). We revised this to trim values above the 99th percentile. We do not show results for the pre-specified approach, as it is non-standard.

The fourth and fifth changes are outlined in detail in the sections below. Fourth, we made some minor adjustments to the pre-specified construction of aspirations, expectations, education, risk preference and time preference variables, none of which change results. We also added six variables and dropped five variables, outlined and justified below in Section G.1. We note if variables are not pre-specified in footnotes in the main text.

Fifth, for multiple hypothesis test correction, we group outcomes using a slightly different (and hopefully clearer) organising framework, but without excluding pre-specified outcome variables (beyond a few specific cases explained below).

At the end of this section, we report results as laid out in the preanalysis plan, showing that the deviations do not affect our substantive conclusions. Importantly, coefficients and sample sizes are very slightly different between the PAP results and those in the paper. The PAP results use the pre-specified non-ANCOVA specification, so observations which do not have baseline values of the outcome are not dropped, while the paper results use an ANCOVA specification. In a few cases variable construction differs.

G.1 Changes to pre-specified outcome variables

Education

- We had specified analysing effects on children aged between 6 and 15 at endline (Cohort 2). We instead report effects on children aged between 7 and 15 at endline as children can only start school at 7. There are no effects on the number of children enrolled whether we consider children aged 6-15 (Table A.13, constructed as in the PAP, at the end of this appendix) or 7-15 (Table 3, paper).
- We added a variable for whether children aged between 16 and 20 at endline (Cohort 1) had completed eighth grade.
- We omitted a measure of absenteeism from school due to data quality issues. Investment
- We added an analysis of extensive margin investment in inputs (in line with pre-specified extensive margin analysis for savings and credit).
- We dropped land area rented or sharecropped as there was little variation. Land is not privately owned and at the time of the study, there was still ambiguity in this area about whether land rental was legal (Deininger, Ali, and Alemu, 2008). *Welfare*
- We added variables capturing the value of the house and indicators for having a non-organic roof and own toilet (which were directly observed by enumerators). *Aspirations and expectations*

- We do not report a pre-specified dimension of aspirations and expectations, related to social status, and exclude it from the aspirations and expectations indices. Bernard and Taffesse (2014) find challenges with internal reliability of this measure during their validation study. They found a correlation coefficient of only 8 per cent when the same households were twice asked questions on aspirations on social status within a two-week period, considerably worse than for aspirations and expectations for income, wealth or education, potentially indicating a lack of understanding of the concept.⁶³
- In Figure II and elsewhere in the paper, we construct the aspirations and expectations indices using the same methodology as the other summary indices, following Anderson (2008), and not using subjective weights for the importance of each dimension to respondents, as we pre-specified, to enable comparison across indices of the magnitude of effects.
- Appendix Table A.12 shows effects on the aspirations and expectations indices, constructed as per the PAP. Effects are positive and statistically significant regardless of index construction. Effects with pre-specified variable construction are slightly smaller than effects in Table A.1 in the paper.

Alternative mechanisms

- We improved the construction of risk and time preferences measures in line with current practice in the behavioural literature.
 - For risk preferences, we ask participants which of five hypothetical payments they would choose for a hypothetical maize sale, if the payout was determined by a coin toss. The first payout is certain, then payouts increase in both mean and variance. In the paper, we report risk preferences as a categorical variable from 1 to 5, where 5 is the least risk averse. We pre-specified an indicator equal to one if the respondent answered 5, which loses some information.
 - We dropped a second measure, a gamble, where individuals bet on the outcome when someone flips a coin. Results are similar using this measure or the maize-related one.
 - For time preferences, we used a simple questionnaire from Ashraf, Karlan, and Yin (2006) for populations with limited numeracy, as described in Section E.3.1. In the paper, we construct the measure as in that paper. We pre-specified an alternative construction of the measure which makes stronger assumptions⁶⁴ and also does not enable us to examine present bias.

^{63.} Beaman et al. (2012) also dropped a dimension with lower internal reliability from their index of measures of aspirations.

^{64.} The pre-specified measure was to construct the subjective discount factor $\beta = 1/(1+\delta)$, where δ is the rate of time preference. We ask if respondents would prefer receiving 100 ETB now or 125 ETB in one month. To those who chose 125 ETB, one ETB in one month is worth between 0.8 and 1 ETB today: they have a monthly discount actor between 1 and 0.8. The measure assigns them the mid-point of 0.9. If they chose 100 ETB, they then choose between 100 ETB now or 150 ETB in one month. If they choose 150 ETB, they have a monthly discount factor between 0.8 and 0.667 and are assigned the midpoint of 0.733, and so on for later choices. This measurement assumes a linear utility function, and will estimate a discount rate which is biased upwards (and a discount factor which is downward biased) if the function is concave.

- The results do not change regardless of which method of constructing the variables we use: results in Table 6 in the paper and Appendix Table A.15 using the PAP definitions are very similar, with no effects of the documentaries on these preferences.
- We dropped two pre-specified indices which combined, first, the internal locus of control plus individual causes of poverty index, and, second, the chance locus of control plus fate causes of poverty index. There were only two variables in each index and we felt they did not add information. Table A.12 shows there are no effects on either index.
- We added an adapted measure of grit, perseverance and passion for long-term goals (Duckworth et al., 2007; Alan, Boneva, and Ertac, 2019), which has become prominent in the economics literature since the PAP.
- We simplified the construction of the information index capturing if households mimicked activities shown in the videos. The PAP contained errors in describing the information index. (i) Four variables (grain milling, using a pump, an extension advice sub-index, using irrigation) were listed twice, and one variable (growing maize, teff, mango, sugarcane, coffee) was almost exactly the same as another one (growing cash crops). (ii) We pre-specified including variables related to whether households took up extension advice in a sub-index, but then specified including this index within the information index alongside other dummy variables. Instead, we construct the index from all pre-specified dummy variables reflecting if households undertook any behaviours shown in the videos.

G.2 Multiple inference adjustment

Current approach: We adjust naive p-values as follows. We group related variables within table panels. A table panel corresponds to a concept in our conceptual framework, such as labour effort, agricultural investment, productive assets, non-productive durables and housing assets, or consumption. We view the group of variables within a panel as testing the same underlying hypothesis. To correct for multiple testing, we use the Benjamini, Krieger, and Yekutieli (2006) resampling procedure to calculate sharpened q-values which correct p-values for multiple tests across outcomes within each panel.

To summarise impacts, we report impact estimates on standardised inverse-covarianceweighted indices (Anderson, 2008) constructed from all outcomes reported in a table in our main exhibits. This results in five main indices, capturing agricultural investment (Table 3), educational investment (Table 4), welfare (Table 5), aspirations, and expectations (Figure II), plus an omnibus index. Following Kling, Liebman, and Katz (2007), we also aggregate the standardised indices into a single omnibus index (Table 7).

Pre-specified approach: The PAP pre-specified the omnibus index and five hypotheses, three primary and two secondary. These largely map onto the five indices tested in the paper. The primary hypotheses and variable groups were:

(H1) The intervention will increase self-beliefs (group 1: aspirations; group 2: expectations; group 3: locus of control; group 4: beliefs about causes of poverty);

- (H2) The intervention will increase labour supply and human capital investments (group 5: labour supply; group 6: educational investment);
- (H3) The intervention will increase forward-looking economic behaviour (group 7: savings and credit behaviours; group 8: investment in inputs).

The secondary hypotheses and groups were:

- (H4) The intervention will not affect other psychological channels (group 9: risk aversion, discount rates, information transmission by adopting activities in videos);
- (H5) The intervention will affect household welfare (group 10: consumption; group 11: food security; group 12: asset stock; group 13: subjective well-being; group 14: revenue).

Deviations from PAP: First, we use the approach of creating summary indices plus an omnibus index to summarise results on multiple similar variables, which reflects current practice in development economics. The PAP had proposed the omnibus index, but proposed selection of one focal outcome per variable group as a summary outcome, a method which is not widely used.

Second, we slightly alter which variables are grouped together in tables and hence indices. We realised groupings could be better mapped to standard concepts in psychology and economics:

- On further reading of the psychology literature (Bandura, 2001), locus of control and beliefs about causes of poverty (groups 3 and 4) are different psycho-social concepts than aspirations and its foundations. So we discuss them as alternative psychological channels i.e. they are now in Hypothesis 4, where we pre-specified them as being in Hypothesis 1.
- Labour supply, spending on inputs, and use of land are all inputs into productive activities so should be grouped together, i.e. labour supply (group 5) is now part of Hypothesis 3, where we had pre-specified it as part of Hypothesis 2.
- Productive assets (livestock and productive durables, i.e. tools, part of group 12) are an input into production and should be grouped with other inputs in Hypothesis 3. We had pre-specified all assets as part of Hypothesis 5.
- Savings and credit behaviours (group 7) are not an input into productive activity in the same way as other inputs in Hypothesis 3. Most variables we specified were about financial behaviours rather than flows. So we keep these variables as a separate group.
- Revenue (group 14) is not commonly used as an indicator of living standards in the same way as consumption, food security, assets or wellbeing. So we keep these variables as a separate group.

But other than these changes, the main exhibits map fairly closely onto these pre-specified hypotheses. Figure II on aspirations and expectations shows most variables in Hypothesis 1; Table 4 shows education variables for Hypothesis 2; Table 3 captures variables on investment in inputs from Hypothesis 3 plus labour supply and productive assets; Table 5 on welfare shows variables related to Hypothesis 5; and Table 6 on alternative mechanisms shows variables related to Hypothesis 4. Savings and credit behaviours (group 7) are shown in Appendix Table A.3 and revenue (group 14) is shown in Appendix Table A.5), with results described in text.

| | (1) | (2) | (3) | (4) |
|--|---------------------|--------------------|-----------------------|------------------------------------|
| | Af | ter five yea | urs | |
| | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. |
| PAP group 1 | | | | |
| Aspirations index | 0.07^{*} | -0.01 | 0.08^{**} | 0.19 |
| Focal outcome | (0.04) | (0.04) | (0.04) | (0.67) |
| Reported with different index construction | [0.18] | [0.85] | [0.11] | 1909 |
| Income (USD) PPP | 2436.44 | -689.07 | 3125.51° | 15446.71 |
| Reported in Table A.1 | (1745.50) [0.27] | (1604.00) | (1697.89) | (27746.43) |
| Wealth (USD) PPP | 1550.68 | [0.85] 836 95 | 71373 | 11978 72 |
| Reported in Table A.1 | (1298.04) | (1338.75) | (1392.27) | (21518.47) |
| | [0.29] | [0.83] | [0.61] | 1889 |
| Social Status (% of individuals) | -1.21 | 0.62 | -1.83 | 80.13 |
| Not reported elsewhere | (1.41) | (1.34) | (1.38) | (22.92) |
| | [0.39] | [0.83] | [0.23] | 1907 |
| Education (years) | 0.31^{*} | -0.13 | 0.43^{***} | 14.26 |
| Reported in Table A.1 | (0.16) | (0.17) | (0.16) | (2.60) |
| DAD 0 | [0.18] | [0.83] | $[0.04]^{**}$ | 1803 |
| PAP group z | | | | |
| Expectations index | 0.13^{***} | 0.01 | 0.11^{***} | 0.06 |
| Focal outcome | (0.04) | (0.04) | (0.04) | (0.67) |
| Reported with different index construction | 206 52 | [0.70] | 100.20 | 1909 |
| Reported in Table A 1 | (188.01) | 107.25 (177.52) | (102.24) | 0409.97 (2820.02) |
| neporica in Table 11.1 | [0.13] | [0.68] | [0.30] | 1894 |
| Wealth (USD) PPP | 612.21** | 256.15 | 356.06 | 4009.69 |
| Reported in Table A.1 | (249.03) | (247.36) | (250.75) | (3997.53) |
| • | [0.02]** | [0.50] | [0.19] | 1888 |
| Social Status (% of individuals) | -0.73 | 1.96 | -2.69* | 65.16 |
| Not reported elsewhere | (1.62) | (1.62) | (1.61) | (25.83) |
| | [0.65] | [0.50] | [0.16] | 1907 |
| Education (years) | 0.67^{***} | -0.35 | 1.02*** | 12.31 |
| Reported in Table A.1 | (0.26) | (0.27) | (0.27) | (3.88) |
| PAP aroum 2 | $[0.02]^{**}$ | [0.50] | [0.00] | 1803 |
| | 0.02 | 0.00 | 0.01 | |
| Individual agency index | (0.02) | (0.03) | -0.01 | 0.00 |
| Focal outcome | (0.06) | (0.06) | (0.06) | (1.00) |
| Internal locus of control | 0.02 | 0.08 | -0.05 | 1909 |
| Reported in Table 6 | (0.11) | (0.11) | (0.11) | (1.90) |
| | [0.84] | [0.89] | [0.90] | 1909 |
| Individual causes of poverty | 0.04 | 0.02 | 0.02 | 9.14 |
| Reported in Table 6 | (0.13) | (0.13) | (0.13) | (2.03) |
| | [0.84] | [0.89] | [0.90] | 1909 |
| PAP group 4 | | | | |
| Belief in chance index | -0.00 | 0.01 | -0.01 | -0.01 |
| Focal outcome | (0.06) | (0.06) | (0.06) | (1.00) |
| Not reported elsewhere | [1.00] | [0.93] | [0.92] | 1909 |
| Chance locus of control | -0.03 | -0.06 | 0.03 | 12.65 |
| Reported in Table 6 | (0.15) | (0.14) | (0.15) | (2.32) |
| Fate causes of powerty | [1.00] 0.01 | [U.93] 0.05 | [0.92] _0.04 | 1902 6.64 |
| Reported in Table 6 | (0.01) | (0.00) | -0.04 (0.10) | (1.71) |
| 1000000 | [1.00] | [0.93] | [0.92] | 1894 |

Table A.12: PRE-SPECIFIED GROUPS 1,2,3,4

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences between parameters reported in the first two columns. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and individual: age, years of education, an indicator for being single, and an indicator for being male. Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets and are calculated over each group (panel) of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in Appendix G.1, or in the main tables where they are reported. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. Heteroskedasticity-robust standard errors are in parentheses. Heteroskedasticity-robust standard errors are in parentheses, clustered at the household-level. The unit of observation is the individual respondent (household head or their spouse). The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 99th percentile, though the indices aggregate all non-missing outcomes. Aspirations index and Expectations index calculated as a standardised weighted average, with each dimensions weighted according to respondents' subjective weight for that dimensions.

| | (1) | (2) | (3) | (4) |
|---|---------------------|------------------|-----------------------|------------------------------------|
| | Af | ter five yea | ars | |
| PAP group 5 (household-level) | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. |
| Children aged 6.15 in school | 0.11 | 0.02 | 0.13 | 1.25 |
| Focal outcome | (0.00) | (0.02) | (0.13) | (1.20) |
| Reported in Table / for children 7 15 | (0.03) [0.21] | [0.00] | [0.00] | 1086 |
| Schooling expenditure (USD) PPP | 0.21 | 1.83 | 752** | 1030 |
| Reported in Table / | (2.04) | (2.54) | (3.11) | (32.73) |
| neporteu in 1401e 4 | (2.34) [0.00]*** | (2.04) [0.83] | [0.03]** | (02.10) |
| PAP group 5: secondary outcomes (household-level) | [0.00] | [0.00] | [0.05] | 1001 |
| Children aged 16-20 in school | 0.07** | 0.01 | 0.07* | 0.17 |
| Reported in Table 4 | (0.03) | (0.03) | (0.04) | (0.41) |
| 1 1 | $[0.05]^{*}$ | [0.87] | $[0.10]^{*}$ | 1086 |
| Daily minutes in school for children aged 6-15 | 57.38* | -15.52 | 72.90** | 550.51 |
| Reported in Table 4 for children 7-15 | (32.86) | (32.25) | (32.65) | (452.69) |
| 1 1 0 | [0.08]* | [0.87] | [0.06]* | 1077 |
| Daily minutes in school for children aged 16-20 | 36.62*** | 3.83 | 32.79** | 58.64 |
| Reported in Table 4 | (13.44) | (11.45) | (13.72) | (149.88) |
| * · · | [0.03]** | [0.87] | [0.06]* | 1085 |
| Daily minutes studying for children aged 6-15 | 22.13** | 8.73 | 13.39 | 93.12 |
| Reported in Table 4 for children 7-15 | (9.09) | (8.95) | (9.21) | (117.61) |
| | [0.04]** | [0.87] | [0.15] | 1076 |
| Daily minutes studying for children aged 16-20 | 9.60** | 1.63 | 7.96 | 17.82 |
| Reported in Table 4 | (4.67) | (4.24) | (4.99) | (52.12) |
| * · · | $[0.05]^{*}$ | [0.87] | [0.14] | 1078 |
| PAP group 6 (household head or spouse) | | | | |
| Daily minutes working | 28.30*** | 8.35 | 19.95*** | 326.16 |
| Focal outcome | (7.67) | (7.77) | (7.69) | (200.01) |
| Reported in Table 3 | $[0.00]^{***}$ | [0.28] | $[0.02]^{**}$ | 1887 |
| Daily minutes in leisure | -51.94*** | -31.72*** | -20.22** | 830.59 |
| Focal outcome | (9.61) | (10.05) | (9.91) | (187.38) |
| Reported in Table 3 | $[0.00]^{***}$ | [0.00]*** | $[0.04]^{**}$ | 1899 |

Table A.13: PRE-SPECIFIED GROUP 5,6

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences between Columns 1 and 2. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and characteristics of the household head: age, years of education, an indicator for being single, and an indicator for being male. The controls for family 2.2 correspond to the individual respondent (household head or spouse). Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets and are calculated over each group (panel) of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in detail in Appendix G.1, or in the main tables where they are reported. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. Heteroskedasticity-robust standard errors are in parentheses. The unit of observation is the household for group 5 and the individual respondent (household head or their spouse) for group 6. The number of observations varies slightly across rows because some respondents do not answer all questions. Ages of children correspond to those recorded at endline. Children aged 6–15 would have been 1-10 at the time of the intervention. Children aged 16–20 would have been 11–15 at the time of the intervention.

| $ \begin{array}{c c c c c c c c c c c c c c c c c c c $ | | | | | |
|---|--|--------------------|--------------------|------------------|----------------------|
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$ | | (1) | (2) | (3) | (4) |
| $ \begin{array}{ c c c c c c c c c c c c c c c c c c c$ | | Af | ter five ye | ars | |
| Preserve Treatment Placebo Test of total obs. PAP group 7 0.00 1.70 -1.51 17.87 Total swings (USD) PPP 0.20 1.70 -1.51 17.87 Reported in Table A.3 0.066 10.75 10.89 1890 Isa any swings -0.01 -0.01 -0.00 0.21 Reported in Table A.3 0.022 (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.04) (0.03) (0.03) (0.03) (0.03) (0.03) (0.03) (0.04) (0.07) (0.03) (0.03) (0.04) (0.07) (0.03) (0.03) (0.04) (0.07) (0.03) (0.03) (0.04) (0.07) (0.05) (0.03) (0.03) (0.03) (0.03) (0.03) (0.04) (0.04) (0.04) (0.05) (0.03) (0.05) (0.05) (0.05) (0.05) (0.05) (0.05) (0.05) (0.03) (0 | | | | | - Control mean |
| PAP group 7 Total obs. Total savings (USD) PPP 0.20 1.70 -1.51 17.87 Food outcome (3.86) (6.477) (0.86) (0.75) (0.89) 1890 Has any savings -0.01 -0.01 -0.00 0.21 (0.02) (0.03) (0.75) [0.95] 1000 Has outside savings -0.01 -0.01 -0.01 -0.01 -0.01 0.00 0.29 Reported in Table A.3 (0.02) (0.02) (0.01) (0.0 | | Treatment | Placebo | Treat. vs. | (SD) |
| PAP group 7 Total savings (USD) PPP 0.20 1.70 -1.51 17.87 Fecod outcome (3.86) (3.75) (3.86) (61.07) Reported in Table A.3 [0.06] [0.17] [0.89] 1890 Has any savings -0.01 -0.01 -0.00 0.21 Reported in Table A.3 [0.02] (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.03) (0.03) (0.03) (0.03) (0.03) (0.03) (0.03) (0.03) (0.04) Has any scripting in Table A.3 (0.01) (0.01) (0.01) (0.01) (0.01) (0.01) (0.01) (0.07) (0.33) 1009 Has any scripting in Table A.3 (0.03) (0.03) (0.03) (0.03) (0.03) (0.03) (0.03) (0.04) (0.07) (0.07) (0.07) (0.07) (0.01) (0.01) (0.01) (0.01) (0.01) (0.01) | | | | placebo | Total obs. |
| Total savings (USD) PPP 0.20 1.70 -1.51 (F407) Reported in Table A.3 [0.66] [0.75] [0.89] 1880 Has any savings -0.01 -0.01 -0.00 0.21 Reported in Table A.3 [0.03] [0.75] [0.98] 1890 Has outside savings -0.01 -0.01 0.00 0.22 Reported in Table A.3 [0.03] [0.75] [0.98] 1909 Has outside savings -0.01 -0.01 0.00 0.20 Reported in Table A.3 [0.03] [0.77] [0.93] [1009] Has any agricultural credit -0.03 0.03 [0.03] [0.71] [0.33] [0.71] [0.33] [0.71] [0.03] [0.77] [0.06]* 0.33 Reported in Table A.3 [0.01] [0.01] [0.01] [0.01] [0.01] [0.01] [0.01] [0.01] [0.07] [0.03] [1.77] [1.78,7] [1.31,8] [1.74,8] [1.79,9] [1.75,7] [1.31,8] [1.79,9] [1.75] | PAP group 7 | | | | |
| Faced nutcome (3.86) (3.75) (0.87) (0.88) (4.07) Reported in Table A.3 (0.02) (0.02) (0.02) (0.02) (0.02) Reported in Table A.3 (0.02) (0.02) (0.02) (0.02) (0.02) Has outside savings -0.01 -0.01 -0.01 0.01 (0.02) (0.02) (0.04) Has any credit (0.02) (0.02) (0.02) (0.02) (0.04) (0.03) (0.047) Has any credit (0.03) (0.03) (0.03) (0.047) (0.03) (0.047) Has any agricultural credit (0.02) (0.01) (0.01) (0.01) (0.01) (0.01) (0.07) Hypothetical loan (1 your USD) PPP -8.39* -9.33 -8.06 39.65 Reported in Table A.3 (0.03) (0.37) (135.57) (134.55) (197.46) 19.95 Hypothetical loan (1 your USD) PPP -8.27 -27.8 -27.62 -21.84 157.91.91 Reported in Table A.3 (137.97) (134.55) (197.46) 19.95 19.92 19.92 19.92 19.92 | Total savings (USD) PPP | 0.20 | 1.70 | -1.51 | 17.87 |
| Iteported in Table A.3[0.99][0.73][0.89][0.89][0.89]Has any swings -0.01 -0.01 -0.01 0.01 0.02 (0.41) Reported in Table A.3 (0.02) (0.02) (0.02) (0.02) (0.02) Has outside savings -0.01 -0.01 0.00 0.20 Reported in Table A.3 (0.02) (0.02) (0.02) (0.02) Has any credit -0.03 0.03 0.03 (0.03) (0.03) Has any agricultural credit 0.03 0.03 (0.03) (0.03) (0.03) Has any agricultural credit 0.02^{**} 0.01 (0.01) (0.01) (0.01) Credit anount (USD) PPP $=8.39^{*}$ -0.33 $=8.06$ 39.65 Reported in Table A.3 (5.03) (5.16) (5.05) (87.98) Hypothetical loan (1 year USD) PPP $=8.278$ -297.62^{**} 214.84 1579.91 Hypothetical loan (1 year USD) PPP -20.44 -353.65 333.12 3037.61 Reported in Table A.3 (215.14) (23.14) (23.64) 1075 Hypothetical loan (10 years USD) PPP -20.34 -405.37 764.99 Hypothetical loan (10 years USD) PPP -20.33 -405.37 764.99 Hypothetical loan (10 years USD) PPP -20.33 -405.37 764.99 Hypothetical loan (10 years USD) PPP -20.44 -353.65 333.12 3037.61 Reported in Table A.3 (21.514) (23.64) $(23$ | Focal outcome | (3.86) | (3.75) | (3.86) | (64.07) |
| Tas any savings -0.01 -0.001 -0.001 -0.001 -0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.001 0.002 (0.02) (0.040) 0.021 0.0401 0.003 0.047° 0.331 0.05° 0.331 0.06° 0.33 0.06° 0.33 0.06° 0.33 0.06° 0.33 0.06° 0.33 0.06° 0.33 10.07° 0.331 10.07° 0.331 10.07° 0.331 10.07° 0.331 10.07° 0.331 10.07° 0.031° 0.01° 0.03° 0.01° 0.03° 0.01° 0.03° 0.01° 0.03° 0.041 0.05° 0.05° 0.031 1057° | Reported in Table A.3 | [0.96] | [0.75] | [0.89] | 1890 |
| Reported in Table A.3 $[0.32]$ (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) (0.03) (0.03) (0.03) (0.04) (0.02) (0.40) (0.02) (0.40) (0.03) (0.03) (0.04) (0.03) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.04) (0.03) (0.03) (0.04) (0.03) (0.03) (0.04) (0.03) (0.03) (0.04) (0.03) (0.04) (0.03) (0.03) (0.04) (0.03) (0.03) (0.04) (0.03) </td <td>Has any savings</td> <td>(0.02)</td> <td>-0.01</td> <td>-0.00</td> <td>(0.21)</td> | Has any savings | (0.02) | -0.01 | -0.00 | (0.21) |
| Has outside savings $[0.53]$ $[0.53]$ $[0.53]$ $[0.53]$ $[0.53]$ $[0.53]$ $[0.53]$ Reported in Table A.3 (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) Has any credit -0.03 0.03 -0.06^{**} 0.33 Reported in Table A.3 (0.03) (0.03) (0.03) (0.03) (0.04) Has any agricultural credit 0.02^{**} 0.01 (0.01) (0.01) (0.07) Reported in Table A.3 (0.01) (0.01) (0.01) (0.07) $(0.05)^*$ (0.33) 1909 Credit amount (USD) PPP -8.39^* -0.33 -8.06 39.65 Reported in Table A.3 $(0.05)^*$ (0.33) $(0.55)^*$ (57.98) Hypothetical loan (1 year USD) PPP -82.78 -297.62^* 21.44 157.91 Hypothetical loan (5 years USD) PPP -20.54^* -33.56 33.12 3037.61 Reported in Table A.3 $(0.51)^*$ $(0.33)^*$ $(0.32)^*$ $(0.33)^*$ 1082 Hypothetical loan (10 years USD) PPP -20.54^* -35.35^* 33.12 3037.61 Reported in Table A.3 $(632.33)^*$ $(682.57)^*$ $(778.067)^*$ $(0.34)^*$ 1072 Hypothetical loan (10 years USD) PPP 15.72^* -18.17 33.90^* 214.72 Fool outcome $(0.34)^*$ 1073 1022 1022 PAP group 8 -1024^* 33.35^* 103.31^* 1022 Parted in Table A.3 $(0.01)^*$ <td< td=""><td>Reported in Table A.5</td><td>(0.02) [0.03]</td><td>(0.02) [0.75]</td><td>(0.02) [0.05]</td><td>(0.41)</td></td<> | Reported in Table A.5 | (0.02) [0.03] | (0.02) [0.75] | (0.02) [0.05] | (0.41) |
| $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | Has outside savings | [0.35] -0.01 | -0.01 | 0.00 | 0.20 |
| Inported in Table 1.05 (0.37) (0.93) (0.97) (0.93) (0.97) (0.93) (0.97) (0.93) (0.97) (0.93) (0.93) (0.97) (0.93) (0.93) (0.79) (0.93) (0.93) (0.79) (0.93) < | Reported in Table A 3 | (0.02) | (0.02) | (0.02) | (0.40) |
| Has any credit -0.03 0.03 -0.06^* 0.33 Reported in Table A.3 (0.03) (0.03) (0.03) (0.03) (0.03) Has any agricultural credit 0.02^{**} 0.01 0.01 0.001 Reported in Table A.3 (0.01) (0.01) (0.01) (0.07) Credit amount (USD) PPP -8.39^* -0.33 -8.06 Reported in Table A.3 (5.05) (5.03) (5.16) (5.55) Reported in Table A.3 (13797) (135.57) (134.58) (1974.66) Reported in Table A.3 (13797) (135.57) (134.58) (1974.66) Reported in Table A.3 (13797) (135.57) (134.58) (1974.66) Reported in Table A.3 (245.14) (234.41) (230.61) (3518.50) Reported in Table A.3 (245.14) (234.41) (230.61) (3518.50) Reported in Table A.3 (245.14) (234.41) (230.61) (3518.50) Reported in Table A.3 (245.14) (234.61) (230.61) (3518.50) Reported in Table A.3 (245.14) (234.61) (238.60) (277.68) Reported in Table A.3 (0.27) (0.38) (0.27) (778.067) Reported in Table A.3 (245.14) (234.61) (23.81) (24.72) Pool outcome (30.46) (29.60) (27.80) (27.80) Reported in Table A.3 (0.27) (0.38) (0.27) (780.67) Reported in Table 3 (0.28) | | [0.93] | [0.75] | [0.95] | 1909 |
| Reported in Table A.3 (0.03) < | Has any credit | -0.03 | 0.03 | -0.06** | 0.33 |
| Image: transmitter in the second s | Reported in Table A.3 | (0.03) | (0.03) | (0.03) | (0.47) |
| Has any agricultural credit 0.02^{++} 0.01^{+} 0.01^{+} 0.01^{+} 0.01^{-} 0.00^{-} Reported in Table A.3 (0.01) (0.01) (0.07) $(0.05)^{+}$ $[0.33]$ $[0.50]^{+}$ $[0.50]^{+}$ $[0.33]$ $[0.55]^{+}$ $[0.50]^{+}$ $[0.33]$ $[0.55]^{-}$ $[0.50]^{+}$ $[0.33]$ $[0.55]^{-}$ $[0.33]$ $[135,57]$ $(135,57)$ | * | [0.90] | [0.71] | [0.33] | 1909 |
| Reported in Table A.3 (0.01) (0.01) (0.01) (0.01) (0.01) (0.07) Credit amount (USD) PPP -8.39* -0.33 -8.06 39.65 Reported in Table A.3 (5.03) (5.16) (5.05) (87.98) Hypothetical loan (1 year USD) PPP -82.78 -297.62** 214.84 1579.91 Reported in Table A.3 (137.97) (135.57) (134.58) (197.466) Hypothetical loan (5 years USD) PPP -20.54 -353.65 333.12 3037.61 Reported in Table A.3 (245.14) (234.41) (230.61) (3518.50) Hypothetical loan (10 years USD) PPP -230.38 -495.37 264.99 5663.60 Reported in Table A.3 (632.33) (583.88) (608.57) (7780.67) Hypothetical loan (10 years USD) PPP 15.72 -18.17 33.90 214.72 Food outcome (30.46) (29.40) (28.00) (412.21) Not reported in Table 3 (30.10) (29.04) (28.00) (411.72) but without including animal purchases $[0.98]$ $[0.86]$ </td <td>Has any agricultural credit</td> <td>0.02***</td> <td>0.01^{*}</td> <td>0.01</td> <td>0.00</td> | Has any agricultural credit | 0.02*** | 0.01^{*} | 0.01 | 0.00 |
| $ \begin{array}{cccccccccccccccccccccccccccccccccccc$ | Reported in Table A.3 | (0.01) | (0.01) | (0.01) | (0.07) |
| $\begin{array}{llllllllllllllllllllllllllllllllllll$ | | $[0.05]^*$ | [0.30] | [0.79] | 1908 |
| Reported in Table A.3(5.03)(5.16)(5.05)(87.98)Hypothetical loan (1 year USD) PPP -82.78 -297.62^{**} 214.84 1579.91 Reported in Table A.3(137.97)(135.57)(134.58)(1974.66)Hypothetical loan (5 years USD) PPP -20.54 -353.65 333.12 307.61 Reported in Table A.3(245.14)(234.41)(230.61)(3518.50)Hypothetical loan (10 years USD) PPP -20.54 -353.65 333.12 307.61 Reported in Table A.3(245.14)(234.41)(230.61)(3518.50)Hypothetical loan (10 years USD) PPP -230.38 -495.37 264.99 566.360 Reported in Table A.3(632.33)(583.88)(608.57)(7780.67)Aggregate investment in livestock and agriculture (USD) PPP 15.72 -18.17 33.90 214.72 Focal outcome(30.46)(20.60)(28.78)(412.21)Not reported elseuhere[0.88][0.94][0.39]1061Total expenditure on inputs for livestock and poultry activities (USD) PPP (30.10) (20.00)(411.72)but without including animal purchases[0.98][0.86][0.39]1072Spending on seed or fertiliser 7.33^* 3.80 3.53 3.49 Reported in Table 3(30.7)(3.32)(3.31)(43.54)Reported in Table 3(30.7)(3.32)(3.31)(43.54)Reported in Table 3(30.7)(3.32)(3.61)(258.03)R | Credit amount (USD) PPP | -8.39^{*} | -0.33 | -8.06 | 39.65 |
| | Reported in Table A.3 | (5.03) | (5.16) | (5.05) | (87.98) |
| Hypothetical loan (1 year USD) PPP-82.78 (137.97) -297.62^{**} (134.58) 214.48 (1974.66)Reported in Table A.3[0.33][0.33][0.33]1082Hypothetical loan (5 years USD) PPP -20.54 -353.65 333.12 3037.61 Reported in Table A.3(245.14)(234.41)(230.61)(3518.50)Hypothetical loan (10 years USD) PPP -220.38 -495.37 264.99 5663.60 Reported in Table A.3(632.33)(583.88)(608.57)(7780.67)Hypothetical loan (10 years USD) PPP 15.72 -18.17 33.90 214.72 PAP group 8(30.46)(29.60)(28.78)(412.21)Not reported elsewhere[0.85][0.94][0.39]1061Total expenditure on inputs for livestock and poultry activities (USD) PPP 6.68 -271.8 27.86 189.80Reported in Table 3(30.10)(29.04)(28.00)(411.72)Not reported elsewhere[0.88][0.86][0.39]1072Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3(30.7)(3.22)(3.31)(4354)Value of family crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3(9.95][0.39]1078Value of family crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3(0.12)(0.05](0.39)1079Spending on hired crop labour (U | | [0.43] | [0.95] | [0.33] | 1897 |
| Reported in Table A.3(137.97)(135.57)(134.58)(1974.66)[0.33][0.33][0.33][0.33][0.33]1082Hypothetical loan (5 years USD) PPP -20.54 -353.65 333.12 3037.61 Reported in Table A.3(245.14)(234.41)(230.61)(3518.50)[0.96][0.40][0.34]1075Reported in Table A.3(632.33)(583.88)(608.57)(7780.67)Reported in Table A.3[0.93][0.71][0.89]1022PAP group 8[0.33][0.39]1022[0.39]1022PAP group 8[0.85][0.94][0.39]1022Focal outcome(30.46)(29.60)(28.78)(412.21)Not reported elsewhere[0.85][0.94][0.39]1061Total expenditure on inputs for livestock and poultry activities (USD) PPP 6.68 -27.18 27.86 189.80Reported in Table 3(30.10)(29.04)(28.00)(411.72)but without including animal purchases[0.98][0.86][0.39]1078Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3(19.73)(19.39)(20.08)(258.03)Value of family crop labour (USD) PPP 1.30 4.977 3.67 54.16 Reported in Table 3(0.12)[0.86][0.50]1078Spending on hired crop labour (USD) PPP 1.30 4.977 3.67 54.16 Reported in Table 3 <td>Hypothetical loan (1 year USD) PPP</td> <td>-82.78</td> <td>-297.62**</td> <td>214.84</td> <td>1579.91</td> | Hypothetical loan (1 year USD) PPP | -82.78 | -297.62** | 214.84 | 1579.91 |
| $\begin{array}{c c c c c c c c c c c c c c c c c c c $ | Reported in Table A.3 | (137.97) | (135.57) | (134.58) | (1974.66) |
| Hypothetical loan (5 years USD) PPP -20.54 -333.65 333.12 3037.61 Reported in Table A.3[0.96][0.40][0.34] 1075 Hypothetical loan (10 years USD) PPP -230.38 -495.37 264.99 5663.60 Reported in Table A.3(632.33) (583.88) (608.57) (7780.67) PAP group 8[0.93] $[0.71]$ $[0.89]$ 1022 PAP group 8(30.46) (22.60) (28.78) (412.21) Not reported elsewhere $[0.85]$ $[0.94]$ $[0.39]$ 1061 Total expenditure on inputs for livestock and poultry activities (USD) PPP 6.68 -27.18 27.86 189.80 Reported in Table 3(30.10) (29.04) (28.00) (411.72) but without including animal purchases $[0.98]$ $[0.86]$ $[0.33]$ 1072 Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3 (3.07) (3.32) (3.31) (43.54) Value of family crop labour (USD) PPP 3.33^{*} 1.27 3.667 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) Nate acultivated (hectares) 0.01 -0.01 0.02 (0.32) Area cultivated (hectares) 0.01 -0.01 0.02 (0.02) Not reported elsewhere (0.01) (0.01) (0.07) Not reported in Table 3 (0.02) (0.02) (0.30) Not reported in Table 3 $(0$ | | [0.93] | [0.25] | [0.33] | 1082 |
| Reported in Table A.3 (245.14) (230.61) (3518.50) Hypothetical loan (10 years USD) PPP -230.38 -495.7 264.99 5663.60 Reported in Table A.3 (632.33) (583.88) (608.57) (7780.67) $[0.93]$ $[0.71]$ $[0.89]$ 1022 PAP group 8 (230.46) (28.78) (142.21) Aggregate investment in livestock and agriculture (USD) PPP 15.72 -18.17 33.90 214.72 Focal outcome (30.46) (29.60) (28.78) (412.21) Not reported elsewhere $[0.85]$ $[0.94]$ $[0.39]$ 1061 Total expenditure on inputs for livestock and poultry activities (USD) PPP 0.68 -27.18 27.86 189.80 Reported in Table 3 (30.10) (29.04) (28.00) (411.72) but without including animal purchases $[0.98]$ $[0.86]$ $[0.39]$ 1072 Spending on seed or fertiliser 7.33^* 3.80 3.53 33.49 Reported in Table 3 (3.07) (3.22) (3.31) (43.54) Value of family crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) Area cultivated (hectares) 0.01 -0.00 0.02 0.02 (0.02) (0.02) </td <td>Hypothetical Ioan (5 years USD) PPP</td> <td>-20.54</td> <td>-353.65</td> <td>333.12</td> <td>3037.61</td> | Hypothetical Ioan (5 years USD) PPP | -20.54 | -353.65 | 333.12 | 3037.61 |
| $ 0.56 $ $ 0.40 $ $ 0.34 $ 1075 Hypothetical loan (10 years USD) PPP -230.38 -495.37 264.99 5663.60 Reported in Table A.3 (632.33) (538.88) (608.57) (7780.67) $ 0.93 $ $[0.71]$ $[0.89]$ 1022 PAP group 8 (30.46) (29.60) (28.78) (412.21) Not reported elsewhere (30.46) (29.60) (28.78) (412.21) Not reported elsewhere (30.46) (29.00) (28.78) (412.21) Not reported in Table 3 (30.10) (29.04) (28.00) (411.72) but without including animal purchases $[0.98]$ $[0.86]$ $[0.39]$ 1072 Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3 (3.07) (3.32) (3.31) (43.54) Value of family crop labour (USD) PPP 1.30 -4.97 3.67 54.16 Reported in Table 3 (19.73) (19.39) (20.08) (258.03) Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (0.01) (0.02) (0.02) (0.02) (0.02) Area cultivated (hectares) 0.01 -0.01 0.02 0.02 (0.02) (0.30) Area rented in or sharecropped (hectares) 0.01 -0.01 0.02 (0.39) 1078 Not reported elsewhere (0.02) (0.02) (0.02) (0.30) (0.07) </td <td>Reported in Table A.3</td> <td>(245.14)</td> <td>(234.41)</td> <td>(230.61)</td> <td>(3518.50)</td> | Reported in Table A.3 | (245.14) | (234.41) | (230.61) | (3518.50) |
| Pypothetical Ioan (10 years USD) PPP -230.38 -495.37 204.39 5005.00 Reported in Table A.3[632.33](583.88)(608.57)(7780.67)[0.93][0.71][0.89]1022PAP group 8[0.30][0.71][0.89]1022Aggregate investment in livestock and agriculture (USD) PPP15.72 -18.17 33.90 214.72 Focal outcome(30.46)(29.60)(28.78)(412.21)Not reported elsewhere[0.85][0.94][0.39]1061Total expenditure on inputs for livestock and poultry activities (USD) PPP 0.68 -27.18 27.86 189.80Reported in Table 3(30.10)(29.04)(28.00)(411.72)Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3(3.07)(3.32)(3.31)(43.54)[0.12][0.86][0.39]1072Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3(19.73)(19.39)(20.08)(258.03)Reported in Table 3(19.73)(19.39)(20.08)(258.03)Reported in Table 3(0.12)[0.95][0.39]1079Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3(0.01)(0.02)(0.02)(0.02)Area cultivated (hectares)(0.01)(0.02)(0.02)(0.03)Area cultivated (hectares)(0 | II with the line (10, and U(D)) DDD | [0.96] | [0.40] | [0.34] | 1075 |
| Reported in Table 3 (052.53) (058.53) (008.51) (1780.61) PAP group 8[0.71] $[0.89]$ 1022 Aggregate investment in livestock and agriculture (USD) PPP 15.72 -18.17 33.90 214.72 Focal outcome (30.46) (29.60) (28.78) (412.21) Not reported elsewhere $[0.85]$ $[0.94]$ $[0.39]$ 1061 Total expenditure on inputs for livestock and poultry activities (USD) PPP 0.68 -27.18 27.86 189.80 Reported in Table 3 (30.10) (29.04) (28.00) (411.72) but without including animal purchases $[0.98]$ $[0.86]$ $[0.39]$ 1072 Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3 (3.07) (3.32) (3.31) (43.54) Value of family crop labour (USD) PPP 10.12 $[0.86]$ $[0.39]$ 1078 Value of family crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (19.73) (19.39) (20.08) (258.03) $[0.32]$ $[0.35]$ $[0.39]$ 1079 Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.30) Area cultivated (hectares) 0.01 -0.00 0.01 0.02 Area cultivated (hectares) | Buy and in Table 4.2 | -230.38 | -493.37 | 204.99 | 0003.00 (7790.67) |
| $PAP \ group 8$ Aggregate investment in livestock and agriculture (USD) PPP15.72-18.1733.90214.72Focal outcome(30.46)(29.60)(28.78)(412.21)Not reported elsewhere[0.85][0.94][0.39]1061Total expenditure on inputs for livestock and poultry activities (USD) PPP0.68-27.1827.86189.80Reported in Table 3(30.10)(29.04)(28.00)(411.72)but without including animal purchases[0.98][0.86][0.39]1072Spending on seed or fertiliser7.33**3.803.5333.49Reported in Table 3(3.07)(3.32)(3.31)(43.54)(D.12][0.86][0.39]10781078Value of family crop labour (USD) PPP33.33*1.2732.06387.81Reported in Table 3(19.73)(19.39)(20.08)(258.03)[0.32][0.95][0.39]1079Spending on hired crop labour (USD) PPP-1.30-4.973.6754.16Reported in Table 3(0.02)(0.02)(0.02)(0.02)(0.30)Area cultivated (hectares)0.01-0.010.020.55Reported in Table 3(0.02)(0.02)(0.02)(0.30)Area rented in or sharecropped (hectares)0.01-0.010.02(0.30)Not reported elsewhere(0.01)(0.01)(0.07)(0.34)1068 | Reported in Tuble A.5 | (052.55) [0.02] | (000.00) [0.71] | [0.80] | (7780.07) |
| Aggregate investment in livestock and agriculture (USD) PPP 15.72 -18.17 33.90 214.72 Focal outcome (30.46) (29.60) (28.78) (412.21) Not reported elsewhere $[0.85]$ $[0.94]$ $[0.39]$ 1061 Total expenditure on inputs for livestock and poultry activities (USD) PPP 0.68 -27.18 27.86 189.80 Reported in Table 3 (30.10) (29.04) (28.00) (411.72) but without including animal purchases $[0.98]$ $[0.86]$ $[0.39]$ 1072 Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3 (3.07) (3.32) (3.31) (43.54) $[0.12]$ $[0.86]$ $[0.39]$ 1078 Value of family crop labour (USD) PPP 33.33^* 1.27 32.06 387.81 Reported in Table 3 (19.73) (19.39) (20.08) (258.03) $[0.32]$ $[0.32]$ $[0.39]$ 1079 Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (0.51) (5.42) (93.01) $[0.32]$ $[0.55]$ $[0.50]$ 1078 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.30) 10.341 $[0.95]$ $[0.39]$ 1071 Area cultivated (hectares) 0.01 -0.00 0.01 0.02 0.51 < | PAP group 8 | [0.30] | [0.71] | [0.09] | 1022 |
| Aggregate investment in investock and agriculture (USD) FFF 15.72 -16.17 35.90 214.72 Focal outcome(30.46)(29.60)(28.78)(412.21)Not reported elsewhere[0.85][0.94][0.39]1061Total expenditure on inputs for livestock and poultry activities (USD) PPP 0.68 -27.18 27.86 189.80 Reported in Table 3(30.10)(29.04)(28.00)(411.72)but without including animal purchases[0.98][0.86][0.39]1072Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3(3.07)(3.32)(3.31)(43.54)Value of family crop labour (USD) PPP 33.33^{*} 1.27 32.06 387.81 Reported in Table 3(19.73)(19.39)(20.08)(258.03) α spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3(5.45)(5.51)(5.42)(93.01) α cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3(0.02)(0.02)(0.02)(0.03) 1071 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3(0.01)(0.01)(0.07) 10.341 10.95 10.39 1072 | Agreement in livestock and agriculture (LICD) DDD | 15 79 | 10 17 | 22.00 | 914 79 |
| Not reported elsewhere (30.40) (25.76) (25.78) (12.21) Not reported elsewhere $[0.85]$ $[0.94]$ $[0.39]$ 1061 Total expenditure on inputs for livestock and poultry activities (USD) PPP 0.68 -27.18 27.86 189.80 Reported in Table 3 (30.10) (29.04) (28.00) (411.72) but without including animal purchases $[0.98]$ $[0.86]$ $[0.39]$ 1072 Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3 (3.07) (3.32) (3.31) (43.54) Value of family crop labour (USD) PPP 33.33^{*} 1.27 32.06 387.81 Reported in Table 3 (19.73) (19.39) (20.08) (258.03) Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) $(9.55]$ $[0.86]$ $[0.50]$ 1078 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.01 Not reported elsewhere (0.01) (0.01) (0.07) (0.07) Not reported elsewhere 0.01 0.01 0.02 0.02 | Focal outcome | (30.46) | (20.60) | (28.78) | (412.01) |
| Note to prote discurre $[0.53]$ < | Not reported elsewhere | [0.85] | (25.00) [0.94] | [0.39] | 1061 |
| Reported in Table 3(30.10)(29.04)(28.00)(411.72)but without including animal purchases $[0.98]$ $[0.86]$ $[0.39]$ 1072 Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3 $[0.12]$ $[0.86]$ $[0.39]$ 1078 Value of family crop labour (USD) PPP 33.33^* 1.27 32.06 387.81 Reported in Table 3 $[0.12]$ $[0.86]$ $[0.39]$ 1078 Value of family crop labour (USD) PPP 33.33^* 1.27 32.06 387.81 Reported in Table 3 $[0.32]$ $[0.95]$ $[0.39]$ 1079 Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 $[0.55]$ $[0.551)$ (5.42) (93.01) Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (0.01) (0.02) (0.02) (0.02) (0.02) Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.30) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.02 Not reported elsewhere (0.01) (0.01) (0.01) (0.07) (0.34) $[0.95]$ $[0.39]$ 1068 | Total expenditure on inputs for livestock and poultry activities (USD) PPP | 0.68 | -27.18 | 27.86 | 189.80 |
| but without including animal purchases $[0.98]$ $[0.86]$ $[0.39]$ 1072 Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3 (3.07) (3.32) (3.31) (43.54) $[0.12]$ $[0.86]$ $[0.39]$ 1078 Value of family crop labour (USD) PPP 33.33^* 1.27 32.06 387.81 Reported in Table 3 (19.73) (19.39) (20.08) (258.03) $[0.32]$ $[0.95]$ $[0.39]$ 1079 Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) $[0.95]$ $[0.86]$ $[0.50]$ 1078 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.02) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.02 Not reported elsewhere (0.01) (0.01) (0.01) (0.07) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | Reported in Table 3 | (30.10) | (29.04) | (28.00) | (411.72) |
| Spending on seed or fertiliser 7.33^{**} 3.80 3.53 33.49 Reported in Table 3 (3.07) (3.32) (3.31) (43.54) $[0.12]$ $[0.86]$ $[0.39]$ 1078 Value of family crop labour (USD) PPP 33.33^* 1.27 32.06 387.81 Reported in Table 3 (19.73) (19.39) (20.08) (258.03) $[0.32]$ $[0.95]$ $[0.39]$ 1079 Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) $[0.95]$ $[0.86]$ $[0.50]$ 1078 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) Not reported in or sharecropped (hectares) 0.01 -0.00 0.01 0.02 Not reported elsewhere (0.01) (0.01) (0.01) (0.07) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | but without including animal purchases | [0.98] | [0.86] | [0.39] | 1072 |
| Reported in Table 3 (3.07) (3.32) (3.31) (43.54) $[0.12]$ $[0.86]$ $[0.39]$ 1078 Value of family crop labour (USD) PPP 33.33^* 1.27 32.06 387.81 Reported in Table 3 (19.73) (19.39) (20.08) (258.03) $[0.32]$ $[0.95]$ $[0.39]$ 1079 Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) $[0.95]$ $[0.86]$ $[0.50]$ 1078 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.02) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.02 Not reported elsewhere (0.01) (0.01) (0.01) (0.07) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | Spending on seed or fertiliser | 7.33** | 3.80 | 3.53 | 33.49 |
| $[0.12]$ $[0.36]$ $[0.39]$ 1078 Value of family crop labour (USD) PPP 33.33^* 1.27 32.06 387.81 Reported in Table 3 (19.73) (19.39) (20.08) (258.03) $[0.32]$ $[0.95]$ $[0.39]$ 1079 Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) $[0.95]$ $[0.86]$ $[0.50]$ 1078 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.02) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.01 Not reported elsewhere (0.01) (0.01) (0.01) (0.07) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | Reported in Table 3 | (3.07) | (3.32) | (3.31) | (43.54) |
| Value of family crop labour (USD) PPP 33.33^* 1.27 32.06 387.81 Reported in Table 3 (19.73) (19.39) (20.08) (258.03) Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.01 0.02 Not reported elsewhere (0.01) (0.01) (0.01) (0.01) (0.01) (0.01) | | [0.12] | [0.86] | [0.39] | 1078 |
| Reported in Table 3 (19.73) (19.39) (20.08) (258.03) [0.32][0.32] $[0.95]$ $[0.39]$ 1079 Spending on hired crop labour (USD) PPP -1.30 -4.97 3.67 54.16 Reported in Table 3 (5.45) (5.51) (5.42) (93.01) $[0.95]$ $[0.86]$ $[0.50]$ 1078 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.02) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.01 Not reported elsewhere (0.01) (0.01) (0.01) (0.07) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | Value of family crop labour (USD) PPP | 33.33^{*} | 1.27 | 32.06 | 387.81 |
| $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$ | Reported in Table 3 | (19.73) | (19.39) | (20.08) | (258.03) |
| $\begin{array}{llllllllllllllllllllllllllllllllllll$ | | [0.32] | [0.95] | [0.39] | 1079 |
| Reported in Table 3 (5.45) (5.51) (5.42) (93.01) [0.95] $[0.86]$ $[0.50]$ 1078 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.02) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.02 Not reported elsewhere (0.01) (0.01) (0.01) (0.07) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | Spending on hired crop labour (USD) PPP | -1.30 | -4.97 | 3.67 | 54.16 |
| [0.95] $[0.86]$ $[0.50]$ 1078 Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.02) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.01 Not reported elsewhere (0.01) (0.01) (0.01) (0.01) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | Reported in Table 3 | (5.45) | (5.51) | (5.42) | (93.01) |
| Area cultivated (hectares) 0.01 -0.01 0.02 0.55 Reported in Table 3 (0.02) (0.02) (0.02) (0.02) (0.02) (0.02) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.02 Not reported elsewhere (0.01) (0.01) (0.01) (0.01) (0.01) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | | [0.95] | [0.86] | [0.50] | 1078 |
| Reported in Table 3 (0.02) (0.03) Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.02 (0.01) (0.01) (0.01) (0.02) Not reported elsewhere (0.01) (0.01) (0.01) (0.01) (0.02) (0.02) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | Area cultivated (hectares) | (0.01) | -0.01 | (0.02) | 0.55 |
| [0.85] $[0.95]$ $[0.39]$ 1071 Area rented in or sharecropped (hectares) 0.01 -0.00 0.01 0.02 Not reported elsewhere (0.01) (0.01) (0.01) (0.07) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | Keported in Table 3 | (0.02) | (0.02) | (0.02) | (0.30) |
| Area reneed in or snarecropped (nectares) 0.01 -0.00 0.01 0.02 Not reported elsewhere (0.01) (0.01) (0.01) (0.01) $[0.34]$ $[0.95]$ $[0.39]$ 1068 | Area repted in an chargement of (hesterne) | [0.85] | [0.95] | [0.39] | 1071 |
| $\begin{bmatrix} 0.01 \\ 0.02 \\ 0.05 \\ 0.05 \\ 0.05 \\ 0.001 \\ 0.01$ | Net reported alcoubere | 0.01 | -0.00 | 0.01 | (0.02) |
| | inor reported elsewhere | [0.34] | [0.95] | [0.39] | 1068 |

Table A.14: PRE-SPECIFIED GROUPS 7,8

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences in Columns 1 and 2. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and characteristics of the household head: age, years of education, an indicator for being single, and an indicator for being male. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each group (panel) of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in detail in Appendix G.1, or in the main tables where they are reported. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian bir) PPP. All columns control for being male. The unit of observation is the individual for PAP group 7, except for the hypothetical loans (that are for the household). The unit of observation is the household (and controls refer to the household head only) for group 8. The number of observations varies slightly across rows because some respondents do not answer all questions.

| | (1) | (2) | (3) | (4) |
|---|------------|------------------|-----------------------|------------------------------------|
| | Aft | er five ye | ars | |
| PAP group 9 (risk preferences) | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. |
| 1 if extremely risk avera (coin) | -0.04 | -0.01 | -0.03 | 0.38 |
| Not reported elsewhere | (0.04) | (0.03) | (0.03) | (0.49) |
| | [0.44] | [0.86] | [0.44] | 1909 |
| 1 if extremely risk averse (market) | -0.02 | 0.01 | -0.02 | 0.37 |
| Raw scores reported in Table 6 | (0.03) | (0.03) | (0.03) | (0.48) |
| | [0.56] | [0.86] | [0.44] | 1909 |
| PAP group 9 (time preferences) | | | | |
| Near term discount factor | -0.03 | -0.02 | -0.01 | 0.42 |
| Reported with different definition in Table 6 | (0.02) | (0.02) | (0.02) | (0.29) |
| | [0.11] | [0.21] | [0.70] | 1909 |
| PAP group 9 (information) | | | | |
| Income from trading | 0.01 | -0.03 | 0.04 | 0.23 |
| Reported in Table A.4 | (0.03) | (0.03) | (0.03) | (0.42) |
| | [0.88] | [0.55] | [0.33] | 1104 |
| Income from grain milling | -0.00 | 0.00 | -0.01 | 0.00 |
| Dropped because of low variation | (0.00) | (0.00) | (0.00) | (0.05) |
| | [0.77] | [0.74] | [0.29] | 1088 |
| Information sub-index pre-specified | 0.13* | 0.02 | 0.11 | 0.00 |
| Components reported separately in Table A.4 | (0.07) | (0.06) | (0.07) | (1.00) |
| | [0.52] | [0.80] | [0.29] | 1089 |
| Produces cash crops | -0.02 | 0.04 | -0.06 | 0.61 |
| Reported in Table A.4 | (0.03) | (0.03) | (0.04) | (0.49) |
| Uses immigration | 0.02 | 0.05* | [0.29] | 1089 |
| Benorted in Table A / | (0.02) | -0.03 | (0.03) | (0.24) |
| Reported in Table A.4 | [0.03] | [0.55] | [0.05] | 1089 |
| Uses stone hands | -0.03 | -0.02 | -0.02 | 0.69 |
| Reported in Table A / | (0.03) | (0.02) | (0.02) | (0.46) |
| heporica in 1000 11.4 | [0.00) | [0.78] | [0.62] | 1104 |
| Applies water harvesting | 0.00 | -0.02 | 0.02 | 0.07 |
| Reported in Table A.4 | (0.02) | (0.02) | (0.02) | (0.26) |
| | [0.99] | [0.55] | [0.36] | 1104 |
| Uses water storage | 0.01^{*} | 0.00 | 0.01^{*} | 0.00 |
| Dropped because of low variation | (0.01) | (0.00) | (0.01) | (0.05) |
| | [0.52] | [0.98] | [0.29] | 1089 |
| Uses crop rotation | -0.03 | -0.05 | 0.02 | 0.53 |
| Reported in Table A.4 | (0.03) | (0.03) | (0.03) | (0.50) |
| | [0.77] | [0.55] | [0.62] | 1089 |
| Uses pump | 0.00 | 0.01 | -0.01 | 0.01 |
| Reported in Table A.4 | (0.01) | (0.01) | (0.01) | (0.10) |
| TT (1) | [1.00] | [0.55] | [0.29] | 1088 |
| Uses cattle | 0.01 | -0.04 | (0.05) | 0.70 |
| Reported in Table A.4 | (0.03) | (0.03) [0.55] | (0.03) | (0.46) |
| | [0.88] | [0.55] | [0.29] | 1088 |
| Information index pre-specified | -0.08 | -0.12 | 0.04 | 0.00 |
| | (0.07) | (0.08) | (0.08) | (1.00) |
| | 0.26] | [0.12] | 0.63] | 1152 |

Table A.15: Pre-specified group 9

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences between parameters reported in the first two columns. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and individual: age, years of education, an indicator for being single, and an indicator for being male. Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets and are calculated over each group (panel) of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in detail in Appendix G.1, or in the main tables where they are reported. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. Heteroskedasticity-robust standard errors are in parentheses. The unit of observation is the household. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 99^{th} percentile. Table 6 reports the raw answers (coded from 1-5, where 5 is extremely risk averse) for the hypothetical maize sale (market), whereas the pre-specified outcome was an indicator equal to one if the respondent was coded to have answered 5 (in either coin-flip scenario or the hypothetical maize sale). The variables in the information panel are measured at the household-level and use the household head characteristics as controls. Two variables that we pre-specified to be part of the index did not have sufficient variation. Only 3 households reported earning income from grain milling and 7 reported using water storage (level basin irrigation).

| | (1) | (2) | (3) | (4) |
|---|--------------------|-------------------|---------------------------|-------------------|
| | | () | (-) | () |
| | Alt | er nve yea | us | |
| | - | D 1 1 | Treat. vs. | Control mean |
| | Treatment | Placebo | placebo | (SD) Tetal ala |
| PAP group 10 | | | | Total obs. |
| | | | | |
| Consumption aggregate (12m recall USD) PPP | 97.87 | 53.33 | 44.54 | 2764.68 |
| Pocal outcome | (111.93) [0.28] | [0.04] | (119.10) [0.85] | (1530.00) |
| Food consumption (USD) por ad ocuiv monthly PPP | 1.08 | 2.20 | 0.32 | 53.01 |
| Reported in Table 5 | (2.05) | (1.92) | (2.07) | (29.98) |
| | [0.38] | [0.70] | [0.88] | 1076 |
| Frequent non-food (1m recall USD) per ad. equiv. PPP | 0.44 | 0.04 | 0.40 | 4.08 |
| Reported in Table 5 | (0.28) | (0.28) | (0.30) | (3.69) |
| | [0.23] | [0.94] | [0.30] | 1076 |
| Nonfood consumption (12m recall USD) per ad. equiv. monthly PPP | 0.70 | -0.54 | 1.24** | 7.47 |
| Reported in Table 5 | (0.51) | (0.43) | (0.48) | (6.35) |
| Computing of signatures and takages (USD) DDD | [0.26] | [0.70] | [0.06]* | 1079 |
| Reported in Table 5 | (0.13) | (0.12) | (0.17) | (1.66) |
| | [0.23] | [0.12] | [0.10] | 1078 |
| General economic position (scale 1 to 4) | 0.09* | 0.00 | 0.09* | 2.10 |
| Reported in Table 5 | (0.05) | (0.05) | (0.05) | (0.73) |
| | [0.23] | [0.94] | [0.25] | 1088 |
| PAP group 11 | | | | |
| Food security index: z-score | -0.06 | -0.10 | 0.04 | 0.48 |
| Focal outcome | (0.06) | (0.06) | (0.06) | (0.92) |
| Reported in Table 5 | [0.31] | [0.20] | [0.54] | 1084 |
| Months of food insecurity | -0.32** | 0.03 | -0.35** | 2.71 |
| Reported in Table 5 | (0.14) | (0.15) | (0.14) | (2.13) |
| DAD man 10 | $[0.05]^*$ | [0.85] | $[0.03]^{**}$ | 1088 |
| PAP group 12 | | | | |
| Asset aggregate (USD) PPP | 272.83* | -88.00 | 360.82** | 2190.42 |
| Focal outcome | (144.82) | (138.95) | (144.27) | (1974.69) |
| Not reported elsewhere Value of livesteek (USD) PDP | [0.08]* 197.29 | [0.70] 199.78 | $[0.04]^{**}$ 210.16** | 1078 |
| Reported in Table 3 | (135.80) | (130.84) | (130.10) | (1021.09) |
| | [0.17] | [0.70] | [0.04]** | 1080 |
| Value of tools (USD) PPP | 27.55** | 12.17 | 15.37 | 106.02 |
| Reported in Table 3 | (11.61) | (12.35) | (13.66) | (126.90) |
| | $[0.07]^*$ | [0.70] | [0.26] | 1077 |
| Value of durable assets excluding tools (USD) PPP | 21.95** | -2.93 | 24.88** | 70.55 |
| Reported in Table 5 | (10.75) | (9.22) | (11.18) | (127.39) |
| DAD man 19 | $[0.08]^*$ | [0.75] | $[0.04]^{**}$ | 1077 |
| PAP group 13 | | | | |
| Subjective well-being index (Anderson, 2008) | 0.10 | 0.05 | 0.05 | 0.01 |
| Focal outcome | (0.06) | (0.06) | (0.06) | (0.99) |
| Not reported elsewhere Bost life | 0.22** | 0.06 | [0.02] | 1909 |
| Reported in Table 3 | (0.11) | (0.11) | (0.17) | (1.80) |
| | [0.13] | [0.61] | [0.43] | 1909 |
| Happiest life | 0.11 | 0.12 | -0.01 | 6.05 |
| Reported in Table 3 | (0.14) | (0.14) | (0.14) | (2.19) |
| | [0.42] | [0.61] | [0.95] | 1909 |
| PAP group 14 | | | | |
| Total revenue (USD) PPP | 66.25 | -43.94 | 110.19 | 1345.87 |
| Focal outcome | (92.41) | (92.30) | (93.35) | (1236.93) |
| Reported in Table A.5 with minor change | [0.79] | [0.63] | [0.52] | 1076 |
| Crop revenue (USD) PPP | 22.06 | 21.39 | 0.67 | 383.70 |
| neported in Table A.d | (22.45) [0.70] | (22.54) [0.62] | (23.97) [0.09] | (300.60) 1077 |
| Livestock revenue (USD) (using self-reported prices) PPP | [0.79] 4 45 | [0.05] -98.49 | [0.96] 102.94 | 726.36 |
| Reported in Table A.5 including own-consumption | (73.05) | (71.25) | (70.72) | (994.60) |
| | [0.95] | [0.63] | [0.52] | 1078 |
| Wage labour revenue (USD) PPP | -3.17 | 6.01 | -9.18 | 25.86 |
| Reported in Table A.5 | (8.04) | (9.16) | (9.07) | (111.32) |
| | [0.87] | [0.63] | [0.52] | 1080 |
| Non-farm revenue (USD) PPP | 21.24 | 14.99 | 6.25 | 159.94 |
| Reported in Table A.5 | (28.81) [0.70] | (29.84) | (31.79) [0.02] | (353.37) |
| | 0.79 | 0.03 | 0.98 | 1010 |

Table A.16: PRE-SPECIFIED GROUPS 10,11,12,13,14

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences between Columns 1 and 2. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and characteristics of the household head: age, years of education, an indicator for being single, and an indicator for being male. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each group (panel) of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in detail in Appendix G.1, or in the main tables where they are reported. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. Heteroskedasticity-robust standard errors are in parentheses. The unit of observation is the household. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 99th percentile. The variables in the group 13 are measured at the individual respondent-level (household head or spouse) and use the respondents' characteristics as controls.

| | After five years | | | |
|--|------------------|---------|-----------------------|------------------------------------|
| | (1) | (2) | (3) | (4) |
| | Treatment | Placebo | Treat. vs. placebo | Control mean (SD) Total obs. |
| Agricultural investment (PAP) index | 0.12^{*} | -0.03 | 0.16** | 0.00 |
| | (0.07) | (0.07) | (0.07) | (1.00) |
| | $[0.08]^*$ | [0.93] | $[0.02]^{**}$ | 1088 |
| Educational investment (PAP) index | 0.18^{**} | 0.01 | 0.17^{**} | 0.00 |
| | (0.07) | (0.07) | (0.07) | (1.00) |
| | $[0.02]^{**}$ | [0.93] | $[0.02]^{**}$ | 1089 |
| Welfare (PAP) index | 0.15^{**} | -0.04 | 0.19^{**} | 0.00 |
| | (0.08) | (0.07) | (0.08) | (1.00) |
| | $[0.06]^*$ | [0.93] | $[0.02]^{**}$ | 1087 |
| Aspirations (PAP) index | 0.07^{*} | -0.01 | 0.08^{**} | 0.19 |
| | (0.04) | (0.04) | (0.04) | (0.67) |
| | $[0.08]^*$ | [0.93] | $[0.05]^{**}$ | 1909 |
| Expectations (PAP) index | 0.13^{***} | 0.01 | 0.11^{***} | 0.06 |
| | (0.04) | (0.04) | (0.04) | (0.67) |
| | $[0.01]^{***}$ | [0.93] | $[0.02]^{**}$ | 1909 |
| Aspirations and expectations aggregate (PAP) index | 0.15^{**} | -0.01 | 0.15^{**} | 0.01 |
| | (0.06) | (0.06) | (0.06) | (1.00) |
| | $[0.02]^{**}$ | [0.93] | $[0.02]^{**}$ | 1909 |
| Omnibus (PAP) index | 0.23*** | -0.03 | 0.26*** | 0.00 |
| | (0.07) | (0.07) | (0.07) | (1.00) |
| | $[0.01]^{***}$ | [0.93] | $[0.00]^{***}$ | 1091 |

Table A.17: Summary indices following PAP variable definitions and groupings

Notes: OLS estimates of within-village treatment and placebo effects five years after the intervention (columns 1-2). Column 3 tests for differences between Columns 1 and 2. The comparison group comprises households from the 64 treated villages that were not invited to any screening. Column 4 displays the control mean, standard deviation, and total number of observations. All columns control for village fixed effects and characteristics of the household head (and spouse for individual-level outcomes): age, years of education, an indicator for being single, and an indicator for being male. For the Aspirations, Expectations, and the Aggregate Index, the unit of observation is the individual and standard errors in parentheses are clustered at the household-level. For the other indices the unit of observation is the household and standard errors in parentheses are robust to heteroscedasticity. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 99^{th} percentile. The Agricultural investment PAP index comprises the outcomes listed in the 'Investment-oriented behaviour' table (PAP, p. 28): specifically: total expenditure on agricultural inputs, livestock inputs, family and hired labour, total land area under cultivation, and land rented or sharecropped. The Educational investment PAP index includes outcomes from the PAP (p. 26), specifically number of children enrolled (ages 6-15) and expenditure on their schooling. The Welfare PAP index comprises the outcomes in the listed under 'household welfare': food consumption, frequent and infrequent expenditures, cigarette and tobacco consumption (positive), and a general well-being index (PAP, p. 28). The Aspirations and Expectations indices are standardised weighted averages of respondents' aspired (expected) income, wealth, social status, and children's education, weighted by subjective weights across these dimensions, with social status the additional category as listed in the PAP. The Aspirations and Expectations Aggregate PAP index is constructed following Anderson (2008), like in Table 7, but also includes the social status component. The Omnibus index combines the Agricultural investment PAP, Educational investment PAP, Welfare PAP, and Aspirations and Expectations Aggregate PAP indices into a single omnibus index, following Kling, Liebman, and Katz (2007).

H Robustness tests

The section presents estimates from two alternative specifications, our pre-specified "non-ANCOVA" specification and one selecting controls via post double selection LASSO. Columns 1 and 2 report the estimates in the main paper for comparison (Equation (1)).

Columns 3 and 4 report our pre-specified "non-ANCOVA" specification. This is the same as Equation (1), except it does not control for the baseline value of the outcome, as we do not have baseline data for all variables. This specification maintains closer comparability across outcomes and with the comparison to pure control villages (where we do not have baseline outcomes). For some variables, the first and second columns are blank, because we do not have baseline values of the outcome variables and cannot run an ANCOVA specification. For those outcomes, the main tables show the pre-specified specification (without controls for baseline outcomes), which appears here in Columns 3 and 4.

Columns 5 and 6 report a specification selecting controls via post double selection LASSO (Belloni, Chernozhukov, and Hansen, 2014), where all variables in Tables A.9 and A.10 can be selected as controls. These variables include pre-specified controls X_{vi1} , other demographic variables, and baseline values of all outcomes. We partial out village-fixed effects and, when available, the baseline value of the outcome, following the recommendation in Cilliers, Elashmawy, and McKenzie (2024).

| After five years | ANC | OVA | Pre-sp | Pre-specified | | PDS Lasso | | |
|--|----------------|-----------------------|----------------|-----------------------|----------------|-----------------------|------------------------------------|--|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | |
| | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Control mean (SD) Total obs. | |
| Agricultural investment index | 0.21^{***} | 0.18^{***} | 0.23*** | 0.21^{***} | 0.17^{***} | 0.12^{**} | -0.00 | |
| - | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (1.00) | |
| | [0.00]*** | $[0.01]^{***}$ | [0.00]*** | [0.00]*** | [0.00]*** | [0.04]** | 1082 | |
| Educational investment index | 0.25^{***} | 0.21^{***} | 0.21^{***} | 0.23^{***} | 0.19^{***} | 0.19^{***} | 0.00 | |
| | (0.07) | (0.08) | (0.07) | (0.07) | (0.06) | (0.06) | (1.00) | |
| | $[0.00]^{***}$ | $[0.01]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | 1082 | |
| Welfare index | 0.15^{**} | 0.14^{**} | 0.15^{**} | 0.17^{**} | 0.11^{*} | 0.10 | -0.00 | |
| | (0.07) | (0.07) | (0.07) | (0.07) | (0.06) | (0.06) | (1.00) | |
| | $[0.03]^{**}$ | $[0.04]^{**}$ | $[0.04]^{**}$ | $[0.01]^{**}$ | $[0.08]^*$ | [0.12] | 1090 | |
| Aspiration index | 0.12^{**} | 0.15^{**} | 0.13^{**} | 0.16^{***} | 0.09 | 0.12^{**} | 0.02 | |
| | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (1.00) | |
| | $[0.04]^{**}$ | $[0.01]^{**}$ | $[0.03]^{**}$ | $[0.01]^{***}$ | [0.13] | $[0.04]^{**}$ | 1904 | |
| Expectations index | 0.21^{***} | 0.22^{***} | 0.22^{***} | 0.23^{***} | 0.18^{***} | 0.18^{***} | 0.01 | |
| | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (1.00) | |
| | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | 1900 | |
| Aspirations and expectations aggregate index | 0.19^{***} | 0.22^{***} | 0.20^{***} | 0.23^{***} | 0.16^{***} | 0.19^{***} | 0.01 | |
| | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (1.00) | |
| | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.01]^{***}$ | $[0.00]^{***}$ | 1904 | |
| Omnibus index | 0.26^{***} | 0.24^{***} | 0.30*** | 0.32^{***} | 0.20*** | 0.21^{***} | -0.00 | |
| | (0.06) | (0.07) | (0.07) | (0.07) | (0.06) | (0.06) | (1.00) | |
| | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | 1091 | |

| Table A.18: ROBUSTNESS | TESTS | FOR | SUMMARY | INDICES |
|------------------------|-------|-----|---------|---------|
|------------------------|-------|-----|---------|---------|

Notes: Estimates of within-village treatment effects five years after the intervention (columns 1-6). The comparison group comprises households from the 64 treated villages that were not invited to any screening. Columns 1-4 control for village fixed effects and characteristics of the household head (or the spouse for individual-level outcomes): age, years of education, an indicator for being single, and an indicator for being male. Columns 1-2 replicate the results in the main tables of the paper. The pre-specified specification (columns 3-4) uses the same controls as the previous two columns but does not control for the baseline value of the outcome. For the educational investment index, the specifications in columns 3-4 additional control for the number of children aged 0-15 at baseline to account for the baseline imbalance in the number of children. The PDS Lasso specification (columns 5-6) estimates treatment effects following the post-double selection Lasso method (Belloni, Chernozhukov, and Hansen, 2014)—selecting controls from those shown in Appendix Tables A.9 and A.10 and partialling out from the selection the village fixed effects and the baseline value of the outcome, when this is available, following Cilliers, Elashmawy, and McKenzie (2024). Column 7 display the control mean; standard deviation; and total number of observations. Heteroskedasticity-robust standard errors (or clustered at the household-level for individual-level outcomes) are in parentheses. Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in detail in Table 7. The unit of observation is the household, except for the aspirations and expectations indices (which are observed for both household head and their spouse). The number of observations varies slightly across rows because some respondents do no

| After five years | ANC | OVA | Pre-sp | ecified | PDS 1 | | |
|--|-----------------------------|-----------------------|----------------------------|--------------------------|---------------------|----------------------------|------------------------------------|
| v | (1) | (2) | (3) | (4) | (5) | (6) | (7) |
| | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Control mean (SD) Total obs. |
| Household head and spouse's labour effort: | | | | | | | |
| Daily minutes working | 28.01*** | 19.14^{**} | 28.30*** | 19.95*** | 26.97^{***} | 18.24^{**} | 326.16 |
| | (7.80) | (7.95) | (7.67) | (7.69) | (7.42) | (7.51) | (200.01) |
| Deile minuter in leinme | [0.00]*** | [0.03]** | $[0.00]^{***}$ | [0.02]** | $[0.00]^{***}$ | [0.03]** | 1754 |
| Daily minutes in leisure | $-48.62^{\circ\circ\circ}$ | (10.22) | $-51.94^{\circ\circ\circ}$ | -20.22° | -51.10^{-11} | -20.02° | 830.59 |
| | (9.90 <i>)</i> [0.00]*** | (10.32) [0.07]* | (9.01) [0.00]*** | (9.91) [0.04]** | (9.39) [0.00]*** | (9.03 <i>)</i> [0.04]** | (107.30) 1773 |
| Agricultural investment: | [0.00] | [0.01] | [0.00] | [0.01] | [0.00] | [0.01] | 1110 |
| % with any spending on modern crop inputs | | | 0.10*** | 0.06* | 0.09*** | 0.04 | 0.58 |
| | | | (0.03) | (0.03) | (0.03) | (0.03) | (0.49) |
| | | | [0.01]** | [0.29] | [0.03]** | [0.57] | 1089 |
| Spending on seed or fertiliser (USD) PPP | | | 7.33** | 3.53 | 5.92** | 1.75 | 33.49 |
| | | | (3.07) | (3.31) | (2.90) | (3.15) | (43.54) |
| | | | $[0.05]^{**}$ | [0.38] | $[0.08]^*$ | [0.66] | 1078 |
| % with any spending on feed or vet supplies | | | 0.10*** | 0.14*** | 0.09*** | 0.12*** | 0.45 |
| | | | (0.03) | (0.03) | (0.03) | (0.03) | (0.50) |
| Constitution of the day and the second secon | | | $[0.01]^{**}$ | [0.00]*** | [0.03]** | $[0.00]^{***}$ | 1089 |
| Spending on feed of vet supplies (USD) PPP | | | 2.08 | 4.32 | 2.09 | (4.26) | 29.30 |
| | | | (4.61) [0.66] | (4.03 <i>)</i> [0.38] | (4.51) [0.86] | (4.20) [0.66] | (70.92) 1081 |
| % with any spending on hired crop labour | | | -0.05** | -0.03 | -0.05** | -0.03 | 0.36 |
| 70 while daily specific ing on miled crop labour | | | (0.02) | (0.02) | (0.02) | (0.02) | (0.48) |
| | | | [0.05]** | [0.38] | [0.05]** | [0.57] | 1089 |
| Spending on hired crop labour (USD) PPP | | | -1.30 | 3.67 | -0.86 | 3.59 | 54.16 |
| | | | (5.45) | (5.42) | (5.27) | (5.26) | (93.01) |
| | | | [0.81] | [0.50] | [0.90] | [0.66] | 1078 |
| Value of family crop labour (USD) PPP | | | 33.33* | 32.06 | 30.36 | 19.94 | 387.81 |
| | | | (19.73) | (20.08) | (18.73) | (19.13) | (258.03) |
| | | | [0.15] | [0.29] | [0.17] | [0.59] | 1079 |
| Area cultivated (hectares) | | | (0.01) | (0.02) | -0.00 | (0.00) | 0.55 |
| | | | (0.02) | (0.02) | (0.02) | (0.02) | (0.30) 1071 |
| Assets: | | | [0.05] | $\left[0.30\right]$ | [0.90] | [0.82] | 1071 |
| Value of livestock (USD) PPP | 112.75 | 89.51 | 187.38 | 310,16** | 40.53 | 71.70 | 2018.22 |
| | (122.97) | (114.28) | (135.89) | (130.45) | (116.16) | (106.57) | (1921.09) |
| | [0.36] | [0.51] | [0.17] | [0.04]** | [0.73] | [0.59] | 1053 |
| Value of tools (USD) PPP | 21.27** | 8.50 | 27.55** | 15.37 | 23.73** | 6.91 | 106.02 |
| · · · | (10.77) | (12.87) | (11.61) | (13.66) | (11.11) | (12.89) | (126.90) |
| | $[0.10]^*$ | [0.51] | $[0.04]^{**}$ | [0.26] | $[0.06]^*$ | [0.59] | 1049 |

Table A.19: ROBUSTNESS TESTS FOR AGRICULTURAL INVESTMENTS

Notes: Estimates of within-village treatment effects five years after the intervention (columns 1-6). The comparison group comprises households from the 64 treated villages that were not invited to any screening. Columns 1-4 control for village fixed-effects and characteristics of the household head (or the spouse for individual-level outcomes): age, years of education, an indicator for being single, and an indicator for being male. Columns 1-2 replicate the results in the main tables of the paper. The pre-specified specification (columns 3-4) uses the same controls as the previous two columns but does not control for the baseline value of the outcome. The PDS Lasso specification (columns 5-6) estimates treatment effects following the post-double selection Lasso method (Belloni, Chernozhukov, and Hansen, 2014)—selecting controls from those shown in Appendix Tables A.9 and A.10 and partialling out from the selection the village fixed effects and the baseline value of the outcome, when this is available, following Cilliers, Elashmawy, and McKenzie (2024). Column 7 display the control mean; standard deviation; and total number of observations. The agricultural investment panel is reported only in columns 3-4 because the baseline value of the outcome was not collected. Heteroskedasticity-robust standard errors (or clustered at the household-level for individual-level outcomes) are in parentheses. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in detail in Table 3. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1. The unit of observation is the household, except for the daily minutes working or in leisure (which are are obs

| After five years | ANCOVA F | | Pre-sp | ecified | PDS | Lasso | | |
|---|----------------|-----------------------|----------------|-----------------------|----------------|-----------------------|------------------------------------|--|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) | |
| | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Control mean (SD) Total obs. | |
| Cohort 1: Children of post-primary school-going age at endline (aged 11–15 at the time of the intervention) | | | | | | | | |
| Children aged 16-20 in school | 0.06* | 0.06 | 0.06* | 0.06* | 0.03 | 0.03 | 0.17 | |
| | (0.03) | (0.04) | (0.03) | (0.04) | (0.03) | (0.03) | (0.41) | |
| | $[0.08]^*$ | [0.13] | $[0.08]^*$ | [0.11] | [0.39] | [0.33] | 1078 | |
| Daily minutes in school for children aged 16-20 | 33.08^{**} | 30.69^{**} | 30.50^{**} | 30.00** | 19.63^{*} | 22.10^{*} | 58.64 | |
| | (13.48) | (13.82) | (12.92) | (13.27) | (11.72) | (12.32) | (149.88) | |
| | $[0.03]^{**}$ | $[0.05]^*$ | $[0.04]^{**}$ | $[0.05]^{**}$ | [0.18] | [0.14] | 1070 | |
| Daily minutes studying for children aged 16-20 | 9.25** | 7.56 | 7.86* | 7.27 | 6.26 | 5.09 | 17.82 | |
| | (4.68) | (5.05) | (4.52) | (4.90) | (4.22) | (4.73) | (52.12) | |
| | $[0.06]^*$ | [0.13] | $[0.08]^*$ | [0.14] | [0.18] | [0.33] | 1063 | |
| Children aged 16-20 that attained 8th grade | 0.09*** | 0.07** | 0.08*** | 0.07** | 0.07*** | 0.07*** | 0.07 | |
| | (0.03) | (0.03) | (0.03) | (0.03) | (0.02) | (0.03) | (0.26) | |
| | $[0.00]^{***}$ | [0.04]** | [0.01]*** | [0.05]** | [0.01]*** | $[0.03]^{**}$ | 1078 | |
| Cohort 2: Children of primary school-going age | at endline | (aged 2–10 | at the time | of the inte | rvention) | | | |
| Children aged 7-15 in school | 0.03 | 0.05 | 0.01 | 0.08 | 0.01 | 0.06 | 1.22 | |
| | (0.08) | (0.08) | (0.07) | (0.07) | (0.07) | (0.06) | (1.18) | |
| | [0.67] | [0.52] | [0.86] | [0.26] | [0.83] | [0.37] | 1078 | |
| Daily minutes in school for children aged 7-15 | 23.47 | 37.11 | 11.47 | 45.84^{*} | 22.73 | 45.67^{*} | 527.12 | |
| | (30.16) | (30.22) | (25.84) | (25.31) | (24.98) | (23.91) | (437.21) | |
| | [0.65] | [0.52] | [0.86] | [0.21] | [0.54] | [0.17] | 1060 | |
| Daily minutes studying for children aged 7-15 | 17.34^{*} | 7.56 | 15.13^{*} | 9.59 | 16.53^{**} | 7.29 | 91.29 | |
| | (8.91) | (9.14) | (8.36) | (8.57) | (7.90) | (8.14) | (115.61) | |
| | [0.16] | [0.52] | [0.21] | [0.26] | [0.11] | [0.37] | 1061 | |
| For all children | | | | | | | | |
| Schooling expenditure (USD) PPP | 6.97^{**} | 5.81^{**} | 8.20*** | 6.88^{**} | 7.70*** | 6.23** | 19.17 | |
| | (2.84) | (2.95) | (2.86) | (3.06) | (2.80) | (2.90) | (32.73) | |
| | $[0.01]^{**}$ | $[0.05]^{**}$ | $[0.00]^{***}$ | $[0.02]^{**}$ | $[0.00]^{***}$ | $[0.03]^{**}$ | 1065 | |

Table A.20: ROBUSTNESS TESTS FOR EDUCATIONAL INVESTMENTS

Notes: Estimates of within-village treatment effects five years after the intervention (columns 1-6). The comparison group comprises households from the 64 treated villages that were not invited to any screening. Columns 1-4 control for village fixed-effects and characteristics of the household head: age, years of education, an indicator for being single, and an indicator for being male. Columns 1-2 replicate the results in the main tables of the paper. The pre-specified specification (columns 3-4) uses the same controls as the previous two columns but does not control for the baseline value of the outcome. Specifications in columns 3-4 additional control for the number of children aged 0-15 at baseline to account for the baseline imbalance in the number of children. The PDS Lasso specification (columns 5-6) estimates treatment effects following the post-double selection Lasso method (Belloni, Chernozhukov, and Hansen, 2014)—selecting controls from those shown in Appendix Tables A.9 and A.10 and partialling out from the selection the village fixed effects and the baseline value of the outcome, when this is available, following Cilliers, Elashmawy, and McKenzie (2024). Column 7 display the control mean; standard deviation; and total number of observations. Heteroskedasticity-robust standard errors are in parentheses. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in detail in Table 4. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1. The unit of observation is the household. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous va

| After five years | ANC | OVA | Pre-sp | ecified | PDS | Lasso | |
|---|--|---|---|---|---|--|---|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) |
| | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Control mean (SD) Total obs. |
| Food security: | | | | | | | |
| Food security index: z-score | -0.07 | 0.03 | -0.06 | 0.04 | -0.05 | 0.03 | 0.48 |
| Months of food insecurity | (0.06) [0.28] -0.33^{**} (0.15) | $(0.06) \\ [0.61] \\ -0.38^{***} \\ (0.14)$ | $(0.06) \\ [0.31] \\ -0.32^{**} \\ (0.14)$ | $(0.06) \\ [0.54] \\ -0.35^{**} \\ (0.14)$ | (0.06) [0.37] -0.25^{*} (0.14) | $(0.06) \\ [0.62] \\ -0.28^{**} \\ (0.13)$ | $(0.92) \\ 1076 \\ 2.71 \\ (2.13)$ |
| | $[0.05]^*$ | $[0.02]^{**}$ | $[0.05]^*$ | $[0.03]^{**}$ | [0.14] | $[0.08]^*$ | 1038 |
| Consumption: | | | | | | | |
| Food consumption (USD) per ad. equiv. monthly PPP Frequent non-food (1m recall USD) per ad. equiv. PPP | | | -1.98 (2.05) [0.33] 0.44 (0.28) | $\begin{array}{c} 0.32 \\ (2.07) \\ [0.88] \\ 0.40 \\ (0.30) \end{array}$ | -1.90 (1.96) [0.33] 0.35 (0.26) | $\begin{array}{c} 0.30 \\ (1.98) \\ [0.88] \\ 0.32 \\ (0.28) \end{array}$ | $53.91 \\ (29.98) \\ 1076 \\ 4.08 \\ (3.69)$ |
| Nonfood consumption (12m recall USD) per ad. equiv. monthly PPP | | | $ \begin{bmatrix} 0.120 \\ 0.19 \end{bmatrix} \\ 0.70 \\ (0.51) \\ \begin{bmatrix} 0.21 \end{bmatrix} $ | [0.25] 1.24^{**} (0.48) $[0.05]^{*}$ | [0.23] 0.65 (0.48) [0.22] | [0.46] 0.98^{**} (0.45) [0.14] | 1076 7.47 (6.35) 1070 |
| Consumption of cigarettes and to bacco (USD) \ensuremath{PPP} (USD) \ensuremath{PPP} | | | (0.21) (0.20) (0.13) | $(0.05)^{\circ}$ $(0.17)^{\circ}$ | [0.23] 0.17 (0.12) | $\begin{bmatrix} 0.14 \\ 0.14 \\ (0.13) \end{bmatrix}$ | |
| General economic position (scale 1 to 4) | | | $[0.19] \\ 0.09^* \\ (0.05) \\ [0.10]$ | $[0.25] \\ 0.09^* \\ (0.05) \\ [0.21]$ | $[0.23] \\ 0.07 \\ (0.05) \\ [0.23]$ | $[0.46] \\ 0.04 \\ (0.05) \\ [0.46]$ | 1078 2.10 (0.73) 1088 |
| Non-productive durables and housing: | | | [0.15] | [0.21] | [0.20] | [0.40] | 1000 |
| Value of durable assets excluding tools (USD) PPP | 20.71** | 22.36** | 21.95** | 24.88** | 17.81* | 15.17 | 70.55 |
| Value of house (USD) PPP | (9.97) $[0.08]^*$ 384.94^{***} (90.70) | $(10.17) \\ [0.06]^* \\ 301.22^{***} \\ (89.10)$ | (10.75) $[0.05]^*$ 415.56^{***} (93.66) | $(11.18) \\ [0.05]^* \\ 352.73^{***} \\ (93.44)$ | (9.42) $[0.08]^*$ 312.29^{***} (78.81) | $(9.59) \\ [0.23] \\ 201.79^{**} \\ (80.55)$ | (127.39) 1049 1384.27 (1235.57) |
| Non-organic roof | $ \begin{array}{c} [0.00]^{***} \\ 0.01 \\ (0.03) \\ [0.50] \end{array} $ | $[0.00]^{***}$ -0.01 (0.03) [0.64] | $[0.00]^{***}$ 0.06^{**} (0.03) $[0.05]^{*}$ | $[0.00]^{***}$ 0.02 (0.03) [0.40] | $[0.00]^{***}$ 0.02 (0.02) [0.22] | $[0.05]^{**}$ -0.02 (0.02) [0.48] | $ 1020 \\ 0.68 \\ (0.47) \\ 1026 $ |
| Own toilet facility | [0.39] 0.07^{*} (0.04) $[0.08]^{*}$ | [0.04] (0.02) (0.04) [0.64] | [0.03] 0.07^{*} (0.03) $[0.05]^{*}$ | [0.49] 0.02 (0.03) [0.49] | [0.32] 0.07^{**} (0.03) $[0.08]^{*}$ | [0.48] 0.01 (0.03) [0.66] | $ \begin{array}{c} 1030 \\ 0.38 \\ (0.49) \\ 1039 \end{array} $ |
| Wellbeing: | L J | ι , | | ι , | | ι , | |
| Best life Happiest life | $\begin{array}{c} 0.22^{**} \\ (0.11) \\ [0.09]^{*} \\ 0.12 \\ (0.14) \end{array}$ | $\begin{array}{c} 0.12 \\ (0.11) \\ [0.58] \\ 0.00 \\ (0.14) \end{array}$ | $\begin{array}{c} 0.23^{**} \\ (0.11) \\ [0.09]^{*} \\ 0.11 \\ (0.14) \end{array}$ | $\begin{array}{c} 0.17\\(0.12)\\[0.28]\\-0.01\\(0.14)\end{array}$ | $\begin{array}{c} 0.12 \\ (0.10) \\ [0.47] \\ 0.02 \\ (0.13) \end{array}$ | $\begin{array}{c} 0.05 \\ (0.11) \\ [0.65] \\ -0.14 \\ (0.13) \end{array}$ | $ \begin{array}{r} 4.83 \\ (1.80) \\ 1901 \\ 6.05 \\ (2.10) \end{array} $ |
| | [0.38] | [0.98] | [0.14] | [0.95] | [0.15] | [0.56] | 1885 |

Table A.21: ROBUSTNESS TESTS FOR WELFARE OUTCOMES

Notes: Estimates of within-village treatment effects five years after the intervention (columns 1-6). The comparison group comprises households from the 64 treated villages that were not invited to any screening. Columns 1-4 control for village fixed-effects and characteristics of the household head (or the spouse for individual-level outcomes): age, years of education, an indicator for being single, and an indicator for being male. Columns 1-2 replicate the results in the main tables of the paper. The pre-specified specification (columns 3-4) uses the same controls as the previous two columns but does not control for the baseline value of the outcome. The PDS Lasso specification (columns 5-6) estimates treatment effects following the post-double selection Lasso method (Belloni, Chernozhukov, and Hansen, 2014)—selecting controls from those shown in Appendix Tables A.9 and A.10 and partialling out from the selection the village fixed effects and the baseline value of the outcome, when this is available, following Cilliers, Elashmawy, and McKenzie (2024). Column 7 display the control mean; standard deviation; and total number of observations. The consumption panel is reported only in columns 3-4 because the baseline value of the outcome was not collected. Heteroskedasticity-robust standard errors (or clustered at the household-level for individual-level outcomes) are in parentheses. Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in detail in Table 5. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian bir) PPP. The conversion is described in Appendix E.1. The unit of observation is the household, except for the variables in the wellbeing panel (which are observed for both house

| After five years ANCOVA | | OVA | Pre-sp | ecified | PDS | | |
|--|-------------------------|-------------------------|-------------------------|--------------------------|---------------------|-------------------------|------------------------------------|
| | (1) | (2) | (3) | (4) | (5) | (6) | (7) |
| Aspirations: What would you like to achieve? | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Treatment | Treat. vs. placebo | Control mean (SD) Total obs. |
| Aspirations index | 0.12** | 0.15** | 0.13** | 0.16*** | 0.09 | 0.12** | 0.02 |
| 1 | (0.06) $[0.04]^{**}$ | (0.06) $[0.01]^{**}$ | (0.06) $[0.03]^{**}$ | (0.06) $[0.01]^{***}$ | (0.06) [0.10] | (0.06) $[0.03]^{**}$ | (1.00) 1904 |
| Aspired income (USD) PPP | 2505.32 | 3148.93* | 2436.44 | 3125.51^{*} | 2216.47 | 2934.92* | 15446.71 |
| | (1773.81) [0.22] | (1701.71) $[0.10]^*$ | (1745.50) [0.23] | (1697.89) $[0.10]^*$ | (1710.26) [0.29] | (1664.24) [0.12] | (27746.43) 1864 |
| Aspired wealth (USD) PPP | 1584.05 | 664.91 | 1550.68 | 713.73 | 989.20 | 210.47 | 11978.72 |
| | (1301.17) | (1395.59) | (1298.04) | (1392.27) | (1260.44) | (1360.61) | (21518.47) |
| Agained adjustion (mana) | [0.22] | [0.63] | [0.23] | [0.61] | [0.43] | [0.88] | 1867 |
| Aspired education (years) | (0.29) | (0.39) | (0.31) | (0.45) | (0.23) | (0.37) | (2.60) |
| | [0.19] | [0.05]** | [0.16] | $[0.02]^{**}$ | [0.29] | $[0.07]^*$ | 1769 |
| Expectations: What do you expect in ten years? | | LJ | LJ | | | ĽJ | |
| Expectations index | 0.21*** | 0.22*** | 0.22*** | 0.23*** | 0.18*** | 0.18*** | 0.01 |
| | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (1.00) |
| | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.01]^{***}$ | $[0.00]^{***}$ | 1900 |
| Expected income (USD) PPP | 309.11 | 238.22 | 306.52 | 199.29 | 149.49 | -12.63 | 3409.97 |
| | (190.28) | (189.95) | (188.01) | (192.24) | (175.78) | (180.79) | (2820.92) |
| | [0.10] | [0.21] | [0.10] | [0.30] | [0.40] | [0.94] | 1861 |
| Expected wealth (USD) PPP | 596.06** | 369.61 | 612.21** | 356.06 | 383.95* | 44.65 | 4009.69 |
| | (250.19) | (247.91) | (249.03) | (250.75) | (230.65) | (237.27) | (3997.53) |
| | $[0.04]^{**}$ | [0.20] | $[0.02]^{**}$ | [0.23] | [0.14] | [0.94] | 1853 |
| Expected education (years) | 0.58** | 0.90*** | 0.67*** | 1.02*** | 0.69*** | 1.04*** | 12.31 |
| | (0.26) | (0.27) | (0.26) | (0.27) | (0.26) | (0.26) | (3.88) |
| | $[0.04]^{**}$ | [0.00]*** | [0.02]** | [0.00]*** | [0.02]** | [0.00]*** | 1706 |
| Aspirations and expectations aggregate index | 0.19^{***} | 0.22^{***} | 0.20^{***} | 0.23^{***} | 0.16^{***} | 0.19^{***} | 0.01 |
| | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (0.06) | (1.00) |
| | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.00]^{***}$ | $[0.01]^{***}$ | $[0.00]^{***}$ | 1904 |

Table A.22: ROBUSTNESS TESTS FOR ASPIRATIONS AND EXPECTATIONS

Notes: Estimates of within-village treatment effects five years after the intervention (columns 1-6). The comparison group comprises households from the 64 treated villages that were not invited to any screening. Columns 1-4 control for village fixed-effects and characteristics of the household head or spouse: age, years of education, an indicator for being single, and an indicator for being male. Columns 1-2 replicate the results in the main tables of the paper. The pre-specified specification (columns 3-4) uses the same controls as the previous two columns but does not control for the baseline value of the outcome. The PDS Lasso specification (columns 5-6) estimates treatment effects following the post-double selection Lasso method (Belloni, Chernozhukov, and Hansen, 2014)—selecting controls from those shown in Appendix Tables A.9 and A.10 and partialling out from the selection the village fixed effects and the baseline value of the outcome, when this is available, following Cilliers, Elashmawy, and McKenzie (2024). Column 7 display the control mean; standard deviation; and total number of observations. Heteroskedasticity-robust standard errors are clustered at the household-level in parentheses. Stars on the coefficient estimates reflect unadjusted *p*-values. Minimum *q*-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. Outcome variables are listed on the left, and described in detail in Appendix Table A.1. All monetary values are in PPP-adjusted USD, set at 2016 (endline) prices and deflated using national non-food CPI. In 2016, USD 1 = 8.67 ETB (Ethiopian birr) PPP. The conversion is described in Appendix E.1. The unit of observation is the individual. The number of observations varies slightly across rows because some respondents do not answer all questions and because we trim continuous variables above the 99th percentile.

I Heterogeneity

This section presents the figure referenced in Section 5.3 of the main paper that show a lack of heterogeneous treatment effects.

| Outcomes: | _ | | | | | | | | | |
|------------------------------|---------------------|------------|----------------------|-------------------|--------------|-------------------|-------------|----------|---------------------|---------------|
| Agricultural Invest Index | .5- t. 0- 5- | •*** | | | | — | | | | |
| Educational Inves Index | .5- t. 0- 5- | •*** | | ¢ | <u> </u> | | ¢ | | ¢ | |
| Welfare Index | - 5. - 0 - 5 | •** | | | ¢ | — | ¢ | ¢ | ¢ | ¢ |
| Aspirations Index | - 5. - 0 - 5 | •** | ¢ | ¢ | | | | ¢ | | |
| Expectations Index | -5. -0 5- | •*** | ¢ | ¢ | | — | | ¢* | ¢ | ¢* |
| Asp. and Exp. Ag Index | -5. -g. 0- 5- | *** | ¢ | | | | | ¢ | ¢ | ф |
| Omnibus Index | - 5. - 0 - 5 | ¢*** | | | \$ | | | ¢ | ¢ | - þ |
| Dimension heterogene | <u>s of</u> eitv | Treatment | Female (Household | Years | Expectations | Internal Locus | Aspirations | Age | Aggregate Assets | Treated peers |
| at baseline | <u>.</u> | (0.33) | Head) (0.103) | Education (1.938) | (0.115) | (15.682) | (0.157) | (40.457) | (393.814) | (0.869) |

Figure A.4: Heterogeneous treatment effects on summary indices

Notes: Each coefficient represents a separate OLS regression using data from five years after the intervention. The outcomes are the summary indices in Table 7. The first column shows within-village intention-to-treat effects, controlling for a placebo-group indicator. The comparison group is households from the 64 treated villages not invited to any screening. In the second column, we control for gender. For individual-level indices (aspirations, expectations, and the combined aspirations-expectations index), we use an indicator for female respondents and its interaction with the treatment. For household-level indices, we use an indicator for female-headed households and its interaction with the treatment. Subsequent columns control for baseline values above the median and their interaction with the treatment indicator. For household-level outcomes (agricultural investment, educational investment, welfare, and omnibus), the dimension of baseline heterogeneity (on the x-axis) is reported by the household head. For individual-level outcomes (aspirations, expectations, and combined indices), the dimension of baseline heterogeneity is reported by the individual (spouse or household head). For aggregate assets, only household-level data are used to test differences based on whether household assets are above or below the median. The internal locus of control measure is detailed in Appendix Section E.3.3. Aggregate assets include non-productive assets, productive assets, savings, and livestock. "Treated peers" are close social connections invited to watch role model videos. Standard errors are robust to heteroskedasticity and clustered at the household level for the aspirations/expectations indices. Bars represent 95 per cent confidence intervals based on naive p-values. For heterogeneous effects, we use FDR-adjusted q-values for multiple testing corrections on interaction terms across outcomes, based on (Benjamini, Krieger, and Yekutieli, 2006). Stars indicate significance according to the minimum q-value for each hypothesis. We do not include FDR-adjusted q-values for multiple interactions across different dimensions, as we did not have the power for such tests.

J Spillovers

Our main results use an individual-level randomisation in 64 villages. The control group live in villages where others receive either treatment or placebo. This subsection describes an alternative specification which tests for spillovers. We find little evidence of spillovers.

Spillover analysis specification: We compare households in 64 treated villages with households in 10 pure control villages, selected as described in Section 5.5:

(4)
$$y_{iv} = \delta^s T_{iv} + \rho^s P_{iv} + \varphi C_{iv} + X'_{iv0} \pi^s + \alpha^s_v + \varepsilon_{iv}$$

 X_{iv0} are the same pre-specified controls as the main specification, Equation (1), which are likely time-invariant and allow us to include the values collected in the pure control group at the five-year follow-up. We also add a list of pre-specified village-level controls. α_v^s denotes screening-site fixed effects, which replace village-level fixed effects in Equation (1). C_{iv} is an indicator equal to one for within-village control households in treatment villages who were not invited to watch the documentary or the TV show. φ is the difference between the within-village control and the pure control group, which measures the extent to which control households in treatment villages were (indirectly) affected by the treatment. The superscript "s" is added to the parameters δ and ρ to distinguish them from the previous within-treatment village estimates. In this specification, these households are compared to the base group of households in pure control villages. Standard errors are clustered at the village level.

Lack of spillover effects: We find little systematic evidence of spillovers. There are significant differences between the within-village control group and pure control group on only one of five indices (Column 3, Table A.23). If anything, we find some evidence of positive spillovers for the welfare index, which could potentially attenuate our main estimates. Our estimates of treatment effects are very close to those in our main specifications although the control group is different (Columns 1, 2 and 4, Table A.23).

Balance tests across villages: Table A.24 reports balance tests on pre-specified villagelevel variables comparing the 64 treated and 10 untreated villages.⁶⁵ There are some differences between treatment and pure control villages, potentially because the status of treatment villages is not randomly allocated. While an omnibus tests does not reject the null hypothesis that treated and pure control villages are the same, the number of villages is small so our tests for balance are not strong. However, differences do not seem to reflect a systematic pattern, for example that the treated villages are more prosperous. Pure control villages have less irrigated land and forests, are more likely to grow sorghum as the main crop, have a smaller share of

^{65.} We exclude pre-specified variables with insufficient variation, such as post-primary school presence, electricity access (absent in all but one village), radio signal availability (nearly universal), the share of non-Orthodox Christians (0 in all but one village), and television ownership rates. We also included the five distance variables, which we had pre-specified as village-level controls, calculated from village coordinates rather than village-level surveys.

non-agricultural business entrepreneurs, and are further away from a river.

| After five years | (1) | (2) | (3) | (4) | (5) | (6) | (7) Puro |
|-------------------------------|----------------|------------|------------|-----------------------|-----------------------|-------------------------|---------------------------------------|
| | Treatment | Placebo | Control | Treat. vs. placebo | Treat. vs. control | Placebo. vs. control | Control mean (SD) Total obs. |
| Agricultural investment index | 0.25*** | 0.03 | 0.01 | 0.22*** | 0.24*** | 0.02 | 0.00 |
| <u> </u> | (0.09) | (0.09) | (0.09) | (0.06) | (0.06) | (0.05) | (1.00) |
| | [0.01]*** | [0.84] | [1.00] | [0.00]*** | [0.00]*** | [0.84] | 1223 |
| Educational investment index | 0.20*** | -0.02 | -0.00 | 0.22*** | 0.20*** | -0.02 | 0.00 |
| | (0.07) | (0.07) | (0.06) | (0.07) | (0.06) | (0.06) | (1.00) |
| | $[0.01]^{***}$ | [0.84] | [1.00] | [0.00]*** | [0.00]*** | [0.84] | 1219 |
| Welfare index | 0.42*** | 0.26*** | 0.25*** | 0.16** | 0.17^{**} | 0.01 | -0.00 |
| | (0.09) | (0.10) | (0.09) | (0.08) | (0.08) | (0.08) | (1.00) |
| | $[0.00]^{***}$ | $[0.06]^*$ | $[0.05]^*$ | $[0.05]^{**}$ | $[0.04]^{**}$ | [0.91] | 1224 |
| Aspirations index | 0.22^{***} | 0.06 | 0.09 | 0.17^{***} | 0.13^{**} | -0.04 | 0.00 |
| | (0.08) | (0.08) | (0.07) | (0.05) | (0.06) | (0.06) | (1.00) |
| | $[0.01]^{***}$ | [0.84] | [0.51] | $[0.00]^{***}$ | $[0.04]^{**}$ | [0.84] | 2231 |
| Expectations index | 0.27^{***} | 0.03 | 0.06 | 0.24^{***} | 0.21^{***} | -0.03 | -0.00 |
| | (0.09) | (0.08) | (0.08) | (0.06) | (0.07) | (0.06) | (1.00) |
| | $[0.00]^{***}$ | [0.84] | [0.63] | $[0.00]^{***}$ | $[0.01]^{***}$ | [0.84] | 2230 |
| Asp. and exp. aggregate index | 0.25^{***} | 0.02 | 0.07 | 0.23^{***} | 0.18^{***} | -0.05 | -0.00 |
| | (0.08) | (0.08) | (0.08) | (0.05) | (0.06) | (0.06) | (1.00) |
| | $[0.01]^{***}$ | [0.84] | [0.63] | $[0.00]^{***}$ | $[0.01]^{***}$ | [0.84] | 2231 |
| Omnibus index | 0.34^{***} | 0.05 | 0.08 | 0.29^{***} | 0.25^{***} | -0.03 | -0.00 |
| | (0.07) | (0.06) | (0.06) | (0.06) | (0.06) | (0.05) | (1.00) |
| | $[0.00]^{***}$ | [0.84] | [0.51] | $[0.00]^{***}$ | $[0.00]^{***}$ | [0.84] | 1219 |

| Table A.23: | SUMMARY | INDICES | IN | SPILLOVER | ANALYSIS |
|-------------|---------|---------|----|-----------|----------|
|-------------|---------|---------|----|-----------|----------|

Notes: OLS estimates of between-village effects five years after the intervention (columns 1, 2 and 3). Column 4 tests for differences in parameters obtained in first two columns. Column 5 tests for differences in parameters obtained in first and third columns. Column 6 tests for differences in parameters obtained in second and third columns. The comparison group comprises households from the ten pure-control villages that were first surveyed five years after the intervention. Column 7 displays the mean, standard deviation for the pure-control group, and total number of observations. All regressions control for screening-site fixed effects, individual characteristics of the respondent (age, years of education, an indicator for being single, and an indicator for being male) and village-level controls (the number of inhabitants, hectares covered by forest, an indicator for whether sorghum is the main crop, costs of trip to nearest market, an indicator for whether the village has a first cycle school, percentage of households with radio, distance to the next market place, distance to the school, distance to the next farmers training centre, distance to the next health centre, distance to the next river). Regressions on the educational investment index and omnibus index additionally control for the number of children aged 0-20 currently in the household to account for the baseline imbalance in the number of children. Heteroskedasticity-robust standard errors are clustered at the village-level and are in parentheses. Stars on the coefficient estimates reflect unadjusted p-values. Minimum q-values are in square brackets and are calculated over each panel of variables. * denotes significance at 10 pct., ** at 5 pct., and *** at 1 pct. level. The unit of observation is the household, except for the aspirations and expectations indices (which are are observed for both household head and their spouse). The number of observations varies slightly across rows because some respondents do not answer all questions, though the indices aggregate all non-missing outcomes. The outcomes are inverse-covariance-weighted averages standardised relative to the pure-control group, following Anderson (2008). The agricultural investment index includes all outcomes reported in Table 3, with daily minutes in leisure being recoded to be negative. The educational investment index includes all outcomes reported in Table 4. The welfare index includes all outcomes reported in Table 5, with months of food insecurity in the last year recoded to be negative. The welfare index averages over the household head's subjective well-being outcomes. The aspirations and expectations aggregate index is made of the reported income, wealth and years of education for children, for aspirations and expectations. The omnibus index aggregates the agricultural investment, educational investment, welfare, and aspirations and expectations aggregate standardised indices into a single index, followingKling, Liebman, and Katz (2007). As the omnibus index is for the whole household, we use the household head's aspirations and expectations aggregate index.

| | Sam | ple Mean | ıs |
|--|-----------|----------|---------|
| Variable | Treatment | Control | p-value |
| Number of inhabitants | 467 | 374 | 0.04 |
| Number of households | 79.4 | 66.4 | 0.09 |
| Majority is Oromo $(=1)$ | 0.934 | 0.800 | 0.31 |
| % inhabitants belonging to the main ethnicity | 93.9 | 94.7 | 0.79 |
| % inhabitants that are Muslim | 86.4 | 85.7 | 0.93 |
| % inhabitants that are Orthodox | 13.6 | 14.2 | 0.94 |
| Agricultural land (hectares) | 39.3 | 32.5 | 0.18 |
| Cultivated land (hectares) | 31.3 | 30.1 | 0.80 |
| Irrigated land (hectares) | 3.19 | 0.10 | 0.00 |
| Grazing land (hectares) | 0.540 | 0.400 | 0.70 |
| Forest (hectares) | 6.34 | 2.10 | 0.01 |
| Most important crop is sorghum | 0.918 | 1.000 | 0.02 |
| Most important livestock is oxen | 0.492 | 0.300 | 0.24 |
| Inhabitants' main source of income is subsistence farming $(=1)$ | 0.639 | 0.700 | 0.71 |
| Inhabitants' main source of income is cash crops $(=1)$ | 0.230 | 0.100 | 0.25 |
| % workers that are farmers | 95.1 | 97.0 | 0.32 |
| % workers that are farm labourers | 2.36 | 5.64 | 0.38 |
| % workers that are non-agricultural business owners | 2.14 | 0.06 | 0.00 |
| % workers that are non-agricultural business workers | 1.26 | 1.67 | 0.73 |
| Cost of trip to nearest market (USD) PPP | 4.33 | 4.04 | 0.61 |
| Walking time to the nearest market (in minutes) | 97.1 | 92.0 | 0.77 |
| Village has first cycle school $(=1)$ | 0.262 | 0.200 | 0.66 |
| Village has second cycle school $(=1)$ | 0.180 | 0.200 | 0.89 |
| Village has TV transmission $(=1)$ | 0.246 | 0.200 | 0.74 |
| Village has mobile network $(=1)$ | 0.934 | 0.700 | 0.12 |
| % inhabitants with a radio | 36.3 | 31.6 | 0.57 |
| % inhabitants with a mobile phone | 0.324 | 0.245 | 0.10 |
| Main source of drinking water is pond/river/stream $(=1)$ | 0.033 | 0.100 | 0.50 |
| Most important transportation mode is mules/donkey/horse $(=1)$ | 0.803 | 0.800 | 0.98 |
| Distance to next school (in kilometers) | 1.28 | 0.93 | 0.08 |
| Distance to next farmer training centre (in kilometers) | 3.56 | 3.74 | 0.79 |
| Distance to the nearest market (in kilometers) | 9.7 | 8.9 | 0.61 |
| Distance to next health centre (in kilometers) | 10.2 | 9.7 | 0.76 |
| Distance to next river (in kilometers) | 2.37 | 3.76 | 0.00 |
| Omnibus F-test | | | 0.19 |

Table A.24: VILLAGE-LEVEL BALANCE (64 TREATED VS. 10 CONTROL VILLAGES)

Notes: Table shows means of pre-specified village characteristics collected from a village survey conducted five years after the experiment for each village group (columns 1-2). All variables are from village-level questionnaires collected five years after the experiment, except for the bottom three distance variables, which come from administrative data collected prior to the intervention (before baseline). Market centers were recorded by the IFPRI/FAO Environment and Natural Resources Service (SDRN) in 2004. Health centers and rivers information was gathered by the FAO Environment and Natural Resources Service (SDRN) in 2007. Column 3 shows p-values for equal means across groups, evaluating balance between treated and control villages. Hypothesis tests are based on heteroskedasticity-robust standard errors. Omnibus test randomised inference p-values on the bottom row, following Kerwin, Rostom, and Sterck (2024).

K Cost effectiveness analysis

In this appendix, we report on the costs of the intervention, both at the study scale and at a reasonable implementation scale, and the implied benefit-cost ratio for a subset of outcomes proxying household overall wealth.

K.1 Cost of the intervention

At the scale of the study: The experiment was designed to ensure internal validity, not to limit costs, so per head costs at the scale of the experiment are higher than if the intervention were delivered at reasonable scale. The documentaries were produced by professional documentary makers at a total cost of \$41,910.25 (USD 2016 PPP) but were only shown to 960 households. We treat these as the relevant *fixed costs* of the intervention *at the experiment's scale*. These documentaries were screened at 16 different sites across a large administrative area, to small audiences deliberately limited for the experiment, adding substantial staffing, transport and projection costs. At the scale of the intervention, we estimate these from our accounts as \$17,990.16 (USD 2016 PPP). These are the relevant *variable costs at the experiment's scale*. Half of these costs were transport costs, and approximately a quarter each staffing costs and projection equipment hire costs. Per household, this suggests a variable cost of \$18.73 (USD 2016 PPP) and a fixed cost of \$43.64 (USD 2016 PPP), yielding a total cost per household at the scale of the experiment of \$62.37 (USD 2016 PPP), shown in the first panel of Table A.25.

At reasonable scale: It is unlikely a policy intervention of this nature would be implemented at such small scale. These documentaries could in principle be shown to very large audiences, bringing these fixed cost per household down considerably. Variable costs could be reduced significantly by including larger fractions of the population in each village and hiring less skilled staff than survey enumerators (or using existing community-based workers).

We hence also show costs and implied benefit cost ratios at a "reasonable" scale, when some of the obvious scale economies are being exploited. We reference a cost-effectiveness analysis of an experiment implemented with the NGO Digital Green in Ethiopia in 2017, which uses video screenings facilitated by extension agents to disseminate agronomic content (Bernard et al., 2019). This experiment targeted 57,750 farmers, and had a variable cost per farmer of \$9.47 in USD 2016 PPP. If our intervention had reached a similar scale to this larger experiment, the fixed cost per household would drop to \$0.73 USD 2016 PPP. This yields a total cost per household at reasonable scale of \$10.20 USD 2016 PPP. Remote digital delivery could significantly reduce costs, but our experiment did not assess this delivery channel, making it a promising area for further research.

K.2 Cost-effectiveness of the intervention

We focus on robust effects related to assets expressed in monetary terms, which proxy for overall lifetime household wealth. This likely underestimates benefits as several robustly significant outcomes are not expressed in monetary terms, such as educational attainment or the reduction in food insecurity.

We show three benefit-cost ratios: considering only gains in durables; considering gains in durables, livestock and productive assets; and considering gains in those three types of assets plus housing values. On average, households gain \$20.71 USD 2016 PPP in durable assets, robust across specifications (Table 5 and A.21). Additional benefits in liquid assets, such as livestock (\$112.75 USD 2016 PPP), and productive assets (\$21.27 USD 2016 PPP) could be counted but these results are not significant in all specifications (Table 3 and A.19). Finally, there are significant benefits in terms of increased house values (\$384.94 USD 2016 PPP, (Table 5 and A.21). We only count the gains after five years in values, not the likely future flows from the assets involved.⁶⁶ Table A.25 shows benefit-cost ratios for per household costs at the experiment's scale (\$62.37 USD 2016 PPP) and at a reasonable scale of actual delivery (\$10.20 USD 2016 PPP).

Table A.25: Cost-effectiveness Ratios at Different Scales

| | (1) | (2) |
|---|------------------|------------------|
| | Experiment Scale | Reasonable Scale |
| | (N=960) | (N=57,750) |
| Costs (per household, USD PPP) | | |
| Fixed cost | 43.64 | 0.73 |
| Variable cost | 18.73 | 9.74 |
| Total cost | 62.37 | 10.20 |
| Benefit-Cost Ratios | | |
| Durable assets | 0.33 | 2.03 |
| Durable + Livestock + Productive assets | 2.48 | 15.18 |
| Durable + Livestock + Productive assets + Housing | 8.65 | 52.93 |

Notes: All costs are presented in 2016 USD PPP per household. Fixed costs at the experiment's scale include video production. Variable costs primarily consisted of transport, staffing, and projection costs. Data on costs at reasonable scale are from Bernard et al. (2019). Benefit-cost ratios are calculated using as benefits the five-year intention-to-treat effects relative to the control group, conservatively excluding any potential flow benefits in future from ownership of the asset. The first row of benefit cost ratios consider only benefits in terms of durable assets. The second row considers benefits in terms of durable assets, livestock assets and productive assets. The third considers in terms of durable assets, livestock assets, productive assets and housing.

We conservatively estimate that, run at a reasonable scale, the intervention generates about a two dollars increase in durable assets per dollar spent (Column 2 and first row of second panel, Table A.25). The relatively high gains for housing drives the benefit-cost ratio up considerably, even at the experiment's scale.

^{66.} Since our aim is to proxy for household lifetime wealth, we do not include consumption. At endline, we found a very small positive effect that was not statistically significant.

L Comparison of effect sizes with other interventions



Figure A.5: Effect size expressed as a fraction of the control group mean across studies

Notes: Treatment effects expressed as a fraction of the control group mean across comparable outcomes in recent studies of psycho-social and anti-poverty interventions. Each line shows the OLS estimate (divided by the control mean) and their corresponding 95 per cent confidence interval for that outcome. Further details on the variables for each study are in Appendix Table A.26.

| Outcome | Paper Name | Variable definition | Expression of value | Timeline |
|------------------------|--|--|---------------------|---------------|
| | Bernard et al. (2025) | Value of tools, livestock, and non-productive durable assets | Control mean $\%$ | 52 months |
| Total assets | Banerjee et al. (2015) | Total assets value | Control mean $\%$ | 36 months |
| | Orkin et al. (2023) | Total assets value | Control mean $\%$ | 17 months |
| | Bernard et al. (2025) | Value of tools and livestock | Control mean % | 52 months |
| Productive assets | Orkin et al. (2023) | Value of tools and livestock | Control mean $\%$ | 17 months |
| | Banerjee et al. (2015) | Sum of productive assets, livestock and agricultural tools | Control mean $\%$ | 36 months |
| | Bernard et al. (2025) | Durable assets, excluding tools | Control mean % | 52 months |
| Dunable acceta | Orkin et al. (2023) | Durable household assets | Control mean $\%$ | 17 months |
| Durable assets | Banerjee et al. (2015) | Household assets value | Control mean $\%$ | 36 months |
| | Blattman, Jamison, and Sheridan (2017) | Durable consumption assets | Control mean $\%$ | 12 months |
| | Bernard et al. (2025) | Respondents' daily minutes working | Control mean $\%$ | 52 months |
| | Banerjee et al. (2015) | Total minutes spent doing productive activities in last day | Control mean $\%$ | 36 months |
| Work time | Blattman, Jamison, and Sheridan (2017) | Hours per week of work in past month | Control mean $\%$ | 12 months |
| | Blattman et al. (2023) | Hours per week in past 7 days | Control mean $\%$ | 10 years |
| | Lubega et al. (2021) | Time spent working in own enterprise | Control mean $\%$ | 6 months |
| | Batista and Seither (2019) | Workdays per week | Control mean $\%$ | 6 months |
| | John and Orkin (2022) | Total hours of work in past 7 days | Control mean $\%$ | 30-36 months |
| | Orkin et al. (2023) | Respondents' days of labour supplied across various activities in a year | Control mean $\%$ | 17 months |
| | Bernard et al. (2025) | Aspirations index | Control SD $\%$ | 52 months |
| | Beaman et al. (2012) | Wishes child to graduate or get higher education | Control mean $\%$ | 36-48 months |
| | Batista and Seither (2019) | Having a goal | Control mean $\%$ | 6 months |
| Aminationa | Orkin et al. (2023) | Aspirations index | Control SD $\%$ | 17 months |
| Aspirations | Lubega et al. (2021) | Ambition defined as a dummy | Control mean $\%$ | 6 months |
| | Macours and Vakis (2014) | Strong positive expectations about the future | Control mean $\%$ | 9 months |
| | Cecchi et al. (2022) | Aspirations index | Control SD $\%$ | 3 months |
| | Rojas Valdes, Wydick, and Lybbert (2021) | Aspirations index | Control SD $\%$ | 12 months |
| | Bernard et al. (2025) | Expected years of education of oldest child | Control mean % | 52 months |
| Expected education | Orkin et al. (2023) | Expected education for eldest child | Control mean $\%$ | 17 months |
| | Baranov et al. (2020) | Expected grade attainment | Control mean $\%$ | 7 years |
| - | Bernard et al. (2025) | Educational expenditure | Control mean $\%$ | 52 months |
| T-l | John and Orkin (2022) | Total educational expenditure | Control SD $\%$ | 30-36 months |
| Educational investment | Baranov et al. (2020) | Monthly expenditure on education (ln) | Logarithmic value | 7 years |
| | Orkin et al. (2023) | Total educational expenditure for children aged between 6 and 20 years old | Control mean $\%$ | 17 months |

Table A.26: DETAILS ON VARIABLE STANDARDIZATION IN APPENDIX FIGURE A.5

Notes: The coefficients reported in Figure A.5 correspond to the "psychological" treatment arms of the different studies cited, except for Banerjee et al. (2015), where we report the three-year effects of a graduation program in Ethiopia. All monetary values are converted to 2016 USD PPP.

References

Alan, S., T. Boneva, and S. Ertac. 2019. "Ever Failed, Try Again, Succeed Better: Results from a Randomized Educational Intervention on Grit." *The Quarterly Journal of Economics* 134 (3): 1121–1162.

Anderson, M. L. 2008. "Multiple Inference and Gender Differences in the Effects of Early Intervention: A Reevaluation of the Abecedarian, Perry Preschool, and Early Training Projects." *Journal of the American Statistical Association* 103 (484): 1481–1495.

Ashraf, N., D. S. Karlan, and W. Yin. 2006. "Tying Odysseus to the Mast: Evidence from a Commitment Savings Product in the Philippines." *Quarterly Journal of Economics* 121 (2): 635–672.

Bandura, A. 2001. "Social Cognitive Theory: An Agentic Perspective." Annual Review of Psychology 52 (1): 1–26.

Banerjee, A., E. Duflo, N. Goldberg, D. Karlan, R. Osei, W. Parienté, J. Shapiro, B. Thuysbaert, and C. Udry. 2015. "A Multifaceted Program Causes Lasting Progress for the Very Poor: Evidence from Six Countries." *Science* 348 (6236): 1260799.

Baranov, V., S. Bhalotra, P. Biroli, and J. Maselko. 2020. "Maternal Depression, Women's Empowerment, and Parental Investment: Evidence from a Randomized Controlled Trial." *American Economic Review* 110 (3): 824–59.

Batista, C. and J. Seither. 2019. "Aspirations, Expectations, Identities: Behavioral Constraints of Micro-entrepreneurs." *NOVAFRICA Working Paper Series* 1906: 1–28.

Beaman, L., E. Duflo, R. Pande, and P. Topalova. 2012. "Female Leadership Raises Aspirations and Educational Attainment for Girls: A Policy Experiment in India." *Science* 335 (6068): 582–586.

Beegle, K., J. De Weerdt, J. Friedman, and J. Gibson. 2012. "Methods of Household Consumption Measurement Through Surveys: Experimental Results from Tanzania." *Journal of Development Economics* 98 (1): 3–18.

Belloni, A., V. Chernozhukov, and C. Hansen. 2014. "Inference on Treatment Effects after Selection among High-Dimensional Controls." *The Review of Economic Studies* 81 (2): 608–650.

Benjamini, Y., A. M. Krieger, and D. Yekutieli. 2006. "Adaptive Linear Step-up Procedures that Control the False Discovery Rate." *Biometrika* 93 (3): 491–507.

Bernard, T., S. Dercon, K. Orkin, and A. S. Taffesse. 2014. "The Future in Mind: Aspirations and Forward-Looking Behaviour in Rural Ethiopia." *Bureau for Research and Economic Analysis of Development Working Paper* 429: 1–42.

Bernard, T., S. Makhija, D. J. Spielman, and G. T. Abate. 2019. "The (Marginal) Cost of Technology Adoption: A Cost-Effectiveness Analysis of Digital Green's Video-Mediated Agricultural Extension Approach in Ethiopia." *International Food Policy Research Institute Project Note* : 1–6.

Bernard, T. and A. S. Taffesse. 2014. "Aspirations: An Approach to Measurement with Validation Using Ethiopian Data." *Journal of African Economies* 23 (2): 189–224.

Binswanger, H. P. 1980. "Attitudes Toward Risk: Experimental Measurement in Rural India." *American Journal of Agricultural Economics* 62 (3): 395–407.

Blattman, C., J. Jamison, and M. Sheridan. 2017. "Reducing Crime and Violence: Experimental Evidence from Cognitive Behavioral Therapy in Liberia." *American Economic Review* 107 (4): 1165–1206.

Blattman, C., S. Chaskel, J. C. Jamison, and M. Sheridan. 2023. "Cognitive Behavioral Therapy Reduces Crime and Violence over Ten Years: Experimental Evidence." *American Economic Review: Insights* 5 (4): 527–45.

Cantril, H. 1966. *The Pattern of Human Concern.* New Brunswick, New Jersey: Rutgers University Press.

Casey, K., R. Glennerster, and E. Miguel. 2012. "Reshaping Institutions: Evidence on Aid Impacts using a Preanalysis Plan." *The Quarterly Journal of Economics* 127 (4): 1755–1812.

Cecchi, F., A. Garcia, R. Lensink, and B. Wydick. 2022. "Aspirational Hope, Dairy Farming Practices, and Milk Production: Evidence from a Randomized Controlled Trial in Bolivia." *World Development* 160: 106087.

Cilliers, J., N. Elashmawy, and D. McKenzie. 2024. "Using Post-Double Selection Lasso in Field Experiments." *World Bank Policy Research Working Paper Series* 10931: 1–44.

Dalton, P., S. Ghosal, and A. Mani. 2016. "Poverty and Aspirations Failure." *Economic Journal* 126: 165–188.

Deaton, A. 1992. Understanding Consumption, Clarendon Lectures in Economics. Oxford: Oxford University Press.

Deininger, K., D. A. Ali, and T. Alemu. 2008. "Assessing the Functioning of Land Rental Markets in Ethiopia." *Economic Development and Cultural Change* 57 (1): 67–100.

Duckworth, A. L., C. Peterson, M. D. Matthews, and D. R. Kelly. 2007. "Grit: Perseverance and Passion for Long-Term Goals." *Journal of Personality and Social Psychology* 92 (6): 1087–1101.

Falk, A., A. Becker, T. Dohmen, B. Enke, D. Huffman, and U. Sunde. 2018. "Global Evidence on Economic Preferences." *Quarterly Journal of Economics* 133 (4): 1645–1692.

Feagin, J. R. 1975. *Subordinating the Poor: Welfare and American Beliefs*. Englewood Cliffs, New Jersey: Prentice Hall.

Genicot, G. and D. Ray. 2017. "Aspirations and Inequality." *Econometrica* 85 (2): 489–519.

Heath, C., R. P. Larrick, and G. Wu. 1999. "Goals as Reference Points." *Cognitive Psychology* 38 (1): 79–109.

John, A. and K. Orkin. 2022. "Can Simple Psychological Interventions Increase Preventive Health Investment?" Journal of the European Economic Association 20 (3): 1001–1047.

Kerwin, J. T., N. Rostom, and O. Sterck. 2024. "Striking the Right Balance: Why Standard Balance Tests Over-reject the Null, and How to Fix It." *University of Washington Working Paper*: 1–56.

Kling, J. R., J. B. Liebman, and L. F. Katz. 2007. "Experimental Analysis of Neighborhood Effects." *Econometrica* 75 (1): 83–119.

Kőszegi, B. and M. Rabin. 2006. "A Model of Reference-Dependent Preferences." *The Quarterly Journal of Economics* 121 (4): 1133–1165.

Levenson, H. 1981. "Differentiating Among Internality, Powerful Others, and Chance." In *Research with the Locus of Control Construct Vol.1*, edited by H. M. Lefcourt. New York: Academic Press, 15–63.

Lubega, P., F. Nakakawa, G. Narciso, C. Newman, A. N. Kaaya, C. Kityo, and G. A. Tumuhimbise. 2021. "Body and Mind: Experimental Evidence from Women Living with HIV." *Journal of Development Economics* 150: 102613.

Macours, K. and R. Vakis. 2014. "Changing Households' Investment Behaviour Through Social Interactions with Local Leaders: Evidence from a Randomised Transfer Programme." *The Economic Journal* 124 (576): 607–633.

Olken, B. A. 2015. "Promises and Perils of Pre-analysis Plans." *Journal of Economic Perspectives* 29 (3): 61–80.

Orkin, K., R. Garlick, M. Mahmud, R. Sedlmayr, J. Haushofer, and S. Dercon. 2023. "Aspiring to a Better Future: Can a Simple Psychological Intervention Reduce Poverty?" *National Bureau of Economic Research Working Paper* 31735: 1–102.

Rojas Valdes, R. I., B. Wydick, and T. J. Lybbert. 2021. "Can Hope Elevate Microfinance? Evidence from Oaxaca, Mexico." Oxford Economic Papers 74 (1): 236–264.

Ubfal, D. 2016. "How General Are Time Preferences? Eliciting Good-Specific Discount Rates." *Journal of Development Economics* 118: 150–170.